# FIVE ESTUARIES OFFSHORE WIND FARM

#### FIVE ESTUARIES OFFSHORE WIND FARM

#### 4.1 BOOK OF REFERENCE (CLEAN)

Application Reference: Document Number: Revision: Pursuant to: Eco-Doc Number: Date: EN010115 4.1 D Change Request 005023892-07 October 2024

#### COPYRIGHT © Five Estuaries Wind Farm Ltd

All pre-existing rights reserved.

In preparation of this document Five Estuaries Wind Farm Ltd has made reasonable efforts to ensure that the content is accurate, up to date and complete for purpose.

Revision	Date	Status/Reason for Issue	Originator	Checked	Approved
А	Mar-24	DOC Application	Dalcour Maclaren	VEOWF	VEOWF
В	Oct-24	Deadline 1	Dalcour Maclaren	VEOWF	VEOWF
С	Oct-24	Change Request	Dalcour Maclaren	VEOWF	VEOWF
D	Oct-24	Change Request	Dalcour Maclaren	VEOWF	VEOWF

### $\vee \Xi$

#### CONTENTS

1	Introduction	V
2	Rights which may be acquired	viii
3	Structure of this Book of Reference	xxvi
4	Book of Reference Notes	xxvii
5	Book of Reference Interests	xxviii

#### **TABLES**

Table 1 Land Plan colour key	vi
Table 2 Categories of new rights and restrictive covenants	viii

#### DEFINITIONS

Term, abbreviation, acronym or initialism	Definition
2008 Act	Planning Act 2008
2009 Regulations	Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009
Applicant	Five Estuaries Offshore Wind Farm Limited
Authorised development	The development authorised by the Five Estuaries development consent order, if made
DCO	Development Consent Order
Order	The Five Estuaries DCO, if made
Order land	The land shown on the Land Plans which is within the limits of land to be acquired or used and described in the Book of Reference.
Plot	Each piece or area of land identified individually in the Book of Reference; any reference to Plots or a numbered Plot shall be construed accordingly.



#### 1 INTRODUCTION

- 1.1 This Book of Reference accompanies the proposed Development Consent Order ('the Order') for the Five Estuaries Offshore Wind Farm (the 'authorised development'), as required by Regulation 5(2)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 (the '2009 Regulations').
- 1.2 The land described in this Book of Reference, which is to be subject to powers of compulsory acquisition and/or temporary possession, is required to deliver the works comprising the authorised development which are described in detail in Part 1 of Schedule 1 of the Order and shown on the works plans. This Book of Reference is part of the application documents for the authorised development and should be read in conjunction with the following:
  - Land Plans (application document 2.3),
  - Statement of Reasons (application document 4.3),
  - Special category Land Plan (application document 2.4),
  - Crown Land Plan (application document 2.16),
  - draft DCO (application document 3.1).
- 1.3 Every parcel of land that is affected is identified on a plot by plot basis and a unique number has been ascribed to each Plot. The Plots are shown on the Land Plans which accompany the Order. For each Plot it identifies whether the Applicant (Five Estuaries Offshore Wind Farm Limited) is seeking the power to acquire that Plot outright, the power to create and/ or acquire permanent rights, and/ or the power to create and/ or acquire temporary rights of possession and use. It also lists the persons with land outside the order limits who may be entitled to make a relevant claim for compensation due to the effects of construction or when the authorised development is operational.
- 1.4 The Order seeks powers to compulsorily acquire land and new rights (both temporary and permanent) for the purposes of the construction, operation and maintenance of the authorised development. Some of the Plots identified in Part 1 of the Book of Reference will be subject to acquisition of permanent rights (including restrictive covenants) pursuant to Article 23 (Compulsory acquisition of rights) and Schedule 7 (Land in which only new rights etc. may be required) of the Order. These Plots are shown coloured blue on the land plans.
- 1.5 Plots that are subject to powers of temporary possession, such as for the purpose of access or areas only required during construction are listed in Schedule 6 of the Order and shown coloured yellow on the land plans.
- 1.6 Plots that are subject to powers of non-exclusive temporary possession, such as for the purpose of access or areas only required during construction are listed in Schedule 6 of the Order and shown coloured green on the land plans.
- 1.7 The colours shown on the land plans indicate the type of acquisition sought as set out in the table below.

#### Table 1 Land Plan colour key

Colour of the Plot on land plans	Description of acquisition sought in Book of reference	Acquisition sought	Principal relevant DCO article(s)
Pink	"Freehold Acquisition"	Compulsory acquisition of all interests and rights in land (including as required, subsoil, surface land)	Article 22, Compulsory acquisition of land; Article 24, Compulsory acquisition of rights
Blue	"Acquisition of Rights"	Creation and compulsory acquisition of new rights (including where necessary, a right to impose restrictive covenants)	Article 24, Compulsory acquisition of rights; Article 26, Private Rights
Yellow	"Temporary Possession"	Temporary possession and use of land, primarily during construction	Article 31, Temporary use of land for carrying out the authorised development; Article 32, Temporary use of land for maintaining the authorised development
Green	"Temporary Possession (Non- exclusive)"	Non-exclusive temporary possession and use of land, primarily during construction	Article 31, Temporary use of land for carrying out the authorised development; Article 32, Temporary use of land for maintaining the authorised development



1.8 The Applicant has taken the cautious approach of seeking powers of compulsory acquisition (or rights of use) in respect of all Plots of land required for the scheme even where it already holds an interest or presumes it holds an interest in the land. This approach has been taken to ensure that it has the right to acquire the interests it needs in the whole of the Order limits even where an unidentified owner later asserts an interest in land which the Applicant believes it owns or has rights.



#### 2 **RIGHTS WHICH MAY BE ACQUIRED**

2.1 For Plots in which rights are to be acquired or restrictive covenants are to be imposed, the rights sought for the authorised development have been categorised as shown in the table below.

Number of Plot shown on Land Plans	Purpose for which rights may be acquired
01-007; 01-008; 01-009; 01- 010; 01-011; 01-012; 02- 001; 02-001A; 02-004; 02- 005; 02-008; 02-009; 02- 010; 03-001; 03-003; 03- 011; 03-014; 04-001; 04- 003; 04-004; 04-007; 04- 003; 04-004; 04-007; 04- 020; 05-001; 05-002; 05- 013; 05-019; 05-020; 05- 024; 06-010; 06-011; 06- 017; 07-007; 07-011; 08- 008; 08-011; 08-012; 08- 013; 08-014; 08-019; 08- 020; 08-021; 08-022; 08- 026; 09-008; 09-010; 09- 013; 09-014; 09-017; 09- 020; 09-024; 10-001; 10- 010; 10-011; 10-013; 11- 001; 11-006; 11-008; 11- 009; 11-012; 12-004; 12- 013; 13-007; 13-008; 13- 022; 13-023; 14-001; 14- 004; 14-031; 14-037; 14- 038; 14-039; 14-040; 14- 041; 14-044; 14-045; 14- 046; 16-004; 16-008; 16- 009; 16-012; 16-013; 16- 014; 16-020; 16-021; 17- 001; 17-002; 17-010; 17- 011	<ul> <li>Cable rights and restrictive covenants</li> <li>1. Cable rights</li> <li>Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to— <ul> <li>a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables (including the removal of materials including spoil) in, under, over and/or on the Land, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, roads and railways;</li> <li>b) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the additional ducts for electricity cables (including the removal of materials including spoil) in, under, over and/or on the Land, allow the installation and use of electrical cables in the additional ducts, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional</li> </ul> </li> </ul>



Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	drilling beneath sea defences, watercourses, roads and railways;
	<ul> <li>c) enter, be on, and break open and break up the surface of the Land and remain with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of installing, operating and maintaining the cables or additional ducts, transmitting electricity along the cables or use of electrical infrastructure and the cables;</li> </ul>
	<ul> <li>d) to benefit from continuous vertical and lateral support for the authorised development;</li> </ul>
	<ul> <li>e) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting, removing and replacing the electrical infrastructure and cables;</li> </ul>
	<ul> <li>f) construct and install and thereafter use the Land for all necessary purposes for the commissioning, construction, repair, testing and maintenance of the cables in, on or under the Land;</li> </ul>
	<ul> <li>g) place and use plant, machinery, structures and temporary structures within the Land for the purposes of the installation, construction, maintenance, repairing, renewing, upgrading, inspecting, removal and replacing of the ducting, electrical infrastructure and cables, and to erect temporary signage and provide measures for the benefit of public and personnel safety;</li> </ul>
	<ul> <li>h) erect temporary bridges and supporting or protective structures for the purposes of access to adjoining land and highway;</li> </ul>
	<ul> <li>erect fencing, gates, walls, barriers or other means of enclosure, and create secure working areas and compounds including trenchless installation technique compounds and working areas;</li> </ul>
	<ul> <li>j) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any</li> </ul>



Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	period during which construction, maintenance, repair or renewal is being carried out;
	<ul><li>k) effect access and egress to the highway;</li></ul>
	<li>I) make such investigations in or on the Land as required;</li>
	<ul> <li>m) alter, fell, lop or cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be standing on the Land or other land which would if not felled, lopped, cut or removed would obstruct or interfere with the operation of the cables and ancillary equipment including ducting;</li> </ul>
	<ul> <li>n) to take and use, remove and discharge water from the Land, and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, install, use, inspect, maintain, adjust, alter, renew, repair, test or cleanse drainage schemes on the Land or reinstate the any existing drainage scheme on the Land;</li> </ul>
	<ul> <li>o) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers and providing connection to the authorised development);</li> </ul>
	<ul> <li>p) erect and remove temporary fencing, gates, walls, barriers or other means of enclosure, remove fences and structures within the Land during any period during which construction, maintenance, repair or renewal is being carried out (subject to erection of any temporary stock-proof fencing as is reasonably required and the re-instatement or suitable replacement of the fences or structures following the end of each period of exercise of the rights);</li> </ul>
	<ul> <li>q) store and stockpile materials (including excavated material);</li> </ul>



Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	<ul> <li>r) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and monitoring the use of any trenchless installation technique, to keep in place and monitor the same through construction, maintenance repair, replacement or decommissioning and to reinstate the Land;</li> </ul>
	<ul> <li>s) to excavate materials below ground level, including soils, and to store and re-use or dispose of the same, and in so excavating to undertake any works, including works of protection or removal of archaeological remains as may be required by any written scheme of investigation approved under this Order;</li> </ul>
	<ul> <li>t) lay out temporary paths and bridleways for public use as temporary diversions for public rights of way which are interfered with during any period in which construction, maintenance, repair or renewal decommissioning is being carried out;</li> </ul>
	<ul> <li>u) to install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, remediation works for environmental or ecological mitigation or enhancement works, including temporary works for noise alleviation measures and the installation of temporary barriers for the protection of fauna;</li> </ul>
	<ul> <li>v) carry out such works (together with associated fencing) required by a planning permission and/or consent now or to be granted over the Land and/or in accordance with any necessary licences relating to protected species and/or wildlife; and</li> </ul>
	<ul> <li>w) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, maintain and inspect underground cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and other conducting media together with conduits or pipes for containing the same in and under the Land.</li> </ul>
	2. Restrictive covenants
	A restrictive covenant over the Land for the benefit of the remainder of the Order land to—
	<ul> <li>a) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any</li> </ul>

Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	buildings or construction erection or works of any kind (including the foundations or footings thereto);
	<ul> <li>b) prevent anything to be done by way of hard surfacing of the Land with concrete of any kind or with any other material or surface whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed if the proposed surfacing would not cause damage to relevant part of the authorised development nor make it materially more difficult or expensive to maintain the authorised development);</li> </ul>
	c) to prevent anything to be done by way of excavation of any kind in the Land nor any activities which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised development, alter, increase or decrease ground cover or soil levels in any manner whatsoever without the consent in writing of the undertaker save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.6m for the purposes of arable farming) or are required to be carried out by National Grid in order to exercise their statutory functions or rights in relation to their apparatus (if any) within the Land;
	<ul> <li>d) to prevent the planting or growing within the Land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised development nor make it materially more difficult or expensive to access the relevant part of the authorised development);</li> </ul>
	<ul> <li>e) to prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised development;</li> </ul>
	<ul> <li>f) to prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised development; and</li> </ul>
	<ul> <li>g) to prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of</li> </ul>



Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	ecological mitigation areas or areas of habitat creation including any ploughing or grazing without the prior written consent of the undertaker
03-007; 03-007A; 04-002; 05-014; 06-001; 06-013; 06- 014; 06-015; 06-018; 07- 013; 08-018; 09-007; 09- 009; 09-012; 11-007; 11- 018; 12-012; 13-018; 13- 019; 13-020; 14-029; 14- 042; 14-043; 16-002; 16- 003; 16-005; 16-006; 16- 007; 16-015; 16-016; 16- 018; 16-019; 17-009; 17- 019; 17-020; 17-021	<ul> <li>Cable rights and restrictive covenants under existing highway and rail infrastructure</li> <li>1.Cable rights</li> <li>Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to— <ul> <li>a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables (including the removal of materials including spoil) in, under, over and/or on the Land, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, roads and railways;</li> <li>b) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the additional ducts for electricity cables (including the removal of materials including spoil) in, under, over and/or on the Land, allow the installation and use of electrical cables in the additional ducts, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables"), and in doing so, to use or resort to trenchless installation aducts, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitti</li></ul></li></ul>

Purpose for which rights may be acquired
vehicles, machinery, apparatus and equipment which is ancillary to the purposes of installing, operating and maintaining the cables or additional ducts, or the use of the electrical infrastructure and cables;
d) to benefit from continuous vertical and lateral support for the authorised development;
e) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting, removing and replacing ducting, electrical infrastructure and the cables;
f) construct and install and thereafter use the Land for all necessary purposes for the commissioning, construction, repair, testing and maintenance of the ducting, electrical infrastructure and cables in, on or under the Land;
g) place and use plant, machinery, structures and temporary structures within the Land for the purposes of the installation, construction, maintenance, repairing, renewing, upgrading, inspecting, removal and replacing of the ducting, electrical infrastructure and cables and to erect temporary signage and provide measures for the benefit of public and personnel safety;
<ul> <li>h) erect temporary bridges and supporting or protective structures for the purposes of access to adjoining land and highway;</li> </ul>
<ul> <li>erect fencing, gates, walls, barriers or other means of enclosure, and create secure working areas and compounds including trenchless installation technique compounds and working areas;</li> </ul>
<ul> <li>j) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair or renewal is being carried out;</li> <li>k) effect access and egress to the highway;</li> </ul>



Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	<li>I) make such investigations in or on the Land as required;</li>
	m) alter, fell, lop or cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be standing on the Land or other land which would if not felled, lopped, cut or removed would obstruct or interfere with the operation of the cables and ancillary equipment including ducting;
	n) to take and use, remove and discharge water from the Land, and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, install, use, inspect, maintain, adjust, alter, renew, repair, test or cleanse drainage schemes on the Land or reinstate the any existing drainage scheme on the Land;
	<ul> <li>o) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers);</li> </ul>
	p) erect and remove temporary fencing, gates, walls, barriers or other means of enclosure, remove fences and structures within the Land during any period during which construction, maintenance, repair or renewal is being carried out (subject to erection of any temporary stock-proof fencing as is reasonably required and the re-instatement or suitable replacement of the fences or structures following the end of each period of exercise of the rights);
	q) store and stockpile materials (including excavated material);
	r) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and monitoring the use of any trenchless installation technique, to keep in place and monitor the same through construction, maintenance repair,

Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	replacement or decommissioning and to reinstate the Land;
	s) to excavate materials below ground level, including soils, and to store and re-use or dispose of the same, and in so excavating to undertake any works, including works of protection or removal of archaeological remains as may be required by any written scheme of investigation approved under this Order;
	<ul> <li>to install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, remediation works for environmental or ecological mitigation or enhancement works, including temporary works for noise alleviation measures and the installation of temporary barriers for the protection of fauna;</li> </ul>
	u) carry out such works (together with associated fencing) required by a planning permission and/or consent now or to be granted over the Land and/or in accordance with any necessary licences relating to protected species and/or wildlife; and
	v) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, maintain and inspect underground cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and other conducting media together with conduits or pipes for containing the same in and under the Land.
	2. Restrictive covenants
	A restrictive covenant over the Land for the benefit of the remainder of the Order land to—
	a) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction erection or works of any kind (including the foundations or footings thereto);
	b) to prevent the planting or growing within the Land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised



Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	development nor make it materially more difficult or expensive to access the relevant part of the authorised development);
	<ul> <li>c) to prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised development;</li> </ul>
	<ul> <li>d) to prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised development; and</li> </ul>
	e) to prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas or areas of habitat creation including any ploughing or grazing without the prior written consent of the undertaker.
02-002; 02-003; 02-006; 02-	Permanent access rights
007; 03-005; 03-013; 03- 015; 03-016; 03-017; 03- 018; 04-014; 04-015; 04-	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to—
016; 04-017; 04-018; 04- 019; 05-003; 05-004; 05- 005; 05-006; 05-009; 05- 012; 05-015; 05-016; 05- 017; 05-018; 05-025; 05- 026; 06-003; 06-004; 06- 006; 06-012; 06-016; 07- 001; 07-004; 07-008; 08- 002; 08-006; 08-009; 08- 010; 08-015; 08-016; 08- 017; 08-023; 09-001; 09-	a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of the Works, the inspection, testing, maintenance, renewal, upgrading, replacement and removal of the cables and connection into any adjacent cables and associated works, to take plant and equipment on to adjoining land and make such investigations in or on the Land which is ancillary for the purposes of exercise of the rights;
006; 09-011; 09-023; 10- 002; 10-003; 10-008; 10- 009; 10-012; 11-002; 11- 003; 11-004; 11-005; 11- 010; 11-011; 11-019; 11- 020; 12-005; 12-014; 13-	b) to construct, use, maintain and improve a permanent means of access including visibility splays, and retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the Land, adjoining land and highway;
001; 13-002; 13-003; 13- 004; 13-005; 13-006; 13- 011; 14-003; 14-008; 14- 010; 16-001; 16-010; 16-	<ul> <li>c) retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing adjoining land and highway;</li> </ul>



Number of Plot shown on Land Plans	Purpose for which rights may be acquired
011; 16-023; 17-015; 17- 016; 17-017; 17-018	d) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning is being carried out;
	<ul> <li>erect temporary bridges and supporting or protective structures for the purposes of access to adjoining land;</li> </ul>
	<ul> <li>f) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety;</li> </ul>
	g) alter, fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land;
	<ul> <li>repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seeding and other ecological measures together with the right to maintain, inspect and replant such trees, shrubs and landscaping;</li> </ul>
	<ul> <li>erect and remove temporary fencing, gates, walls, barriers or other means of enclosure; and</li> </ul>
	j) lay out temporary paths and bridleways for public use as temporary diversions for public rights of way which are interfered with during any period in which construction, maintenance, repair or renewal decommissioning is being carried out.
	National Grid substation works area rights
17-026; 17-027; 17-028; 17- 029; 17-030; 17-031; 18- 001; 18-002	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to—
	a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables (including the removal of materials including spoil) in, under, over and/or on the Land, together with such telemetry and fibre-optic lines,

Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (all collectively referred to as the "cables");
	<ul> <li>b) enter, be on, and break open and break up the surface of the Land and remain with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables;</li> </ul>
	c) to benefit from continuous vertical and lateral support for the authorised development;
	d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting the authorised development and for removing and replacing the cables;
	e) to use, maintain, renew improve and alter existing accesses, roads, streets, tracks or ways over the Land, providing that such use is not exclusive and exercise of this right must not prevent or unreasonably inhibit use by other parties;
	f) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning is being carried out;
	<ul> <li>g) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety;</li> </ul>
	<ul> <li>h) fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; and</li> </ul>

Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	i) remove and discharge water from the Land and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, to lay down, install, adjust, alter, construct, create, use, maintain, repair, renew, upgrade, inspect, remove and replace a drainage scheme on the Land; and
	<ul> <li>j) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers).</li> </ul>
	2.A restrictive covenant over the Land for the benefit of the remainder of the Order land to—
	<ul> <li>a) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction erection or works of any kind (including the foundations or footings thereto) without the prior written consent of the undertaker;</li> </ul>
	<ul> <li>b) to prevent the planting or growing within the Land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised development nor make it materially more difficult or expensive to access the relevant part of the authorised development);</li> </ul>
	<ul> <li>c) to prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised development;</li> </ul>
	<ul> <li>d) to prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with</li> </ul>

Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	the exercise of the rights or damage the authorised development;
19-001; 19-002; 19-003; 19- 004; 19-005; 19-006; 19- 007; 20-001	Lesser black-backed gull compensation access rights Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to—
	<ul> <li>a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of the Works, and associated works, to take plant and equipment on to adjoining land and make such investigations in or on the Land which is ancillary for the purposes of exercise of the rights;</li> </ul>
	<ul> <li>b) to use and maintain any quay, slipway, jetty or similar structure, with or without vessels, vehicles, plant, machinery, apparatus, equipment and materials for the purposes of taking accessing the Land, carrying out, monitoring and constructing, maintaining, improving repairing or decommissioning the Works;</li> </ul>
	<ul> <li>c) to use, maintain, repair and improve any permanent means of access including retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the Land, adjoining land and highway;</li> </ul>
	<ul> <li>d) construct, lay down, use and remove temporary access roads, ramps and other temporary crossings including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, footpaths, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains, and alter, widen, upgrade and improve existing roads, tracks and footpaths;</li> </ul>
	<ul> <li>erect temporary bridges and supporting or protective structures for the purposes of access to adjoining land;</li> </ul>
	<ul> <li>f) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety;</li> </ul>



Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	<ul> <li>g) alter, fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; and</li> <li>h) erect and remove temporary fencing, gates, walls,</li> </ul>
	barriers or other means of enclosure.
	Lesser black-backed gull compensation work rights and restrictive covenants
	1. Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to—
	<ul> <li>a) erect, maintain, repair, improve and remove permanent fencing, gates, barriers or other means of enclosure in order to create areas where predators are excluded;</li> </ul>
	<ul> <li>enter upon, pass, re-pass and remain on the Land with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of carrying out and maintaining the Works, surveying or to carry out monitoring of fauna;</li> </ul>
20-003	<ul> <li>place, retain, and maintain apparatus on the Land for the purposes of surveying or carrying out monitoring of fauna, including use of recording devices;</li> </ul>
	<ul> <li>install, execute, implement, retain, repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seeding and other ecological measures together with the right to maintain, inspect and replant such trees, shrubs and landscaping;</li> </ul>
	e) install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, remediation works for environmental or ecological mitigation or enhancement works, including temporary works and the installation of fencing for the protection of fauna; and
	<ul> <li>f) carry out such works (together with associated fencing or other means of enclosure) required by a planning permission and/or consent now or to be granted over the Land in accordance with any</li> </ul>

Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	necessary licences relating to protected species and/or wildlife.
	2. Restrictive covenants
	A restrictive covenant over the Land for the benefit of the remainder of the Order Land to—
	<ul> <li>a) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction erection or works of any kind (including the foundations or footings thereto);</li> </ul>
	<ul> <li>b) to prevent the planting or growing within the Land of any trees, shrubs or underwood without the prior consent in writing of the undertaker</li> </ul>
	<ul> <li>c) to prevent any part of any fence, gate, barrier or other enclosure erected as part of the works being altered, modified or removed without the prior consent in writing of the undertaker;</li> </ul>
	<ul> <li>d) to prevent any activity which would in the reasonable opinion of the undertaker result in the harm to or diminishment in the function of the ornithological compensation measures or areas of habitat creation including any ploughing or grazing without the prior written consent of the undertaker.</li> </ul>
	Drainage rights and restrictive covenants
	1. Drainage rights
17-004; 17-006; 17-007; 17- 008; 17-022	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to—
	<ul> <li>a) remove and discharge water from the Land and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, to lay down, install, adjust, alter, construct, create, use, maintain, repair, renew, upgrade, inspect, remove and replace a drainage scheme on the Land (the "drainage works");</li> </ul>
	<ul> <li>b) inspect, use mechanical excavation (including directional drilling and/or digging), reinstate,</li> </ul>

Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	remove, move or alter such part or parts of any drainage system on the Land for the purposes of the drainage works (including connecting the drainage works to any land drain as at the date of the drainage works);
	<ul> <li>c) enter, be on, and break up the surface of the Land and remain with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of the drainage works;</li> </ul>
	<ul> <li>d) store and stockpile materials (including excavated material);</li> </ul>
	<ul> <li>e) make such investigations in or on the Land as required for the purposes of the drainage works, include to create boreholes and trail excavation pits for the purposes of intrusively surveying the land;</li> </ul>
	<ul> <li>f) to excavate materials below ground level, including soils, and to store and re-use or dispose of the same, and in so excavating, to undertake any works, including works of protection or removal of archaeological remains as may be required by any written scheme of investigation approved under this Order;</li> </ul>
	<ul> <li>g) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of the drainage works;</li> </ul>
	<ul> <li>h) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety for the purposes of the drainage works;</li> </ul>
	<ul> <li>erect fencing, gates, walls, barriers or other means of enclosure, and create secure works areas or compounds and working areas for the purposes of the drainage works;</li> </ul>
	<ul> <li>j) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair or renewal of any drainage work is being carried out;</li> </ul>

Number of Plot shown on Land Plans	Purpose for which rights may be acquired
	k) effect access to the highway;
	<ol> <li>alter, fell, lop or cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be standing on the Land or other land which would if not felled, lopped, cut or removed obstruct or interfere with the drainage works;</li> </ol>
	<ul> <li>m) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers);</li> </ul>
	<ul> <li>n) to install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, remediation works for environmental or ecological mitigation or enhancement works, including temporary works for noise alleviation measures and the installation of temporary barriers for the protection of fauna; and</li> </ul>
	<ul> <li>o) to carry out such works (together with associated fencing) required by a planning permission and/or consent now or to be granted over the Land in accordance with any necessary licences relating to protected species and/or wildlife.</li> </ul>
	2. Restrictive Covenant
	A restrictive covenant over the Land for the benefit of the remainder of the Order land to:
	<ul> <li>a) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or</li> </ul>
	<ul> <li>b) construction erection or works of any kind (including the foundations or footings thereto); and</li> </ul>
	<ul> <li>c) c) prevent anything to be done by way of hard surfacing of the Land with concrete of any kind or with any other material or surface whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed if the proposed surfacing would not cause damage drainage works.</li> </ul>



#### **3 STRUCTURE OF THIS BOOK OF REFERENCE**

- 3.1 This Book of Reference is comprised of five Parts, in accordance with Regulation 7(1) of the Regulations, as follows:
  - a) Part 1 contains the names and addresses of those who own, lease, occupy or have another interest in the land that will be affected by the authorised development and the rights contained in the Order. A person is within Category 1 if the Applicant after making diligent inquiry, knows that they are an owner, lessee, tenant, or occupier of the land. A person is within Category 2 if they are interested in the land or have the power to sell and convey or release the land.

Parties recorded in Part 3 of this Book of Reference have also been included in Part 1.

- b) Part 2 lists persons who may be entitled to make a relevant claim, also called category 3 persons. A person is within Category 3 if the Applicant believes that, if the Order were to be made and fully implemented, they would or might be entitled to make a relevant claim as defined in section 57(6) of the Planning Act 2008 (the '2008 Act'). A relevant claim is a claim under section 10 of the Compulsory Acquisition Act 1965, a claim under part 1 of the Land Compensation Act 1973, or a claim under section 152(3) of the 2008 Act. This part therefore contains the names and addresses of those who have an interest in the Order land and may be entitled to claim compensation for loss resulting from the implementation of the Order and use of the authorised development. This part also contains the names and addresses of those without an interest in the Order land, who may be entitled to claim compensation for loss resulting from the implementation of the Order and use of the authorised development. This part also contains the names and addresses of those without an interest in the Order land, who may be entitled to claim compensation for loss resulting from the implementation of the Order and use of the authorised development.
- c) Part 3 contains the names and addresses of those entitled to enjoy easements or other private rights which it is proposed may be extinguished, suspended or interfered with in connection with the authorised development, pursuant to the Order;
- d) Part 4 identifies Plots in which there is a Crown interest that will be affected by the authorised development and the rights contained in the Order; and
- e) Part 5 identifies Plots which constitute "special category land" for the purposes of section 130 and 132 of the Planning Act 2008 that will be affected by the authorised development and the rights contained in the Order.



#### 4 BOOK OF REFERENCE NOTES

4.1 This Book of Reference provides the area in square metres of all land included in the DCO.

4.2 The term "approximately" is used before all Plot area measurements as these measurements are given in square metres and each measurement is rounded up to the nearest whole square metre.

4.3 Plot numbering follows the format *sheet number/ Plot number*. So for example, Plot "05-001" would describe Plot 001 on sheet 05 of the land plans.

4.4 In Part 1, a person is within Category 1 if the applicant, after making diligent inquiry, knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land.

4.5 In Part 1, a person is within Category 2 if the applicant, after making diligent inquiry, knows that the person (a) is interested in the land, or (b) has power (i) to sell and convey the land, or (ii) to release the land.

4.6 The Applicant confirms that relevant persons, including any newly identified interests, have been informed of their rights under section 102A of the Planning Act 2008.

#### $\vee \Xi$

#### 5 BOOK OF REFERENCE INTERESTS

[page intentionally left blank]

Schedule of Changes to plot interests							
Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reasons for Change	Revision	
16-016, 16-017, 16-018, 16-019, 16-020, 16-021, 16-022, 16-023	1/03/2024	David William Salmon	Category 1	Address Change	Ongoing due diligence has identified a new address for David William Salmon	Rev A	
N/A	19/03/2024	Patricia Maestrani	Category 3	Address Change	Ongoing due diligence has identified a new address for Patricia Maestrani	Rev A	
12-007	25/03/2024	Unknown	Category 2	Added interest	Ongoing due diligence has identified that an unknown rights interest was missed from the plot	Rev A	
12-006	28/03/2024	Harry Weavers	Category 2	Added interest	Ongoing due diligence has identified that Harry Weavers has an interest in the land	Rev A	
03-002, 03-003, 03-005, 03-006	02/04/2024	Barclays Security Trustee Limited	Category 2	Added interest	Ongoing due diligence has identified that Barclays Security Trustee Limited has an interest in the land	Rev A	
12-008, 12-009	03/04/2024	Anglian Maltings (Holdings) Limited	Category 1	Removed interest	Ongoing due diligence has identified that Anglian Maltings (Holdings) Limited no longer has an interest in the land	Rev A	
12-008	03/04/2024	Tungsten Colchester Limited	Category 2	Added interest	Ongoing due diligence has identified that Tungsten Colchester Limited has an interest in the land	Rev A	
12-008	03/04/2024	Foxes Property 3 S.A.R.L	Category 1	Added interest	Ongoing due diligence has identified that Foxes Property 3 S.A.R.L has an interest in the land	Rev A	
14-045, 14-046	03/04/2024	East Anglian Farm Rides	Category 2	Removed interest	Ongoing due diligence has identified that East Anglian Farm Rides no longer has an interest in the land	Rev A	

	nanges to plot in					
Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reasons for Change	Revision
N/A	04/04/2024	Charles James Tabor	Category 1	Removed interest	Ongoing due diligence has identified that Charles James Tabor no longer has an interest in the land	Rev A
17-025	04/04/2024	Charles James Tabor	Category 1	Removed interest	Ongoing due diligence has identified that Charles James Tabor no longer has an interest in the land	Rev A
N/A	04/04/2024	The Executor of the Estate of the late Charles James Tabor	Category 1	Added interest	Ongoing due diligence has identified that The Executor of the Estate of the late Charles James Tabor has an interest in the land	Rev A
17-025	04/04/2024	The Executor of the Estate of the late Charles James Tabor	Category 1	Added interest	Ongoing due diligence has identified that The Executor of the Estate of the late Charles James Tabor has an interest in the land	Rev A
N/A	08/04/2024	Nicholas Paul Maestrani	Category 3	Removed interest	Ongoing due diligence has identified that Nicholas Paul Maestrani no longer has an interest in the land	Rev A
N/A	08/04/2024	The Executor of the Estate of the Late Nicholas Paul Maestrani	Category 3	Added interest	Ongoing due diligence has identified that The Executor of the Estate of the Late Nicholas Paul Maestrani has an interest in the land	Rev A
07-013	09/04/2024	Michael George Robert Goosetree	Category 1	Address Change	Ongoing due diligence has identified a new address for Michael George Robert Goosetree	Rev A
N/A	09/04/2024	Michael George Robert Goosetree	Category 3	Address Change	Ongoing due diligence has identified a new address for Michael George Robert Goosetree	Rev A
N/A	16/04/2024	Foxes Property 3 S.A.R.L	Category 3	Added interest	Ongoing due diligence has identified that Foxes Property 3 S.A.R.L has an interest in the land	Rev A

Schedule of Changes to plot interests							
Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reasons for Change	Revision	
13-021	16/04/2024	Anglian Maltings (Holdings) Limited	Category 2	Removed interest	Ongoing due diligence has identified that Anglia Maltings (Holdings) Limited no longer has an interest in the land	Rev A	
N/A	16/04/2024	Simon David Chidgey	Category 3	Removed interest	Ongoing due diligence has identified that Simon David Chidgey no longer has an interest in the land	Rev A	
N/A	16/04/2024	Graham Peter Lucas	Category 3	Added interest	Ongoing due diligence has identified that Graham Peter Lucas has an interest in the land	Rev A	
N/A	16/04/2024	Sarah Kate Lucas	Category 3	Added interest	Ongoing due diligence has identified that Sarah Kate Lucas has an interest in the land	Rev A	
08-018	18/04/2024	Spencer Leigh Brown	Category 1	Address Change	Ongoing due diligence has identified a new address for Spencer Leigh Brown	Rev A	
N/A	18/04/2024	Spencer Leigh Brown	Category 3	Address Change	Ongoing due diligence has identified a new address for Spencer Leigh Brown	Rev A	
07-007, 07-011	24/04/2024	M Scott Property Group Limited	Category 2	Added interest	Ongoing due diligence has identified that M Scott Property Group Limited has an interest in the land	Rev A	
N/A	07/05/2024	Paul Calkin	Category 3	Address Change	Ongoing due diligence has identified a new address for Paul Calkin	Rev A	
N/A	13/05/2024	Ruth Naomi Fernandez	Category 3	Address Change	Ongoing due diligence has identified a new address for Ruth Naomi Fernandez	Rev A	
N/A	13/05/2024	Jemma White	Category 3	Address Change	Ongoing due diligence has identified a change in surname for Jemma from Keleher to White	Rev A	

Schedule of Changes to plot interests								
Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reasons for Change	Revision		
N/A	13/05/2024	Graham Clive Gilbert	Category 3	Address Change	Ongoing due diligence has identified a new address for Graham Clive Gilbert	Rev A		
01-002, 01-004, 01-005, 01-006, 03-002, 03-003, 03-004, 03-005, 03-009, 03-014, 03-015, 03-016, 04-001, 04-002, 04-003, 04-004, 05-024, 05-026, 06-001, 06-007, 06-008, 06-011, 06-012, 06-014, 06-015, 06-014, 06-015, 06-016, 06-017, 06-018, 07-013, 08-001, 09-002, 09-003, 09-004, 09-003, 09-004, 09-003, 09-004, 09-005, 09-006, 09-007, 09-008, 09-011, 09-012, 10-018, 10-012, 10-013, 11-018, 12-004, 12-006, 12-007, 12-008, 12-009, 12-010, 13-003, 13-004, 13-005	23/05/2024	Openreach Limited	Category 2	Address Change	Ongoing due diligence has identified a new address for Openreach Limited	Rev A		

Schedule of Changes to plot interests							
Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reasons for Change	Revision	
$\begin{array}{r} 13-006, 13-008, \\ 13-013, 13-014, \\ 13-019, 13-020, \\ 14-003, 14-016, \\ 14-018, 14-019, \\ 14-021, 14-022, \\ 14-027, 14-029, \\ 14-027, 14-029, \\ 14-030, 14-031, \\ 14-033, 14-034, \\ 14-036, 14-038, \\ 14-036, 14-038, \\ 14-043, 14-046, \\ 15-001, 15-002, \\ 15-007, 15-011, \\ 15-013, 15-014, \\ 15-016, 15-017, \\ 15-022, 16-003, \\ 16-004, 16-005, \\ 17-002, 17-004, \\ 17-006, 17-007, \\ 17-008, 17-011, \\ 17-014, 17-020, \\ 17-021, 17-022, \\ 17-023, 17-024, \\ 17-025, 17-026, \\ 17-027, 17-028, \\ 17-030, 17-031, \\ 19-002, 19-005, \\ \end{array}$	23/05/2024	Openreach Limited	Category 2	Address Change	Ongoing due diligence has identified a new address for Openreach Limited	Rev A	

Schedule of Changes to plot interests							
Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reasons for Change	Revision	
19-006, 19-007, 20-001	23/05/2024	Openreach Limited	Category 2	Address Change	Ongoing due diligence has identified a new address for Openreach Limited	Rev A	
11-018, 12-001	17/06/2024	Richard Harrison- Osborne	Category 1	Address Change	Ongoing due diligence has identified a new address for Richard Harrison-Osbourne	Rev A	
11-018, 12-001	17/06/2024	Sara Carol Harrison- Osborne	Category 1	Address Change	Ongoing due diligence has identified a new address for Sara Carol Harrison-Osborne	Rev A	
N/A	17/06/2024	James Hendrie Fairley	Category 3	Removed interest	Ongoing due diligence has identified that James Hendrie Fairley no longer has an interest in the land	Rev A	
N/A	17/06/2024	June Mary Fairley	Category 3	Removed interest	Ongoing due diligence has identified that June Mary Fairley no longer has an interest in the land	Rev A	
N/A	17/06/2024	Christine Barber	Category 3	Added interest	Ongoing due diligence has identified that Christine Barber has an interest in the land	Rev A	
N/A	17/06/2024	James Francis Fairley	Category 3	Added interest	Ongoing due diligence has identified that James Frances Fairley has an interest in the land	Rev A	
17-025	26/06/2024	Rebecca Mason	Category 1	Added interest	Ongoing due diligence has identified that Rebecca Mason has an interest in the land	Rev A	
N/A	26/06/2024	Rebecca Mason	Category 3	Added interest	Ongoing due diligence has identified that Rebecca Mason has an interest in the land	Rev A	

Schedule of Changes to plot interests								
Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reasons for Change	Revision		
17-025	26/06/2024	Michael Hughes	Category 1	Added interest	Ongoing due diligence has identified that Michael Hughes has an interest in the land	Rev A		
N/A	26/06/2024	Michael Hughes	Category 3	Added interest	Ongoing due diligence has identified that Michael Hughes has an interest in the land	Rev A		
15-007, 15-009	26/06/2024	Holly Marie Florence Johnson	Category 1	Name change	Ongoing due diligence has identified middle names for Holly Johnson	Rev A		
15-007, 15-009	26/06/2024	Russell Albert Johnson	Category 1	Name change	Ongoing due diligence has identified middle names for Russell Johnson	Rev A		
15-007, 15-009	26/06/2024	Rachael Donna Thackery	Category 1	Name change	Ongoing due diligence has identified middle names for Rachel Thackery	Rev A		
15-007, 15-009	26/06/2024	John Paul Jeffery Traveller	Category 1	Name change	Ongoing due diligence has identified middle names for John Traveller	Rev A		
15-007, 15-009	26/06/2024	Kimberly Jane McFarlane	Category 1	Interest Removed	Ongoing due diligence has identified that Kimberly Jane McFarlane no longer has an interest in the land	Rev A		
15-007, 15-009	26/06/2024	The Executor of the Estate of the Late Stephen James McFarlane	Category 1	Removed interest	Ongoing due diligence has identified that The Executor of the Estate of the Late Stephen James McFarlane no longer has an interest in the land	Rev A		
15-009	26/06/2024	Mortgage Agency Services Number Two Limited	Category 2	Removed interest	Ongoing due diligence has identified that Mortgage Agency Services Number Two Limited no longer has an interest in the land	Rev A		

Schedule of Changes to plot interests							
Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reasons for Change	Revision	
15-009	26/06/2024	Skipton Building Society	Category 2	Added interest	Ongoing due diligence has identified that Skipton Building Society has an interest in the land	Rev A	
15-011, 15-013	28/06/2024	Linda Maureen Clachan	Category 1	Added interest	Ongoing due diligence has identified that Linda Maureen Clachan has an interest in the land	Rev A	
15-011, 15-013	28/06/2024	James Andrew Clachan	Category 1	Added interest	Ongoing due diligence has identified that James Andrew Clachan no longer needs a qualifier	Rev A	
15-011, 15-013	28/06/2024	Richard John Clachan	Category 1	Added interest	Ongoing due diligence has identified that Richard John Clachan has an interest in the land	Rev A	
15-007, 15-011, 15-013	28/06/2024	Richard John Clachan	Category 1	Address Change	Ongoing due diligence has identified a new address for Richard John Clachan	Rev A	
15-011	28/06/2024	The Agricultural Mortgage Corporation PLC	Category 2	Added interest	Ongoing due diligence has identified that a new qualifier for The Agricultural Mortgage Corporation PLC	Rev A	
15-007, 15-011	28/06/2024	The Executor of the Estate of the Late Marian Dorothy Burgoyne	Category 1	Removed interest	Ongoing due diligence has identified that The Executor of the Estate of the Late Marian Dorothy Burgoyne no longer has an interest in the land	Rev A	
15-007, 15-011	28/06/2024	James Tweed	Category 1	Removed interest	Ongoing due diligence has identified that James Tweed no longer has an interest in the land	Rev A	
15-007, 15-011	28/06/2024	Christopher Burgoyne	Category 1	Removed interest	Ongoing due diligence has identified that Christopher Burgoyne no longer has an interest in the land	Rev A	

Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reasons for Change	Revision
02-006, 02-007, 02-008, 02-009, 02-010	30/07/2024	Julie Watson	Category 2	Added interest	Ongoing due diligence has identified that Julie Watson has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	30/07/2024	Barry Cullum	Category 2	Added interest	Ongoing due diligence has identified that Barry Cullum has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	30/07/2024	Courtney Byrne	Category 2	Added interest	Ongoing due diligence has identified that Courtney Byrne has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	30/07/2024	Victoria Oxland	Category 2	Added interest	Ongoing due diligence has identified that Victoria Oxland has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	30/07/2024	Andrew Robinson	Category 2	Added interest	Ongoing due diligence has identified that Andrew Robinson has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	30/07/2024	Wendy Robinson	Category 2	Added interest	Ongoing due diligence has identified that Wendy Robinson has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	30/07/2024	Lindsey-Cher Johnson	Category 2	Added interest	Ongoing due diligence has identified that Lindsey-Cher Johnson has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	30/07/2024	Mazy King	Category 2	Added interest	Ongoing due diligence has identified that Mazy King has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	30/07/2024	Pat Watson	Category 2	Added interest	Ongoing due diligence has identified that Pat Watson has an interest in the land	Rev A

Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reasons for Change	Revision
02-006, 02-007, 02-008, 02-009, 02-010	30/07/2024	Louisa Thatcher	Category 2	Added interest	Ongoing due diligence has identified that Louisa Thatcher has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	30/07/2024	Katy O'Donnell	Category 2	Added interest	Ongoing due diligence has identified that Katy O'Donnell has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	30/07/2024	Tanya Wheeler	Category 2	Added interest	Ongoing due diligence has identified that Tanya Wheeler has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	19/08/2024	Yvonne Cullum	Category 2	Added interest	Ongoing due diligence has identified that Yvonne Cullum has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	19/08/2024	Andrea Woods	Category 2	Added interest	Ongoing due diligence has identified that Andrea Woods has an interest in the land	Rev A
02-006, 02-007, 02-008, 02-009, 02-010	21/08/2024	Michelle Miller	Category 2	Added interest	Ongoing due diligence has identified that Michelle Miller has an interest in the land	Rev A
13-008, 13-009, 13-010, 13-011, 13-012	29/08/2024	Barclays Bank UK PLC	Category 2	Interest Removed	Ongoing due diligence has identified that Barclays Bank UK PLC no longer has an interest in the land	Rev A
13-008, 13-009, 13-010, 13-011, 13-012	29/08/2024	Barclays Security Trustee Limited	Category 2	Added interest	Ongoing due diligence has identified that Barclays Security Trustee Limited has an interest in the land	Rev A

Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reason for change	Revision
11-016, 11-017, 11-019, 11-020, 12-002, 12-003, 12-004, 12-005, 12-006, 12-011	11/09/2024	Barclays Bank PLC	Category 2	Interest Removed	Ongoing due diligence has identified that Barclays Bank PLC no longer has a second charge on the land	Rev A
03-004	18/09/2024	Affinity Water Limited	Category 2	Interest Removed	Ongoing due diligence has identified that Affinity Water Limited no longer has an interest in the land	Rev A
03-004A	18/09/2024	Adam Charles Brown	Category 1	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-004A	18/09/2024	Essex County Council	Category 1	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-004A	18/09/2024	Joanna Marie Brown	Category 1	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-004A	18/09/2024	Unknown	Category 1	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-004A	18/09/2024	Adam Charles Brown	Category 2	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-004A	18/09/2024	Affinity Water Limited	Category 2	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-004A	18/09/2024	Joanna Marie Brown	Category 2	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A

Plot Number	Date of	Contact Name	Nature of Land	Change Made	Reason for change	Revision
	Change		Interest		-	
03-004A	18/09/2024	Openreach Limited	Category 2	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-004A	18/09/2024	Unknown	Category 2	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-007	18/09/2024	Affinity Water Limited	Category 2	Interest Removed	Ongoing due diligence has identified that Affinity Water Limited no longer has an interest in the land	Rev A
03-007	18/09/2024	Openreach Limited	Category 2	Interest Removed	Ongoing due diligence has identified that Openreach no longer has an interest in the land	Rev A
03-007A	18/09/2024	Adam Charles Brown	Category 1	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-007A	18/09/2024	Essex County Council	Category 1	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-007A	18/09/2024	Joanna Marie Brown	Category 1	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-007A	18/09/2024	Unknown	Category 1	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-007A	18/09/2024	Affinity Water Limited	Category 2	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A

Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reason for change	Revision
03-007A	18/09/2024	Openreach Limited	Category 2	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
03-007A	18/09/2024	Unknown	Category 2	Added interest	Plot extents have been amended which has created plot splits. Plot interests added	Rev A
N/A	19/09/2024	The Executor of the Estate of the Late Douglas Maitland Roberts	Category 3	Interest Removed	Ongoing due diligence has identified that The Executor of the Estate of the Late Douglas Maitland Roberts no longer has an interest in the land	Rev A
N/A	19/09/2024	Daphne Patricia Calkin	Category 3	Interest Removed	Ongoing due diligence has identified that Daphne Patricia Calkin no longer has an interest in the land	Rev A
N/A	19/09/2024	Paul Calkin	Category 3	Interest Removed	Ongoing due diligence has identified that Paul Calkin no longer has an interest in the land	Rev A
N/A	19/09/2024	Ruth Naomi Fernandez	Category 3	Interest Removed	Ongoing due diligence has identified that Ruth Naomi Fernandez no longer has an interest in the land	Rev A
N/A	19/09/2024	Barbara Jane Roscoe	Category 3	Added interest	Ongoing due diligence has identified that Barbara Jane Roscoe has an interest in the land	Rev A
N/A	19/09/2024	David Graham Roscoe	Category 3	Added interest	Ongoing due diligence has identified that David Graham Roscoe has an interest in the land	Rev A
05-008	27/09/2024	Eastern Power Networks	Category 2	Added interest	Ongoing due diligence has identified that Eastern Power Networks has an interest in the land	Rev B - CR

Schedule of Changes to plot interests							
Plot Number	Date of Change	Contact Name	Nature of Land Interest	Change Made	Reason for change	Revision	
02-001A	30/09/2024	Tendring District Council	Category 1	Added interest	Ongoing due diligence has identified that Tendring District Council has an interest in the land	Rev B - CR	

Schedule of Changes to plots							
Plot Number	Date of Change	Change Made	Reasons for Change	Revision			
03-004	18/09/2024	Extent of area size updated	Ongoing due diligence has identified an amendment to the plot extent	Rev A			
03-004a	18/09/2024	New plot added to Book of Reference and introduction pages	Ongoing due diligence has identified the requirement for a new plot creation	Rev A			
03-007	18/09/2024	Extent of area size updated	Ongoing due diligence has identified an amendment to the plot extent	Rev A			
03-007a	18/09/2024	New plot added to Book of Reference and introduction pages	Ongoing due diligence has identified the requirement for a new plot creation	Rev A			
08-001	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A			
08-005	03/04/2024	Amended plot description	As the result of a clerical error in the production of the Book of Reference, the plot description has been updated to rectify a mistake in direction	Rev A			
08-013	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A			
08-014	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A			
08-020	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A			
08-021	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A			
08-022	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A			

Schedule of Changes to plots								
Plot Number	Date of Change	Change Made	Reasons for Change	Revision				
08-023	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
08-025	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
08-026	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
09-001	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
09-002	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
09-003	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
09-005	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
09-006	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
09-008	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
09-015	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
09-016	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				

Schedule of Changes to plots								
Plot Number	lot Number Date of Change Change Made		Reasons for Change	Revision				
12-005	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
12-009	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
13-023	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
14-038	03/04/2024	Amended plot description	Ongoing due diligence has identified a more accurate plot description	Rev A				
01-007	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR				
02-001	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR				
05-008	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR				
07-002	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR				
08-024	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR				
09-023	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR				
12-005	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR				

Schedule of Changes to plots									
Plot Number	Date of Change	Change Made	Reasons for Change	Revision					
13-021	27/09/2024	Plot removed	The plot is no longer needed after the change application. The plot number is no longer in use	Rev B - CR					
13-024	27/09/2024	Plot removed	The plot is no longer needed after the change application. The plot number is no longer in use	Rev B - CR					
14-002	27/09/2024	Plot removed	The plot is no longer needed after the change application. The plot number is no longer in use	Rev B - CR					
14-035	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR					
15-002	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR					
15-010	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR					
15-013	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR					
17-008	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR					
20-002	27/09/2024	Plot removed	The plot is no longer needed after the change application. The plot number is no longer in use	Rev B - CR					
20-003	27/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR					
20-004	27/09/2024	Plot removed	The plot is no longer needed after the change application. The plot number is no longer in use	Rev B - CR					

Schedule of Changes to plots									
Plot Number	Date of Change	Change Made	Reasons for Change	Revision					
02-001A	30/09/2024	Plot added	Ongoing due diligence has identified a change in the Mean High Water, which has created a new plot	Rev B - CR					
01-006	30/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR					
01-008	30/09/2024	Extent of area size updated & Amended plot description	Ongoing due diligence has identified an amendment to the plot extent	Rev B - CR					
09-018	30/09/2024	Extent of area size updated & Amended plot description	Small change in size, no change to interests	Rev B - CR					
09-019	30/09/2024	Extent of area size updated & Amended plot description	Small change in size, no change to interests	Rev B - CR					
09-021	30/09/2024	Plot removed	The plot is no longer needed after the change application. The plot number is no longer in use	Rev B - CR					
09-022	30/09/2024	Plot removed	The plot is no longer needed after the change application. The plot number is no longer in use	Rev B - CR					
01-006	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D					
01-007	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D					
01-008	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D					
03-004	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D					

Schedule of Changes to plots								
Plot Number	Date of Change	Change Made	Reasons for Change	Revision				
03-007	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D				
03-007A	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D				
09-018	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D				
09-019	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D				
09-023	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D				
13-002	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D				
13-009	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D				
15-010	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D				
15-013	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D				
17-008	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D				
20-003	28/10/2024	Amended plot description	Due to clerical error, plot size has been updated in the book of reference	Rev D				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
	Temporary Possession (Non exclusive)		Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (as reputed owner) Unknown (as reputed owner)	NONE	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE Unknown	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Brian Leonard Cross 2 Sluice Cottages Manor Way CLACTON-ON-SEA CO15 5TZ (in respect of rights of access) Gunfleet Boating Club Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access) Irene Cross 2 Sluice Cottages Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access) Irene Cross 2 Sluice Cottages Manor Way CLACTON-ON-SEA CO15 5TZ (in respect of rights of access)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
01-001 cont'd						Maria Ann Martin 1 Sluice Cottages Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access) Mountview Estates PLC Mountview House 151 High Street LONDON N14 6EW (in respect of rights of access) Orsted Energy Solutions (UK) Limited 5 Howick Place LONDON SW1P 1WG (in respect of electricity apparatus) Port Of London Authority London River House Royal Pier Road GRAVESEND Kent DA12 2BG (in respect of rights of access)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
01-001 cont'd						Stephen Philip Martin 1 Sluice Cottages Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access) TC Gunfleet Sands OFTO Limited 3 More London Riverside LONDON SE1 2AQ (in respect of electricity apparatus) Unknown				
	exclusive)	way (off Frinton Road) leading	Station Road CLACTON-ON-SEA Essex	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 29 167))	Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of rights granted by a Conveyance dated 21 October 1930, in Conveyance dated 3 August 1932, reserved in a Conveyance dated 9 August 1977 and in respect of right of way and easement)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
01-002 cont'd					Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE	Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of sewerage apparatus) Brian Leonard Cross 2 Sluice Cottages Manor Way CLACTON-ON-SEA CO15 5TZ (in respect of rights of access) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
01-002 cont'd						Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of the legal easements granted contained in a Deed of Easement dated 11 October 2019, in a Deed of Grant dated 26 February 1986, in a Deed of Grant dated 4 April 1989 and in a Deed of Grant dated 16 January 1998) Gunfleet Boating Club Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access) Irene Cross 2 Sluice Cottages Manor Way CLACTON-ON-SEA CO15 5TZ (in respect of rights of access)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
01-002 cont'd						Maria Ann Martin 1 Sluice Cottages Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access) Mountview Estates PLC Mountview House 151 High Street LONDON N14 6EW (in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Orsted Energy Solutions (UK) Limited 5 Howick Place LONDON SW1P 1WG (in respect of electricity apparatus)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
01-002 cont'd						Port Of London Authority London River House Royal Pier Road GRAVESEND Kent DA12 2BG (in respect of rights of access) Stephen Philip Martin 1 Sluice Cottages Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access) TC Gunfleet Sands OFTO Limited 3 More London Riverside LONDON SE1 2AQ (in respect of electricity apparatus) Unknown (in respect of covenants contained in Conveyance dated 16 May 1902) Unknown (in respect of covenants contained in a Conveyance dated 7 October 1930)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-002 cont'd						Unknown (in respect of covenants contained in a Conveyance dated 23 April 1938)			
01-003	Temporary Possession	10490 square metres of access way off Frinton Road leading to public road (Manor Way) grassland, verges, access tracks and buildings (south of The Esplanade, Holland-on-Sea) and public footpath (FP 29 167)	Town Hall Station Road CLACTON-ON-SEA Essex	NONE		Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of rights granted by a Conveyance dated 21 October 1930, in Conveyance dated 3 August 1932, reserved in a Conveyance dated 9 August 1977 and in respect of right of way and easement) Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of sewerage apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-003 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of the legal easements granted contained in Deed of Easement dated 11 October 2019, in a Deed of Grant dated 26 February 1986, in a Deed of Grant dated 4 April 1989 and in a Deed of Grant dated 16 January 1998) Port Of London Authority London River House Royal Pier Road GRAVESEND Kent DA12 2BG (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-003 cont'd						Unknown (in respect of covenants contained in Conveyance dated 16 May 1902) Unknown (in respect of covenants contained in a Conveyance dated 7 October 1930) Unknown (in respect of covenants contained in a Conveyance dated 23 April 1938)			
01-004	Temporary Possession (Non exclusive)	63 square metres of hardstanding (south of Holland Brook) and public footpath (FP 29 167)			Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 29 167))	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-004 cont'd			Unknown (as reputed owner)		Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE Unknown				
	Temporary Possession (Non exclusive)	hardstanding (south of Holland Brook) and public footpath (FP 29 167)		NONE	County Hall Market Road Chelmsford Essex	Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of rights as contained in a transfer dated 1 September 1989) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers				
01-005 cont'd						Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of rights granted by a Deed of Grant dated 12 April 1938) The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP (in respect of rights granted by a Deed of Grant dated 12 December 1938) Unknown (in respect of restrictive covenants contained in a Conveyance dated 6 November 1900) Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-006	Temporary Possession (Non exclusive)	track (east of Manor Way,	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as reputed owner) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as reputed owner) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (as reputed owner)	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of maintaining the sea wall) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 29 167)) Frinton Golf Club Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
01-006 cont'd			The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP (as reputed owner) Unknown (as reputed owner)		Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP Unknown				
01-007	Acquisition of Rights	21703 square metres of foreshore, sea wall, rock armour, grassland and sloping masonry (west of Second Avenue, Frinton-on-Sea)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as reputed owner)	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of maintaining the sea wall)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-007 cont'd			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as reputed owner) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (as reputed owner) Unknown (as reputed owner)		Chelmsford	Unknown (in respect of an Admitted Claim for C W Hayne dated 28 September 1933 and an Admitted claim for the executors of G F Beaumont dated 1961)			
01-008	Acquisition of Rights	foreshore, grassland, access track, sea wall, rock armour, sloping masonry and drains (west of Second Avenue,	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as reputed owner)	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of maintaining the sea wall)	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
01-008 cont'd			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as reputed owner)		Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public bridleway (BR 2 164) and public footpaths (FP 3 164, FP 41 164 and FP 29 167))				
			Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (as reputed owner) The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP (as reputed owner)		Frinton Golf Club Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE				

	Land v	vhich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	d Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to of Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-008 cont'd			Unknown (as reputed owner)		The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP Unknown	
01-009	Acquisition of Rights	9511 square metres of agricultural land (east of Manor Way, Holland-on-Sea)	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE	NONE	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights granted by a Deed of Grant dated 12 April 1938 and Deed of Grant dated 12 December 1938) Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX (in respect of rights reserved by a Conveyance dated 30 March 1990)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-009 cont'd						Unknown (in respect of restrictive covenants as contained in Conveyance dated 6 November 1900) Unknown (in respect of restrictive covenants as contained in Conveyance dated 14 November 1929)			
01-010	Acquisition of Rights	course, ponds, public footbridges, drains,	The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP	NONE	County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public bridleway public (BR 2 164) public footpath (FP 3 164)) Frinton Golf Club Limited 1 The Esplanade	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights granted by the Deed of Grant dated 12 December 1938 and Conveyance dated 16 May 1951) Unknown (in respect of covenants contained in a Conveyance dated 24 April 1920)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-011	Acquisition of Rights	2423 square metres of watercourse (Kirby Brook)	Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX (as reputed owner) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (as reputed owner) The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP (as reputed owner) Unknown (as reputed owner)	NONE	Frinton Golf Club Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP Unknown	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Owners	Regulations 2009	ations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
01-012	Acquisition of Rights	82161 square metres of	or Reputed Owners Great Holland Hall Limited	or Reputed Lessees or Tenants Aviadale Limited	or Reputed Occupiers Aviadale Limited	Tendring District Council		
		agricultural land, footbridge and drain (west of Long Lane, Frinton-on-Sea)	27 Old Gloucester Street London WC1N 3AX	71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners)	71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners)	Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of rights granted by a Conveyance dated 30 March 1990)		
					Sentry Limited Unit 7A Hillview Business Park Old Ipswich Road Claydon Ipswich IP6 0AJ (trading as Frinton Farm Partners)	Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929)		
02-001	Acquisition of Rights	16315 square metres of beach, rock armour, sloping masonry, groynes and foreshore at Frinton-On-Sea (south of Frinton Golf Course)	5	NONE	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
02-001A	Acquisition of Rights	rock armour and sloping masonry at Frinton-On-Sea	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE	NONE	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE	NONE		
02-002	Acquisition of Rights	937 square metres of access track (west of Second Avenue, Frinton-on-Sea) and public bridleway (BR 2 164)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as reputed owner) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (as reputed owner)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public bridleway (BR 2 164)) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE	Unknown		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
02-002 cont'd			The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP (as reputed owner) Unknown (as reputed owner)		The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP Unknown			
02-003	Acquisition of Rights	1231 square metres of access track, verges and private car park (west of Second Avenue, Frinton-on-Sea)	The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP	NONE	Frinton Golf Club Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP	Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 GXU (in respect of sewerage apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 GNP (in respect of rights granted by a Transfer dated 5 December 2000)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
02-003 cont'd						Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights granted by the Deed of Grant dated 12 December 1938 and Conveyance dated 16 May 1951) Unknown (in respect of covenants contained in a Conveyance dated 16 March 1905) Unknown (in respect of restrictive covenants contained in a Conveyance dated 16 March 1905) Unknown (in respect of covenants contained in a Conveyance dated 24 April 1920)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
02-004	Acquisition of Rights	(west of Long Lane, Frinton-on-	Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX	Aviadale Limited 71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners)	Aviadale Limited 71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners) Sentry Limited Unit 7A Hillview Business Park Old Ipswich Road Claydon Ipswich IP6 0AJ (trading as Frinton Farm Partners)	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of rights granted by a Conveyance dated 30 March 1990) Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929)		
02-005	Acquisition of Rights	Short Lane, Frinton-on-Sea)	Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX (as reputed owner)	NONE	Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX	Unknown		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-005 cont'd			The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP (as reputed owner) Unknown (as reputed owner)		The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP Unknown				
02-006	Acquisition of Rights	561 square metres of access track, agricultural land and hedgerow (west of Long Lane, Frinton-on-Sea)	Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX	Aviadale Limited 71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners)	Aviadale Limited 71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners) Sentry Limited Unit 7A Hillview Business Park Old Ipswich Road Claydon Ipswich IP6 0AJ (trading as Frinton Farm Partners)	Andrea Woods 9 Beaumont Close WALTON ON THE NAZE CO14 8TX (in respect of a right of way) Andrew Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-006 cont'd						Barry Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way) Courtney Byrne 53 Cornflower Road Jaywick CLACTON-ON-SEA CO15 2SA (in respect of a right of way) Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH CO12 5BY (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-006 cont'd						Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE NAZE CO14 8RN (in respect of a right of way) Louisa Thatcher 104 Salisbury Road Holland-on-Sea CLACTON-ON-SEA CO15 5LT (in respect of a right of way) Mazy King 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way) Michelle Miller 22 Homerton Close CLACTON-ON-SEA CO15 4UJ (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-006 cont'd						Pat Watson 25 Hillcrest CLACTON-ON-SEA CO15 4HU (in respect of a right of way) Tanya Wheeler 2 St. Andrews Close Weeley CLACTON-ON-SEA CO16 9FX (in respect of a right of way) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of rights granted by a Conveyance dated 30 March 1990) Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation					
Land Plans		Sesenpriori di lana	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-006 cont'd						Victoria Oxland 8 Colthorpe Road CLACTON-ON-SEA CO15 4PU (in respect of a right of way) Wendy Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way) Yvonne Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)			
02-007	Acquisition of Rights			Aviadale Limited 71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners)	Aviadale Limited 71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners)	Andrea Woods 9 Beaumont Close WALTON ON THE NAZE CO14 8TX (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-007 cont'd					Market Road Chelmsford Essex CM1 1QH (in respect of public footpaths (FP 1 164 and FP 3 164) and public bridleway (BR 2 164)) Sentry Limited Unit 7A Hillview Business Park Old Ipswich Road Claydon Ipswich IP6 0AJ (trading as Frinton Farm Partners)	Andrew Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way) Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of sewerage apparatus) Barry Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Qualifying persons under Regulation				
Land Plans	Extent of dequisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-007 cont'd						Courtney Byrne 53 Cornflower Road Jaywick CLACTON-ON-SEA CO15 2SA (in respect of a right of way) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH CO12 5BY (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-007 cont'd						Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE NAZE CO14 8RN (in respect of a right of way) Louisa Thatcher 104 Salisbury Road Holland-on-Sea CLACTON-ON-SEA CO15 5LT (in respect of a right of way) Mazy King 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way) Michelle Miller 22 Homerton Close CLACTON-ON-SEA CO15 4UJ (in respect of right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-007 cont'd						Pat Watson 25 Hillcrest CLACTON-ON-SEA CO15 4HU (in respect of a right of way) Tanya Wheeler 2 St. Andrews Close Weeley CLACTON-ON-SEA CO16 9FX (in respect of a right of way) Victoria Oxland 8 Colthorpe Road CLACTON-ON-SEA CO15 4PU (in respect of a right of way) Wendy Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-007 cont'd						Yvonne Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)			
02-008	Acquisition of Rights	access track (west of Long Lane,	Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX	Aviadale Limited 71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners)		Andrea Woods 9 Beaumont Close WALTON ON THE NAZE CO14 8TX (in respect of a right of way) Andrew Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way) Barry Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-008 cont'd						Courtney Byrne 53 Cornflower Road Jaywick CLACTON-ON-SEA CO15 2SA (in respect of a right of way) Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH CO12 5BY (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way) Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE NAZE CO14 8RN (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-008 cont'd						Louisa Thatcher 104 Salisbury Road Holland-on-Sea CLACTON-ON-SEA CO15 5LT (in respect of a right of way) Mazy King 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way) Michelle Miller 22 Homerton Close CLACTON-ON-SEA CO15 4UJ (in respect of a right of way) Pat Watson 25 Hillcrest CLACTON-ON-SEA CO15 4HU (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-008 cont'd						Tanya Wheeler 2 St. Andrews Close Weeley CLACTON-ON-SEA CO16 9FX (in respect of a right of way) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of rights granted by a Conveyance dated 30 March 1990) Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929) Victoria Oxland 8 Colthorpe Road CLACTON-ON-SEA CO15 4PU (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	e Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-008 cont'd						Wendy Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 OUF (in respect of a right of way) Yvonne Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)			
02-009	Acquisition of Rights	agricultural land, access track, hedgerow and drain (south of	Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX	Aviadale Limited 71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners)	Aviadale Limited 71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners) Sentry Limited Unit 7A Hillview Business Park Old Ipswich Road Claydon Ipswich IP6 0AJ (trading as Frinton Farm Partners)	Andrea Woods 9 Beaumont Close WALTON ON THE NAZE CO14 8TX (in respect of a right of way) Andrew Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-009 cont'd						Barry Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way) Courtney Byrne 53 Cornflower Road Jaywick CLACTON-ON-SEA CO15 2SA (in respect of a right of way) Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH CO12 5BY (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-009 cont'd						Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE NAZE CO14 8RN (in respect of a right of way) Louisa Thatcher 104 Salisbury Road Holland-on-Sea CLACTON-ON-SEA CO15 5LT (in respect of a right of way) Mazy King 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way) Michelle Miller 22 Homerton Close CLACTON-ON-SEA CO15 4UJ (in respect of right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-009 cont'd						Pat Watson 25 Hillcrest CLACTON-ON-SEA CO15 4HU (in respect of a right of way) Tanya Wheeler 2 St. Andrews Close Weeley CLACTON-ON-SEA CO16 9FX (in respect of a right of way) Victoria Oxland 8 Colthorpe Road CLACTON-ON-SEA CO15 4PU (in respect of a right of way) Wendy Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-009 cont'd						Yvonne Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)			
02-010	Acquisition of Rights	66097 square metres of agricultural land, hedgerow and access track at Manor Farm (east of Clacton Road, B1032)	Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX	Aviadale Limited 71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners)	Unit 7A Hillview Business Park Old Ipswich Road Claydon Ipswich IP6 0AJ (trading as Frinton Farm Partners)	Andrea Woods 9 Beaumont Close WALTON ON THE NAZE CO14 8TX (in respect of a right of way) Andrew Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way) Barry Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-010 cont'd						Courtney Byrne 53 Cornflower Road Jaywick CLACTON-ON-SEA CO15 2SA (in respect of a right of way) Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH CO12 5BY (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way) Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE NAZE CO14 8RN (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-010 cont'd						Louisa Thatcher 104 Salisbury Road Holland-on-Sea CLACTON-ON-SEA CO15 5LT (in respect of a right of way) Mazy King 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way) Michelle Miller 22 Homerton Close CLACTON-ON-SEA CO15 4UJ (in respect of a right of way) Pat Watson 25 Hillcrest CLACTON-ON-SEA CO15 4HU (in respect of a right of way)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-010 cont'd						Tanya Wheeler 2 St. Andrews Close Weeley CLACTON-ON-SEA CO16 9FX (in respect of a right of way) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of rights granted by a Conveyance dated 30 March 1990) Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929) Victoria Oxland 8 Colthorpe Road CLACTON-ON-SEA CO15 4PU (in respect of a right of way)			

	Land w	which is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (i	nd Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to of Essex and Suffolk	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-010 cont'd						Wendy Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 OUF (in respect of a right of way) Yvonne Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)
03-001	Acquisition of Rights	863 square metres of agricultural land at Manor Farm (east of Clacton Road, B1032)	Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX	Aviadale Limited 71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners)	Aviadale Limited 71 Queen Victoria Street LONDON EC4V 4BE (trading as Frinton Farm Partners) Sentry Limited Unit 7A Hillview Business Park Old Ipswich Road Claydon Ipswich IP6 OAJ (trading as Frinton Farm Partners)	NONE

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on Land Plans	Extent of acquisition or use	uisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning						
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
03-002	Temporary Possession	43609 square metres of agricultural land, copse and hedgerow (east of Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	NONE	Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Adam Charles Brown and Joanna Marie Brown as contained in the charge dated 25 May 2021) Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
03-002 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)				
03-003	Acquisition of Rights	agricultural land and hedgerow (east of Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	NONE	Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Adam Charles Brown and Joanna Marie Brown as contained in the charge dated 25 May 2021)				

	Land w	/hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-003 cont'd						Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)
03-004	Temporary Possession	highway, verges and field	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-004 cont'd			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of subsoil beneath half width of public highway) John George Bellingham Gladwyn House 180 Thorpe Road Kirby Cross FRINTON-ON-SEA Essex CO13 0NH (in respect of subsoil beneath half width of public highway)		Unknown	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
03-004 cont'd			John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 OJU (in respect of subsoil beneath half width of public highway) Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 OJU (in respect of subsoil beneath half width of public highway) Marian Sarah Reynolds Lodge Farm Bungalow Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 OJU (in respect of subsoil beneath half width of great Holland FRINTON-ON-SEA Essex CO13 OJU (in respect of subsoil beneath half width of public highway)							

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
03-004 cont'd			Unknown (as reputed owner)							
03-004A	Temporary Possession	297 square metres of public highway, verges and field accessway (Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of subsoil beneath full width of public highway) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of rights of access) Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-004A cont'd			Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of subsoil beneath full width of public highway) Unknown			Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			
03-005	Acquisition of Rights	92 square metres of agricultural land (east of Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	NONE	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-005 cont'd			Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG		FRINTON-ON-SEA Essex CO13 ONG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Adam Charles Brown and Joanna Marie Brown as contained in the charge dated 25 May 2021) Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
03-006	Temporary Possession	16 square metres of agricultural land and hedgerow (east of Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	NONE	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Adam Charles Brown and Joanna Marie Brown as contained in the charge dated 25 May 2021) Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
03-007	Acquisition of Rights	highway, verges, hedgerow and	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of subsoil beneath full width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)				
			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 ONG (in respect of subsoil beneath full width of public highway)		Unknown	Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Qualifying persons under Regulation				
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
03-007 cont'd			Unknown (as reputed owner)							
03-007A	Acquisition of Rights	1640 square metres of public highway, verges, hedgerow and access splay (Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of subsoil beneath full width of public highway) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Besenption of failed	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-007A cont'd			Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of subsoil beneath full width of public highway) Unknown			Unknown			
03-008	Temporary Possession	agricultural land and hedgerow (west of Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG		Great Holland FRINTON-ON-SEA Essex CO13 0NG	John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
03-008 cont'd			Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG		Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)				
03-009	Temporary Possession	740 square metres of public highway and verges (Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of subsoil beneath full width of public highway) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-009 cont'd			Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX (in respect of subsoil beneath half width of public highway) Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of subsoil beneath full width of public highway) Unknown (as reputed owner)						
03-010	Temporary Possession	50 square metres of agricultural land, verge and hedgerow (Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	NONE	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	NONE			

	Land v	vhich is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-010 cont'd			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG		Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	
03-011	Acquisition of Rights	12346 square metres of agricultural land and hedgerow (west of Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 ONG	NONE	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-011 cont'd			Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG		Essex CO13 ONG	John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 OJU (in respect of rights reserved as contained in a Deed dated 27 March 1951) Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland PRINTON-ON-SEA Essex CO13 OJU (in respect of rights reserved as contained in a Deed dated 27 March 1951)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-012	Temporary Possession	10870 square metres of agricultural land and hedgerows (west of Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	NONE	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			
			Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG			John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)			

	Land v	vhich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii	d Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation		
Land Plans		Sconpion of faire	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-012 cont'd						Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 OJU (in respect of rights reserved as contained in a Deed dated 27 March 1951)
03-013	Acquisition of Rights	agricultural land and hedgerow	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 ONG	NONE	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 OJU (in respect of rights reserved as contained in a Deed dated 27 March 1951)

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-013 cont'd			Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG		Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)
03-014	Acquisition of Rights	56162 square metres of agricultural land hedgerow (Little Clacton Road, Great Holland)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	NONE	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights as contained in a Transfer dated 4 February 2019 and in a Conveyance dated 22 December 1982)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-014 cont'd			Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG		Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Adam Charles Brown and Joanna Marie Brown as contained in a Charge dated 25 May 2021) Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respects of rights contained in a Transfer dated 4 February 2019) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
03-014 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)				
03-015	Acquisition of Rights	3759 square metres of agricultural land, access track, hardstanding, hedgerow, access splay and public footpath (FP 5 164) (Dairy House Farm, Little Clacton Road, Great Holland)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG Voodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG	NONE	FRINTON-ON-SEA Essex CO13 ONG	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights as contained in a Transfer dated 4 February 2019 and in a Conveyance dated 22 December 1982) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Adam Charles Brown and Joanna Marie Brown as contained in a Charge dated 25 May 2021)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
03-015 cont'd					Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 ONG	Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respects of rights contained in a Transfer dated 4 February 2019) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
03-016	Acquisition of Rights	track and public footpath (FP 7 164 and FP 10 164) (north of Little Clacton Road, Great Holland)	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Brian Lawrence)	NONE	Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpaths (FP 7 164 and FP 10 164))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-016 cont'd			Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Brian Lawrence) Unknown (in respect of mines and minerals) Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX						
03-017	Acquisition of Rights	27 square metres of access track (north of Little Clacton Road, Great Holland) and public footpath (FP 11 164)	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN (as reputed owner)	NONE	Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX	Brenda Jane Gibson-Wynes Flat 11 Linkswood Compton Place Road EASTBOURNE BN21 1EE (in respect of covenants by a Transfer dated 5 November 1982)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-017 cont'd			Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB (as reputed owner)		Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 10 164))	Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access)			
			Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence)			Unknown (in respect of rights granted by a Deed 23 August 1968)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-017 cont'd			Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence) Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX (as reputed owner)						
03-018	Acquisition of Rights	249 square metres of access track (north of Little Clacton Road, Great Holland) and public footpath (FP 10 164)	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN (as reputed owner)	NONE	Little Clacton	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-018 cont'd			Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB (as reputed owner) Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence)		Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 10 164))	Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans		besen priori or land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-018 cont'd			Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence) Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX (as reputed owner)						
04-001	Acquisition of Rights	23 square metres of hedgerow and garden (Shorelmist Cottage)	June Hilda Varley Shorelmist Cottage Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0ET	NONE	Little Clacton Road Great Holland FRINTON-ON-SEA Essex	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of street apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
04-001 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)				
04-002	Acquisition of Rights	highway, verges, access splay (Little Clacton Road) and public footpath (FP 6 164)	Woodthorpe House		County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Essex County Council County Hall Market Road Chelmsford Essex	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of gas apparatus)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-002 cont'd			David William Rampling Great Holland Mill Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU (in respect of subsoil beneath half width of public highway) Emma Jane Parker Seawinds Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU (in respect of subsoil beneath half width of public highway) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)		Unknown	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of street furniture) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-002 cont'd			Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN (in respect of subsoil beneath half width of public highway) Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB (in respect of subsoil beneath half width of public highway) Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of subsoil beneath half width of public highway)			Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-002 cont'd			June Hilda Varley Shorelmist Cottage Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0ET (in respect of subsoil beneath half width of public highway) Louis Parker Seawinds Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU (in respect of subsoil beneath half width of public highway) Martin Terence Clary Dankeer Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU (in respect of subsoil beneath half width of great Holland FRINTON-ON-SEA Essex CO13 0EU (in respect of subsoil beneath half width of public highway)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-002 cont'd			Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Brian Lawrence) (in respect of subsoil beneath half width of public highway) Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Brian Lawrence) (in respect of subsoil beneath half width of public highway)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-002 cont'd			Tracey Jane Clary Dankeer Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU (in respect of subsoil beneath half width of public highway) Tracy Jane Rampling Great Holland Mill Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-002 cont'd			Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX (in respect of subsoil beneath half width of public highway)						
04-003	Acquisition of Rights	96 square metres of agricultural land (east of Mill Lane, Little Clacton Road)	David William Rampling Great Holland Mill Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU (as reputed owner) Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN (as reputed owner)	NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX David William Rampling Great Holland Mill Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-003 cont'd			Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB (as reputed owner) Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Lawrence) Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Lawrence) Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Lawrence)		Tracy Jane Rampling Great Holland Mill Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-003 cont'd			Tracy Jane Rampling Great Holland Mill Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU (as reputed owner) Unknown (as reputed owner) Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX (as reputed owner)						
04-004	Acquisition of Rights	agricultural land (north of Little	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN	NONE	Holland Road Little Clacton CLACTON-ON-SEA	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-004 cont'd			Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Brian Wilfred Lawrence) Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Essex CO16 9RX (as executor of the estate for the late Brian Wilfred Lawrence)			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of street apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-004 cont'd			Unknown (in respect of mines and minerals) Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX			
04-005	Temporary Possession	597 square metres of agricultural land and track (north of Little Clacton Road, Great Holland)	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB	NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX	NONE

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans	Land Plans		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
04-005 cont'd			Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Brian Wilfred Lawrence) Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Brian Wilfred Lawrence) Unknown (in respect of mines and minerals)					

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-005 cont'd			Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX			
04-006	Temporary Possession	1263 square metres of agricultural land, hedgerow (east of Great Holland Mill, Little Clacton Road) and public footpaths (FP 38 164 and FP 11 164)		NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpaths (FP 38 164 and FP 11 164))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

	Land w	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-007	Acquisition of Rights	30147 square metres of agricultural land, hedgerow (east of Great Holland Mill, Little Clacton Road) and public footpaths (FP 38 164 and FP 11 164)		NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpaths (FP 38 164 and FP 11 164))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
04-008	Temporary Possession	1479 square metres of agricultural land (north of Little Clacton Road, Great Holland)	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN	NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-008 cont'd			Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Brian Wilfred Lawrence) Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Brian Wilfred Lawrence)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-008 cont'd			Unknown (in respect of mines and minerals) Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX						
04-009	Temporary Possession	156 square metres of agricultural land (east of Great Holland Mill, Little Clacton Road) and public footpath (FP 38 164)	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN (as reputed owner) Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB (as reputed owner)	NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 38 164))	NONE			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
04-009 cont'd			Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence) Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence)					

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans	Land Plans		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-009 cont'd			Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX (as reputed owner)						
04-010	Temporary Possession	187 square metres of agricultural land, hedgerow (east of Great Holland Mill, Little Clacton Road) and public footpath (FP 38 164)	Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX	NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 38 164))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-011	Temporary Possession	19 square metres of agricultural land (east of Great Holland Mill, Little Clacton Road) and public footpath (FP 38 164)	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN (as reputed owner) Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB (as reputed owner) Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence)	NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 38 164)) Unknown	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
04-011 cont'd			Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence) Unknown (as reputed owner) Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX (as reputed owner)					

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-012	Temporary Possession	799 square metres of agricultural land (east of Great Holland Mill, Little Clacton Road)	Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX	NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX	NONE
04-013	Temporary Possession	64 square metres of agricultural land (east of Great Holland Mill, Little Clacton Road)	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN (as reputed owner)	NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX	NONE

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-013 cont'd			Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB (as reputed owner) Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-013 cont'd			Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence) Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX (as reputed owner)						
04-014	Acquisition of Rights	agricultural land (east of Great	Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX		A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX	NONE			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-014 cont'd			Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX						
04-015	Acquisition of Rights	3 square metres of agricultural land (east of Great Holland Mill, Little Clacton Road)	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN (as reputed owner) Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB (as reputed owner)	NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Unknown	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-015 cont'd			Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence) Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence) Unknown (as reputed owner)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
04-015 cont'd			Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX (as reputed owner)							
04-016	Acquisition of Rights	185 square metres of access track (north of Little Clacton Road, Great Holland) and public footpath (FP 10 164)	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN (as reputed owner) Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB (as reputed owner)	NONE	Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Essex County Council	Brenda Jane Gibson-Wynes Flat 11 Linkswood Compton Place Road EASTBOURNE BN21 1EE (in respect of covenants by a Transfer dated 5 November 1982) Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-016 cont'd			Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence) Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence) Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX (as reputed owner)			Unknown (in respect of rights granted by a Deed 23 August 1968)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
04-017	Acquisition of Rights	1893 square metres of agricultural land, access track and public footpaths (FP 10 164, FP 38 164 and FP 11 164) (east of Great Holland Mill, Little Clacton Road)	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN (as reputed owner) Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB (as reputed owner) Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence)	NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpaths (FP 10 164, FP 11 164 and FP 38 164))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-017 cont'd			Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence) Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX (as reputed owner)						
04-018	Acquisition of Rights	track (west of Pork Lane, Great	Helen Peirson The Pavilion High Street Castle Camps CAMBRIDGE CB21 4SN (as reputed owner)	NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-018 cont'd			Janet Philp New House Farm Kirby Hall Road Castle Hedingham Halstead CO9 3EB (as reputed owner) Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence)		Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 10 164)) Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-018 cont'd			Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner, as executor of the estate for the late Brian Wilfred Lawrence and the late Pamela Lawrence) Unknown (as reputed owner) Wendy Harwood Woodlands Norwich Road Long Stratton NORWICH NR15 2PX (as reputed owner)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-019	Acquisition of Rights	552 square metres of agricultural land, copse, hedgerow (west of Pork Lane, Great Holland) and public footpath (FP 10 164)	Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX	NONE	A Lawrence & Sons Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 10 164))	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Nicholas David Lawrence and Samuel William Lawrence as contained in a Charge dated 24 November 2000) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of statutory rights as contained in a Deed dated 19 December 1966) Unknown (in respect of right to enter to maintain a water main and in respect of restrictive covenants as stated in a Deed dated 16 July 1968)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
04-020	Acquisition of Rights	38061 square metres of agricultural land, copse and hedgerow (west of Pork Lane, Great Holland)	Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX	NONE		Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Nicholas David Lawrence and Samuel William Lawrence as contained in a Charge dated 24 November 2000) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of gas apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of statutory rights as contained in a Deed dated 19 December 1966)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
04-020 cont'd						Unknown (in respect of right to enter to maintain a water main and in respect of restrictive covenants as stated in a Deed dated 16 July 1968)				
05-001	Acquisition of Rights	agricultural land and hedgerow (west of Pork Lane, Great Holland)	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (as reputed owner) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (as reputed owner) Unknown (as reputed owner)		John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Unknown	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-002	Acquisition of Rights	brook and access track at Birch	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER	NONE	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for John Hutley and Lorna Marion Hutley as contained in a Charge dated 21 February 2003) Unknown (in respect of rights as stated in Conveyance dated 31 March 1982)			
05-003	Acquisition of Rights	129 square metres of agricultural land and hedgerow (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (as reputed owner)	NONE	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-003 cont'd			John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (as reputed owner) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (as reputed owner) Unknown (as reputed owner)		John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Unknown	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-004	Acquisition of Rights	track (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ	NONE	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of right of access as stated in a Conveyance dated 16 July 1968) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of right of access as stated in a Conveyance dated 16 July 1968) Unknown (in respect of rights reserved as contained in a Conveyance dated 16th July 1968)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(:	Category 2 Qualifying persons under Regulation						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
05-005	Acquisition of Rights	track and access splay (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ	NONE	Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Lorna Marion Hutley	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of right of access as stated in a Conveyance dated 16 July 1968) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of right of access as stated in a Conveyance dated 16 July 1968)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
05-006	Acquisition of Rights	3 square metres of agricultural land and access splay (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (as reputed owner) Nicholas David Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner) Samuel William Lawrence Reedlands Farm Holland Road Little Clacton CLACTON-ON-SEA Essex CO16 9RX (as reputed owner) Unknown (as reputed owner)	NONE	Tamblin Way HATFIELD Hertfordshire AL10 9EZ Unknown	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of rights of access) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of rights of access) Lorna On-SEA Essex CO13 0ER (in respect of rights of access) Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-007	Temporary Possession	agricultural land, access track, brook and hedgerow (west of Pork Lane, Great Holland)	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (as reputed owner) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (as reputed owner) Unknown (as reputed owner)	NONE	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Unknown	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown			
05-008	Temporary Possession	agricultural land, brook and	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-008 cont'd						HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013) Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y 5AD (as beneficiary in respect of an Option Agreement) Unknown (in respect of rights granted by a Conveyance dated 29 September 1954) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998) Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-009	Acquisition of Rights	agricultural land, brook and	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (as reputed owner) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (as reputed owner) Unknown (as reputed owner)	NONE	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Unknown	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown			
05-010	Temporary Possession	Ŭ ,	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (as reputed owner)	NONE	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement)			

	Land v	vhich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to of Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-010 cont'd			Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (as reputed owner) Unknown (as reputed owner)		Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Unknown	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown
05-011	Temporary Possession	25 square metres of agricultural land and brook (west of Pork Lane, Great Holland)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013) Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y 5AD (as beneficiary in respect of an Option Agreement)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-011 cont'd						Unknown (in respect of rights granted by a Conveyance dated 29 September 1954)			
						Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998) Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)			
05-012	Acquisition of Rights	889 square metres of agricultural land and brook (south of Thorpe Park Lane, Thorpe-le-Soken)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-012 cont'd						Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y 5AD (as beneficiary in respect of an Option Agreement) Unknown (in respect of rights granted by a Conveyance dated 29 September 1954) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998) Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)			
05-013	Acquisition of Rights		Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-013 cont'd						HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013) Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y 5AD (as beneficiary in respect of an Option Agreement) Unknown (in respect of rights granted by a Conveyance dated 29 September 1954) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998) Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)			

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-014	Acquisition of Rights		Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (as reputed owner) Unknown (as reputed owner)	NONE	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW Unknown	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown
05-015	Acquisition of Rights	1021 square metres of agricultural land (south of Thorpe Park Lane, Thorpe-le- Soken)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 OHN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 OHN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013) Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y SAD (as beneficiary in respect of an Option Agreement)

	Land w	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-015 cont'd						Unknown (in respect of rights granted by a Conveyance dated 29 September 1954)
						Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)
						Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)
05-016	Acquisition of Rights	103 square metres of railway (Thorpe-le-Soken and Kirby Cross) access track and scrubland (south of Thorpe Park Lane)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (as reputed owner)	NONE	Thorpe Park Lane Thorpe-Le-Soken	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus)
			Unknown (as reputed owner)		Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW	Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-016 cont'd			Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN (as reputed owner, in respect of access track)		Unknown				
05-017	Acquisition of Rights	5373 square metres of access track (south of Thorpe Park Lane)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	CLACTON-ON-SEA Essex CO16 0HN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus)			

# $\lor \Xi \lor$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-017 cont'd						Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y 5AD (as beneficiary in respect of an Option Agreement) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)			
05-018	Acquisition of Rights	408 square metres of private road (Thorpe Park Lane)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	Andrzej Tomasz Wiecek 1 Thorpe Park Cottages Thorpe Park Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 OHN (in respect of right of access) Annis Roberts Thorpe Park House Thorpe Park Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 OHN (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
05-018 cont'd						Benjamin Mark Worrallo 3 Thorpe Park Cottages Thorpe Park Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access) Georgios Tsaousellis 4 Thorpe Park Cottages Thorpe Park Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
05-018 cont'd						John Whiten 2 Thorpe Park Cottages Thorpe Park Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access) Justyna Magdalena Wiecek 1 Thorpe Park Cottages Thorpe Park Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access) Rhiannon Wheeler 4 Thorpe Park Lane Thorpe Park Lane Thorpe Park Lane Thorpe Park Lane Thorpe Park Lane Thorpe Park Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-018 cont'd						Sam Worrallo 3 Thorpe Park Cottages Thorpe Park Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access) Shirley Whiten 2 Thorpe Park Cottages Thorpe Park Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access) Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y 5AD (as beneficiary in respect of an Option Agreement) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
05-019	Acquisition of Rights	56801 square metres of agricultural land (south of Thorpe Park Lane)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)			
						HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013)			
						Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-019 cont'd						Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y 5AD (as beneficiary in respect of an Option Agreement) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)			
05-020	Acquisition of Rights	14516 square metres of agricultural land (to west of Pork Lane)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013)			

	Land v	vhich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-021	Temporary Possession	1349 square metres of agricultural land (to west of Pork Lane)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 OHN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013)
05-022	Temporary Possession	61 square metres of agricultural land (west of Pork Lane, Great Holland)	Unknown (as reputed owner) Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN (as reputed owner)	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN Unknown	Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use		Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Fields			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
05-023	Temporary Possession	3249 square metres of agricultural land and hedgerow (south of Thorpe Road, B1035)		NONE		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013) Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y 5AD (as beneficiary in respect of an Option Agreement) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-024	Acquisition of Rights	38017 square metres of agricultural land, woodland (Parklane Grove) and hedgerow (south of Thorpe Road, B1035)		NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y SAD (as beneficiary in respect of an Option Agreement) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-025	Acquisition of Rights	232 square metres of agricultural land and hedgerow (south of Thorpe Road, B1035)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 OHN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013) Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y 5AD (as beneficiary in respect of an Option Agreement) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)			
05-026	Acquisition of Rights	Farm) (west of Pork Lane)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 OHN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
05-026 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			
06-001	Acquisition of Rights	3067 square metres of public highway, verges and lay-by (Thorpe Road, B1033)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of subsoil beneath half width of public highway) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Chelmsford	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
06-001 cont'd			J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW (in respect of subsoil beneath half width of public highway) Sheik Kemal Kadar The Firs Thorpe Road Kirby Cross FRINTON-ON-SEA Essex CO13 0NJ (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

# $\lor \Xi \lor$

	Land v	which is proposed to be subject to	BOOK OF RI c (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-001 cont'd			Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN (in respect of subsoil beneath half width of public highway)			
06-002	Temporary Possession	86 square metres of agricultural land and hedgerow (north of Thorpe Road, B1035)		NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013)
06-003	Acquisition of Rights	65 square metres of agricultural land (north of Thorpe Road, B1035)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013)

	Land v	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-004	Acquisition of Rights	17 square metres of agricultural land (north of Thorpe Road, B1035)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013)
06-005	Temporary Possession	23298 square metres of agricultural land and hedgerow (south of Thorpe Road, B1035)	•	NONE	James Roberts Thorpe Park Farm Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013) Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y 5AD (as beneficiary in respect of an Option Agreement) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Hans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
06-006	Acquisition of Rights	249 square metres of agricultural land (south of Thorpe Road, B1035)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013) Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y 5AD (as beneficiary in respect of an Option Agreement) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)			
06-007	Temporary Possession	224 square metres of agricultural land (west of Thorpe Road, B1033)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-007 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Thorpe Park Solar Farm Limited Stirling Square 5-7 Carlton Gardens LONDON SW1Y 5AD (as beneficiary in respect of an Option Agreement) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)				
06-008	Temporary Possession		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of subsoil beneath half width of public highway)	NONE	County Hall Market Road Chelmsford	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
06-008 cont'd			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown (as reputed owner) Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 OHN (in respect of subsoil beneath half width of public highway)			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			
06-009	Temporary Possession	268 square metres of agricultural land (east of Thorpe Road, B1033)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
06-010	Acquisition of Rights	13456 square metres of agricultural land (north of Thorpe Road, B1035)	Valerie Joan Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	NONE	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0HN	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Valerie Joan Roberts as contained in a Charge dated 17 June 2013)			
06-011	Acquisition of Rights	29988 square metres of agricultural land (south of Sneating Hall Lane, B1034)	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for J B Fairley & Son Limited as contained in a Charge dated 22 December 1998)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
06-011 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			
06-012	Acquisition of Rights		J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for J B Fairley & Son Limited as contained in a Charge dated 22 December 1998) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-013	Acquisition of Rights	196 square metres of agricultural land, verge and drain (Sneating Hall Lane, B1034)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for J B Fairley & Son Limited as contained in a Charge dated 22 December 1998)
06-014	Acquisition of Rights	2785 square metres of public highway, verges, hardstanding and drain (Sneating Hall Lane, B1034)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

	Land v	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-014 cont'd			J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW (in respect of subsoil beneath full width of public highway) Unknown (as reputed owner)		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown
06-015	Acquisition of Rights	263 square metres of agricultural land and verge (Sneating Hall Lane, B1034)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for J B Fairley & Son Limited as contained in a Charge dated 22 December 1998) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
06-016	Acquisition of Rights	1057 square metres of agricultural land (north of Sneating Hall Lane, B1034)	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for J B Fairley & Son Limited as contained in a Charge dated 22 December 1998) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
06-017	Acquisition of Rights	53667 square metres of agricultural land (north of Sneating Hall Lane, B1034)	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for J B Fairley & Son Limited as contained in a Charge dated 22 December 1998) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			
06-018	Acquisition of Rights	2430 square metres of public highway, verges and access splay (Damants Farm Lane, Thorpe-le-Soken)	Bobby Swift Damonts Farm Damants Farm Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 ONP (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
06-018 cont'd			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW (in respect of subsoil beneath half width of public highway) Penelope Swift Damonts Farm Damants Farm Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0NP (in respect of subsoil beneath half width of public highway)		Unknown	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
06-018 cont'd			Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)						
07-001	Acquisition of Rights	51 square metres of agricultural land (south of Walton Road)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Main Road Ford End CHELMSFORD Essex CM3 1LN	Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
07-002	Temporary Possession	agriculural land and drain (south of Walton Road, Thorpe- Le-Soken)	Ford End CHELMSFORD Essex CM3 1LN	NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			
07-003	Temporary Possession	agricultural land and private	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
07-004	Acquisition of Rights	agricultural land, hedgerow and private access track (south of Walton Road)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Ford End CHELMSFORD Essex CM3 1LN	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
07-005	Temporary Possession	agricultural land (south of Walton Road, Thorpe-Le- Soken)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE		Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			
07-006	Temporary Possession	agricultural land and hedgerow (south of Walton Road, Thorpe- Le-Soken)		NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
07-007	Acquisition of Rights	woodland (south of Walton Road, Thorpe-Le-Soken)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019) M Scott Property Group Limited Suite 5 Oyster House Severalls Lane Colchester CO4 9PD (in respect of a restriction dated 22 March 2021)			
07-008	Acquisition of Rights	1272 square metres of agricultural land and hedgerow (south of Walton Road)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
07-008 cont'd						Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			
07-009	Temporary Possession	5430 square metres of agricultural land and hedgerow (south of Walton Road, Thorpe- Le-Soken)		NONE	Essex CM3 1LN	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			

	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-010	Temporary Possession	agricultural land and hedgerow	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)
07-011	Acquisition of Rights	128000 square metres of agricultural land and hedgerow (south of Walton Road) and public footpath (FP 13 180)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 13 180))	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
07-011 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019) M Scott Property Group Limited Suite 5 Oyster House Severalls Lane Colchester CO4 9PD (in respect of a restriction dated 22 March 2021)		

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-012	Temporary Possession	2460 square metres of agricultural land and hedgerow (east of Landermere Road, B1414)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)
07-013	Acquisition of Rights	3120 square metres of public highway (Landermere Road, B1414)	Anita Higgs 82 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 ONF (in respect of subsoil beneath half width of public highway) Bellway Homes Limited Woolsington House International Drive Woolsington NEWCASTLE UPON TYNE NE13 8BF (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of gas apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
07-013 cont'd			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Ian Douglas Garner 80 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NF (in respect of subsoil beneath half width of public highway) James Leonard George Higgs 82 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NF (in respect of subsoil beneath half width of public highway)			Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
07-013 cont'd			Mandy Kathleen Garner 80 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 ONF (in respect of subsoil beneath half width of public highway) Michael George Robert Goosetree 101 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 ONG (in respect of subsoil beneath half width of public highway) Pauline Jarrold 99 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 ONG (in respect of subsoil beneath half width of public highway)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
07-013 cont'd			Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)						
08-001	Temporary Possession		Anita Higgs 82 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 ONF (as reputed owner) Bellway Homes Limited Woolsington House International Drive Woolsington NEWCASTLE UPON TYNE NE13 8BF (as reputed owner)	NONE		Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of gas apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-001 cont'd			Ian Douglas Garner 80 Landermere Road Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0NF (as reputed owner) James Leonard George Higgs 82 Landermere Road Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0NF (as reputed owner) Mandy Kathleen Garner 80 Landermere Road Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0NF (as reputed owner) Unknown (as reputed owner)		Ian Douglas Garner 80 Landermere Road Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0NF James Leonard George Higgs 82 Landermere Road Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0NF Mandy Kathleen Garner 80 Landermere Road Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0NF Unknown	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-002	Acquisition of Rights	83 square metres of agricultural land and hedgerow (south of Walton Road)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			
08-003	Temporary Possession	33 square metres of agricultural land and hedgerow (east of Landermere Road, B1414)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			
08-004	Temporary Possession	70 square metres of hedgerow, verge and access splay (Landermere Road, B1414)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans Extent of	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
		beserption of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-004 cont'd			Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN (in respect of subsoil beneath full width of public highway) Unknown (as reputed owner)		Unknown				
08-005		agricultural land and hedgerow	, , , , , , , , , , , , , , , , , , ,	NONE	Main Road Ford End CHELMSFORD Essex	Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-006	Acquisition of Rights	agricultural land, access splay and public footpath (FP 7 180) (west of Landermere Road)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 7 180)) Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			
08-007	Temporary Possession	agricultural land, hedgerow (east of Landermere Road, B1414) and public footpath (FP 7 180)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 7 180))	Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			

	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-007 cont'd					Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	
08-008	Acquisition of Rights	agricultural land and hedgerow (west of Landermere Road)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Ford End CHELMSFORD Essex CM3 1LN	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)

	Land w	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
	Acquisition of Rights	1029 square metres of agricultural land and public footpath (FP 7 180)	Andrew William Bacon Green Lane Farm Colchester Road Weeley CLACTON-ON-SEA Essex CO16 9AD	NONE	Andrew William Bacon Green Lane Farm Colchester Road Weeley CLACTON-ON-SEA Essex CO16 9AD Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 7 180))	Unknown (in respect of covenants contained in various Conveyances)
08-010	Acquisition of Rights	418 square metres of private road and verges (Lonsdale Road, Thorpe-Le-Soken)	Andrew William Bacon Green Lane Farm Colchester Road Weeley CLACTON-ON-SEA Essex CO16 9AD (as reputed owner)	NONE	Andrew William Bacon Green Lane Farm Colchester Road Weeley CLACTON-ON-SEA Essex CO16 9AD	Andrew William Bacon Green Lane Farm Colchester Road Weeley CLACTON-ON-SEA Essex CO16 9AD (in respect of rights of way and maintenance)

	Land wh	ich is proposed to be subject	BOOK OF R to: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-010 cont'd			Benjamin Furness 7 Crownfields Crown Street Dedham COLCHESTER CO7 6AT (as reputed owner)		Benjamin Furness 7 Crownfields Crown Street Dedham COLCHESTER CO7 6AT	Benjamin Furness 7 Crownfields Crown Street Dedham COLCHESTER CO7 6AT (in respect of rights of way and maintenance)
			Brendan Malachy Gormley Gunfleet Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (as reputed owner)		Brendan Malachy Gormley Gunfleet Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF	Brendan Malachy Gormley Gunfleet Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance)
			Colin Ernest Anson Framble Barn Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (as reputed owner)		Colin Ernest Anson Framble Barn Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF	Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of gas apparatus)

	Land wi	hich is proposed to be subject	BOOK OF R to: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to f Essex and Suffolk	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-010 cont'd			Delicia Maria Ransom The Old Cottage The Street Copdock IPSWICH IP8 3HS (as reputed owner) Dominic Furness Porttiniemntie 21 Hameenlinna 1320 FINLAND (as reputed owner)		Delicia Maria Ransom The Old Cottage The Street Copdock IPSWICH IP8 3HS Dominic Furness Porttiniemntie 21 Hameenlinna 1320 FINLAND	Colin Ernest Anson Framble Barn Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance) Delicia Maria Ransom The Old Cottage The Street Copdock IPSWICH IP8 3HS (in respect of rights of way and maintenance)
			Gwendolyn Rose Batley 87 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LW (as reputed owner)		Gwendolyn Rose Batley 87 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LW	Dominic Furness Porttiniemntie 21 Hameenlinna 1320 FINLAND (in respect of rights of way and maintenance)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
08-010			Jane Elizabeth Anson		Jane Elizabeth Anson	Eastern Power Networks PLC			
cont'd			Framble Barn		Framble Barn	Newington House			
			Lonsdale Road		Lonsdale Road	237 Southwark Bridge Road			
			Thorpe-le-Soken		Thorpe-le-Soken	LONDON			
			CLACTON-ON-SEA			SE1 6NP			
			CO16 0LF (as reputed owner)		CO16 0LF	(in respect of electricity apparatus)			
			John Frederick Peirson		John Frederick Peirson	Gwendolyn Rose Batley			
			Blacksmiths Farm		Blacksmiths Farm	87 Landermere Road			
			Harwich Road		Harwich Road	Thorpe-le-Soken			
			Beaumont		Beaumont	CLACTON-ON-SEA			
			CLACTON-ON-SEA		CLACTON-ON-SEA	CO16 0LW			
			CO16 0AS (as reputed owner)		CO16 0AS	(in respect of rights of way and maintenance)			
			Julian Furness		Julian Furness	Jane Elizabeth Anson			
			Cambridge House		Cambridge House	Framble Barn			
			Amberfield Drive		Amberfield Drive	Lonsdale Road			
			Nacton		Nacton	Thorpe-le-Soken			
			IPSWICH		IPSWICH	CLACTON-ON-SEA			
			IP10 0GQ		IP10 0GQ	CO16 0LF			
			(as reputed owner)			(in respect of rights of way and			
						maintenance)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	t of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-010 cont'd			Linda Draper Hamford Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (as reputed owner)		Linda Draper Hamford Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF	John Frederick Peirson Blacksmiths Farm Harwich Road Beaumont CLACTON-ON-SEA CO16 0AS (in respect of rights of way and maintenance)			
			Margaret Shona Gormley Gunfleet Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (as reputed owner)		Margaret Shona Gormley Gunfleet Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF	Julian Furness Cambridge House Amberfield Drive Nacton IPSWICH IP10 0GQ (in respect of rights of way and maintenance)			
			Mark Terry Sangwine 87 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LW (as reputed owner)		Mark Terry Sangwine 87 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LW	Linda Draper Hamford Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
08-010 cont'd			or Reputed Owners Michael David Pollard 89 Landermere Road	or Reputed Lessees or Tenants	or Reputed Occupiers Michael David Pollard 89 Landermere Road	Margaret Shona Gormley Gunfleet			
cont u			Thorpe-le-Soken CLACTON-ON-SEA		Thorpe-le-Soken CLACTON-ON-SEA	Lonsdale Road Thorpe-le-Soken			
			CO16 0NA (as reputed owner)		CO16 0NA	CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance)			
			Michael William Gay Trinity Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (as reputed owner)			Mark Terry Sangwine 87 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LW (in respect of rights of way and maintenance)			
			Orwell Housing Association Limited Crane Hill Lodge 325 London Road IPSWICH Suffolk IP2 OBE (as reputed owner)			Maintenance) Michael David Pollard 89 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NA (in respect of rights of way and maintenance)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of failed	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-010 cont'd			Principal Homes Limited Oak House Church Lane Great Holland FRINTON-ON-SEA Essex CO13 0JS (as reputed owner) Sharon Susan Gay Trinity Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (as reputed owner) Simon Stone Jimilda Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (as reputed owner)		Principal Homes Limited Oak House Church Lane Great Holland FRINTON-ON-SEA Essex CO13 0JS Sharon Susan Gay Trinity Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF Simon Stone Jimilda Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF	Michael William Gay Trinity Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Orwell Housing Association Limited Crane Hill Lodge 325 London Road IPSWICH Suffolk IP2 0BE (in respect of rights of way and maintenance)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	nt of acquisition or use Description of land ,	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-010 cont'd			Susan Mary Pollard 89 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NA (as reputed owner)		Susan Mary Pollard 89 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NA	Principal Homes Limited Oak House Church Lane Great Holland FRINTON-ON-SEA Essex CO13 0JS (in respect of rights of way and maintenance)			
			Unknown (as reputed owner)		Unknown	Sharon Susan Gay Trinity Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance) Simon Stone Jimilda Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-010 cont'd						Susan Mary Pollard 89 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NA (in respect of rights of way and maintenance) Unknown			
08-011	Acquisition of Rights	17889 square metres of agricultural land (west of Landermere Road)	Andrew William Bacon Green Lane Farm Colchester Road Weeley CLACTON-ON-SEA Essex CO16 9AD	NONE	Andrew William Bacon Green Lane Farm Colchester Road Weeley CLACTON-ON-SEA Essex CO16 9AD	Unknown (in respect of covenants contained in various Conveyances)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-012	Acquisition of Rights	26690 square metres of agricultural land (south of Golden Lane)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			
08-013	Acquisition of Rights	31370 square metres of agricultural land and hedgerows (south of Golden Lane)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 OLE (in respect of rights reserved by the Transfer dated 21 July 2010)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
08-013 cont'd						Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019) Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by a Transfer dated 21 July 2010)		
08-014	Acquisition of Rights	agricultural land and hedgerow (south of Golden Lane)	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE	NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-014 cont'd			Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE			Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN (in respect of rights granted by a Deed dated 11 August 1967) Unknown (in respect of rights contained in a Conveyance dated 18 February 1952)			
08-015	Acquisition of Rights	405 square metres of agricultural land (east of Golden Lane)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 OLE (in respect of rights reserved by the Transfer dated 21 July 2010)			

## $\lor \Xi \lor$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
08-015 cont'd						Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019) Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by a Transfer dated 21 July 2010)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-016	Acquisition of Rights	176 square metres of agricultural land (east of Golden Lane)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE		Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 OLE (in respect of rights reserved by the Transfer dated 21 July 2010) Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-016 cont'd						Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by a Transfer dated 21 July 2010)			
08-017	Acquisition of Rights	56 square metres of agricultural land (east of Golden Lane)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 OLE (in respect of rights reserved by the Transfer dated 21 July 2010)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
08-017 cont'd						Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019) Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by a Transfer dated 21 July 2010)		

	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-018	Acquisition of Rights	1801 square metres of public highway and verges (Golden Lane)	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 OLE (in respect of subsoil beneath half width of public highway) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Martin Roy Bowers Charity Cottage Thorpe Road Preston St. Mary SUDBURY CO10 9NA (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-018 cont'd			Sharon Joy Bowers Charity Cottage Thorpe Road Preston St. Mary SUDBURY CO10 9NA (in respect of subsoil beneath half width of public highway) Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of subsoil beneath half width of public highway) Spencer Leigh Brown Brown Roofing Valley Barns Golden Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0LE (in respect of subsoil beneath half width of public highway)						

	Land v	vhich is proposed to be subject t	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-018 cont'd			Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)			
08-019	Acquisition of Rights	34286 square metres of agricultural land, copse and public footpath (FP 3 180) (west of Golden Lane)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	NONE	Market Road Chelmsford Essex	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-019 cont'd					Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN	Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)			
08-020	Acquisition of Rights	4361 square metres of paddock and copse (east of Tendring Road, B1035)	Frank Leach Warmans Gate Tendring Road Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AA	NONE	Frank Leach Warmans Gate Tendring Road Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AA	Julie Wendy Thompson 92 Station Road CLACTON-ON-SEA Essex CO15 1SG (in respect of rights granted by a Transfer dated 28 May 2015)			
08-021	Acquisition of Rights	777 square metres of paddock, hedgerow and public footpath (FP 3 180) (east of Tendring Road, B1035)	·	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 3 180))	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
08-021 cont'd			Frank Leach Warmans Gate Tendring Road Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AA (as reputed owner) J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW (as reputed owner) Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN (as reputed owner) Unknown (as reputed owner)		Frank Leach Warmans Gate Tendring Road Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AA J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN Unknown					

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-022	Acquisition of Rights	2452 square metres of paddock and hedgerow (east of Tendring Road, B1035)	Frank Leach Warmans Gate Tendring Road Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AA	NONE	Frank Leach Warmans Gate Tendring Road Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AA	Julie Wendy Thompson 92 Station Road CLACTON-ON-SEA Essex CO15 1SG (in respect of rights granted by a Transfer dated 28 May 2015)
08-023	Acquisition of Rights	1827 square metres of agricultural land and public footpaths (FP 1 180) (east of Tendring Road, B1035)	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE	Sneating Hall Lane	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) HSBC Bank PLC 8 Canada Square LONDON E14 5EQ (as mortgagee for J B Fairley & Son Limited)

	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans	extent of acquisition or use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-024	Temporary Possession	12800 square metres of agricultural land, private access track and public footpaths (FP 3 180 and FP 4 180) (west of Golden Lane)		NONE	CM1 1QH (in respect of public footpaths (FP 3 180 and FP 4 180)) Strutt & Parker (Farms) Limited Unit 1 Old Park Farm	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Liana Enterprises Ltd Blake Morgan New Kings Court Tollgate Chandler's Ford EASTLEIGH SO53 3LG (as mortgagee to Strutt & Parker (Farms) Limited as contained in a Charge dated 13 September 2019)
08-025	Temporary Possession	14504 square metres of agricultural land (east of Tendring Road, B1035)	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	HSBC Bank PLC 8 Canada Square LONDON E14 5EQ (as mortgagee for J B Fairley & Son Limited)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation						
Land Plans		Description of failu	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
08-026	Acquisition of Rights	and public footpaths (FP 18 180 and FP 1 180) (east of Tendring Road, B1035)	0	NONE	Essex CM1 1QH (in respect of public footpaths (FP 1 180 and FP 18 180)) J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) HSBC Bank PLC 8 Canada Square LONDON E14 5EQ				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-001	Acquisition of Rights	1136 square metres of agricultural land (east of Tendring Road, B1035)	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) HSBC Bank PLC 8 Canada Square LONDON E14 5EQ (as mortgagee for J B Fairley & Son Limited) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			
09-002	Temporary Possession	1910 square metres of public highway and verges (Tendring Road, B1035)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-002 cont'd			J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW (in respect of subsoil beneath half width of public highway) Richard John Lord The Barn Bradley Hall Farm Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
	Temporary Possession	Tendring Road, B1035)	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 18 180)) J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	HSBC Bank PLC 8 Canada Square LONDON E14 5EQ (as mortgagee for J B Fairley & Son Limited) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)			
09-004	Temporary Possession	62 square metres of public highway and verges (Tendring Road, B1035)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
09-004 cont'd			J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW (in respect of subsoil beneath half width of public highway) Richard John Lord The Barn Bradley Hall Farm Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown				

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans		bescription of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-005	Temporary Possession	694 square metres of agricultural land (east of Tendring Road, B1035)	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	HSBC Bank PLC 8 Canada Square LONDON E14 5EQ (as mortgagee for J B Fairley & Son Limited) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)
09-006	Acquisition of Rights	771 square metres of agricultural land (east of Tendring Road, B1035)	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	NONE	J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW	HSBC Bank PLC 8 Canada Square LONDON E14 5EQ (as mortgagee for J B Fairley & Son Limited) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 Owners Lessees or Tenants Occupiers			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and			
09-007	Acquisition of Rights	852 square metres of public		or Reputed Lessees or Tenants NONE	or Reputed Occupiers Essex County Council	Procedures) Regulations 2009 Affinity Water Limited			
		highway and verges (Tendring Road, B1035)	County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) J B Fairley & Son Limited Kings Farm Sneating Hall Lane Kirby-Le-Soken FRINTON-ON-SEA Essex CO13 0EW (in respect of subsoil beneath half width of public highway) Richard John Lord The Barn Bradley Hall Farm Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH (in respect of subsoil beneath half width of public highway)		County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation					
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-007 cont'd			Unknown (as reputed owner)						
09-008	Acquisition of Rights	13526 square metres of agricultural land (south of Thorpe Road and west of Tendring Road, B1035)	Richard John Lord The Barn Bradley Hall Farm Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH	NONE	Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH	Laura Lilian Lord The Barn Bradley Hall Farm Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH (in respect of rights reserved as contained in a Transfer dated 25 October 2017) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-009	Acquisition of Rights	1798 square metres of public highway and verges (Thorpe Road)	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN (in respect of subsoil beneath half width of public highway) David Milburn Cannell The Old Barn Thorpe Road Tendring CLACTON-ON-SEA Essex CO16 9AR (in respect of subsoil beneath half width of public highway) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-009 cont'd			Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH (in respect of subsoil beneath half width of public highway) Lesley Elizabeth Mclean Smith 1 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA CO16 0AJ (in respect of subsoil beneath half width of public highway) Richard John Lord The Barn Bradley Hall Farm Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH (in respect of subsoil beneath half width of public highway)						

	Land w	hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-009 cont'd			Unknown (as reputed owner)			
09-010	Acquisition of Rights	8269 square metres of agricultural land, woodland (Pond Farm) and public footpath (FP 18 159) (south of Barker's Lane)	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH (as reputed owner)	NONE	Swan Road Beaumont CLACTON-ON-SEA	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)

	Land w	/hich is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-011	Acquisition of Rights	449 square metres of agricultural land (south of Swan Road)		NONE	Essex CO16 0AN Juliet Wallis Pond Farm Swan Road Beaumont	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)
09-012	Acquisition of Rights	737 square metres of public highway and verges (Swan Road)	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 OAN (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-012 cont'd			Cherie Ann Bomando 2 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA Essex CO16 OAJ (in respect of subsoil beneath half width of public highway) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 OAH (in respect of subsoil beneath half width of public highway)		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-012 cont'd			Ricky Gerard Bomando 2 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA Essex CO16 OAJ (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)						
09-013	Acquisition of Rights	6 square metres of hedgerow (west of Swan Road)	Cherie Ann Bomando 2 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA Essex CO16 0AJ (as reputed owner)	NONE	Cherie Ann Bomando 2 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA Essex CO16 0AJ	Unknown			

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-013 cont'd			Ricky Gerard Bomando 2 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA Essex CO16 0AJ (as reputed owner) Unknown (as reputed owner)		Ricky Gerard Bomando 2 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA Essex CO16 0AJ Unknown	
09-014	Acquisition of Rights	(2 Barkers Hall Cottages),	Cherie Ann Bomando 2 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA Essex CO16 0AJ	NONE	Cherie Ann Bomando 2 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA Essex CO16 0AJ	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 OAN (in respect of rights as granted by a Transfer dated 15 June 2006)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-014 cont'd			Ricky Gerard Bomando 2 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA Essex CO16 0AJ		Ricky Gerard Bomando 2 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA Essex CO16 0AJ				
09-015	Temporary Possession	339 square metres of agricultural land (north of Thorpe Road, B1035)	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN (as reputed owner) Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH (as reputed owner) Unknown (as reputed owner)	NONE	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH Unknown	Unknown			

	Land w	which is proposed to be subject	BOOK OF F to: (i) powers of compulsory acquisition, (ii	d Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to of Essex and Suffolk	carry out protective works (Regulation 7(1	.)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-016	Temporary Possession	4388 square metres of agricultural land (north of Thorpe Road, B1035)	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH (as reputed owner)	NONE	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
09-017	Acquisition of Rights	9105 square metres of agricultural land (north of Thorpe Road)	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN (as reputed owner)	NONE	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-017 cont'd			Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH (as reputed owner)		Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			
09-018	Temporary Possession	22937 square metres of agricultural land (north of Thorpe Road)	Unknown (as reputed owner) Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN (as reputed owner)	NONE	Unknown Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN	Unknown Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)			
			Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH (as reputed owner)		Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-018 cont'd			Unknown (as reputed owner)		Unknown	Unknown			
09-019	Temporary Possession	1363 square metres of agricultural land (north of Thorpe Road, B1035)	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH (as reputed owner)	NONE	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH	NONE			
09-020	Acquisition of Rights	42129 square metres of agricultural land and hedgerow (north of Thorpe Road, B1035)		NONE	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)			

	Land w	which is proposed to be subject	BOOK OF F to: (i) powers of compulsory acquisition, (ii	d Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to of Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-020 cont'd			Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH (as reputed owner)		Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	
09-022	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	
09-023	Acquisition of Rights	581 square metres of agricultural land (north of Thorpe Road, B1035)	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN (as reputed)	NONE	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN (as reputed)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		besen priori or land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-023 cont'd			Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH (as reputed owner) Unknown (as reputed owner)		Juliet Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AH Rob Long Jasmine Cottage Tendring Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0AA Unknown	Rob Long Jasmine Cottage Tendring Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0AA (in respect of rights of access) Unknown			
09-024	Acquisition of Rights	42538 square metres of agricultural land and hedgerow (north of Thorpe Road, B1035)		NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	David Milburn Cannell The Old Barn Thorpe Road Tendring CLACTON-ON-SEA Essex CO16 9AR (in respect of sporting rights granted by a Transfer dated 7 May 2004)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-024 cont'd						HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for James Fairley & Sons (Farms) Limited as contained in a Charge dated 7 May 2004)			
10-001	Acquisition of Rights	66919 square metres of agricultural land, hedgerow (north of Thorpe Road, B1035) and public footpaths (FP 22 179)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG Unknown (in respect of mines and minerals)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 22 179)) James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 6 November 2012)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-002	Acquisition of Rights	515 square metres of agricultural land (north of Thorpe Road, B1035)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) David Milburn Cannell The Old Barn Thorpe Road Tendring CLACTON-ON-SEA Essex CO16 9AR (in respect of sporting rights granted by a Transfer dated 7 May 2004) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for James Fairley & Sons (Farms) Limited as contained in a Charge dated 7 May 2004)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-003	Acquisition of Rights	3447 square metres of agricultural land, private access track and hedgerow (north of Thorpe Road, B1035) and public footpaths (FP 22 179 and FP 8 179)	Wolves Hall Lane Tendring	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpaths (FP 22 179 and FP 8 179)) James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 06 November 2012)			
10-004	Temporary Possession	5435 square metres of agricultural land (north of Thorpe Road, B1035)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-004 cont'd			Unknown (in respect of mines and minerals)			HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 6 November 2012)			
10-005	Temporary Possession	224 square metres of agricultural land (north of Thorpe Road, B1035) and public footpath (FP 8 179)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG Unknown (in respect of mines and minerals)	NONE	Essex CM1 1QH	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 6 November 2012)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-006		agricultural land, woodland and private access track (north of	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited) Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011) Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
10-007	Temporary Possession	4639 square metres of agricultural land (north of Thorpe Road, B1035)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited) Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)			

	Land w	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 2 Qualifying persons under Regulation		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-007 cont'd						Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)
10-008	Acquisition of Rights		James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 8 179)) James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-008 cont'd						Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011) Joanna Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
10-008 cont'd						Stephen Nigel Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access) Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)			
10-009	Acquisition of Rights		James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE		HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-009 cont'd						Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011) Joanna Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access) Stephen Nigel Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-009 cont'd						Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)			
10-010	Acquisition of Rights	3765 square metres of woodland and watercourse (Tendring Brook) (east of Lodge Lane)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited)			
10-011	Acquisition of Rights	road (known as Lodge Lane)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-011 cont'd						Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011) Joanna Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-011 cont'd						Stephen Nigel Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access) Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)			
10-012	Acquisition of Rights	• · · · ·	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE		HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-012 cont'd						Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011) Joanna Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Land w	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 ) right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	ise Description of land	Qualifying persons under Regulation 7(:	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-012 cont'd						Stephen Nigel Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 OBS (in respect of rights of access) Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 OBS (in respect of rights granted by a Transfer dated 9 June 2011)
10-013	Acquisition of Rights	agricultural land (east of Heath Road) and public footpaths (FP 3 179 and FP 8 179)		NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 3 179 and FP 8 179))	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-013 cont'd					James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 18 October 1962) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of the unknown covenants contained in Conveyance dated 29 November 1982) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 15 October 1954)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-013 cont'd						Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 6 October 1956)			
11-001	Acquisition of Rights	7341 square metres of agricultural land (north of Lodge Lane)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 18 October 1962)			
11-002	Acquisition of Rights	742 square metres of agricultural land and private access track (north of Lodge Lane)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 18 October 1962)			
11-003	Acquisition of Rights	578 square metres of private access track (south of Wolves Hall Lane)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
11-003 cont'd						HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 18 October 1962) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of the unknown covenants contained in Conveyance dated 29 November 1982) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 15 October 1954) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 16 October 1956)			

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-004	Acquisition of Rights	10 square metres of private access track (south of Wolves Hall Lane)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG (as reputed owner) Unknown (as reputed owner)	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG Unknown	Unknown
11-005	Acquisition of Rights	332 square metres of agricultural land (south of Wolves Hall Lane)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for James Fairley & Sons (Farms) Limited as contained in a Charge dated 10 April 2003)
11-006	Acquisition of Rights	24808 square metres of agricultural land (south of Wolves Hall Lane)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for James Fairley & Sons (Farms) Limited as contained in a Charge dated 10 April 2003)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009					
Land Plans	Extent of dequisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
	Acquisition of Rights	highway and verge (Wolves Hall Lane)	Market Road Chelmsford Essex CM1 1QH (as highway authority) James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG (in respect of subsoil beneath full width of public highway) Unknown (as reputed owner)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Unknown			
11-008	Acquisition of Rights	agricultural land and hedgerow	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	CLACTON-ON-SEA Essex	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 18 October 1962)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-009	Acquisition of Rights	agricultural land and hedgerow	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for James Fairley & Sons (Farms) Limited as contained in a Charge dated 10 April 2003)			
11-010	Acquisition of Rights		James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for James Fairley & Sons (Farms) Limited as contained in a Charge dated 10 April 2003)			
11-011	Acquisition of Rights	l v '	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 1 179))	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 18 October 1962)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
11-011 cont'd					James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG				
11-012	Acquisition of Rights	51258 square metres of agricultural land (south of Stones Green Road) and public footpath (FP 1 179)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	Essex CM1 1QH	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 18 October 1962) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 15 October 1954)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
11-013	Temporary Possession	2959 square metres of agricultural land (south of Stones Green Road)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 18 October 1962) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 15 October 1954)			
11-014	Temporary Possession	474 square metres of agricultural land (south of Stones Green Road)	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	NONE	James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to James Fairley & Sons (Farms) Limited as contained in a Charge dated 18 October 1962)			
11-015	Temporary Possession	39 square metres of public highway and verge (Stones Green Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans	Land Plans		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
11-015 cont'd			James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG (in respect of subsoil beneath half width of public highway) John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)		Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use				ations: Prescribed Forms and Procedures) Occupiers	Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and			
			or Reputed Owners	or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009			
11-016	Temporary Possession	244 square metres of agricultural land (north of Stones Green Road)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 12 May 1981) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
11-016 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 11 October 2010 and a Charge dated 23 September 2014) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)			
11-017	Temporary Possession	agricultural land and hedgerow (north of Stones Green Road)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 12 May 1981)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
11-017 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 11 October 2010 and a Charge dated 23 September 2014) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)			
11-018	Acquisition of Rights	3001 square metres of public highway and verges (Stones Green Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)			

	Land wi	hich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-018 cont'd			James Fairley & Sons (Farms) Limited Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG (in respect of subsoil beneath half width of public highway) John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of subsoil beneath half width of public highway) Richard Harrison-Osborne The Granary Stones Green Road Tendring CLACTON-ON-SEA CO16 0DD (in respect of subsoil beneath half width of public highway)		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown

	Land v	vhich is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans	Extent of acquisition of use	Description of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-018 cont'd			Sara Carol Harrison-Osborne The Granary Stones Green Road Tendring CLACTON-ON-SEA CO16 0DD (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)			
11-019	Acquisition of Rights	28 square metres of agricultural land (north of Stones Green Road)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 12 May 1981)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
11-019 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 11 October 2010 and a Charge dated 23 September 2014) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)			
11-020	Acquisition of Rights	agricultural land and public	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 31 183))	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	nt of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Procedures) Regulations 2009		
11-020 cont'd					MANNINGTREE Essex CO11 2NZ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 12 May 1981) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 11 October 2010 and a Charge dated 23 September 2014) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016)		

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-020 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)
12-001	Temporary Possession	highway and hedgerow (Stones Green Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Richard Harrison-Osborne The Granary Stones Green Road Tendring CLACTON-ON-SEA CO16 0DD	NONE	County Hall Market Road Chelmsford	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)

# $\lor \Xi \lor$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-001 cont'd			Sara Carol Harrison-Osborne The Granary Stones Green Road Tendring CLACTON-ON-SEA CO16 0DD		Sara Carol Harrison-Osborne The Granary Stones Green Road Tendring CLACTON-ON-SEA CO16 0DD				
12-002	Temporary Possession	680 square metres of agricultural land (north of Stones Green Road)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 12 May 1981) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 11 October 2010 and a Charge dated 23 September 2014)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-002 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)			
12-003	Temporary Possession	1825 square metres of agricultural land (north of Stones Green Road)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 12 May 1981) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 11 October 2010 and a Charge dated 23 September 2014)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-003 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)			
12-004	Acquisition of Rights	agricultural land and public footpaths (FP 32 183 and FP 31	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpaths (FP 32 183 and FP 31 183)) John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 12 May 1981)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
12-004 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 11 October 2010 and a Charge dated 23 September 2014) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-004 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)			
12-005	Acquisition of Rights	agricultural land, private track and public footpaths (FP 32 183, FP 14 183 and FP 31 183)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpaths (FP 32 183, FP 14 183 and FP 31 183)) John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 12 May 1981) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-005 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 11 October 2010 and a Charge dated 23 September 2014) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)			
12-006	Temporary Possession	agricultural land, private access track to Hempstalls Farm and public footpath (FP 15 183)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 15 183))	Aurlius Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
12-006 cont'd					MANNINGTREE Essex CO11 2NZ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 12 May 1981) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 11 October 2010 and a Charge dated 23 September 2014)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-006 cont'd						Chris Driver Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Chris Mudd Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-006 cont'd						Harry Weavers Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Jay McDermot Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Lewis Clarke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-006 cont'd						Lisa Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS (in respect of rights of access to Hempstalls Farm) Mark Taylor Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Nick Bowingcoke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
12-006 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)			
12-007	Temporary Possession	92705 square metres of agricultural land and hedgerow (south of A120)	James Burke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) Jiggens Trust Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (as reputed owner)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of an Agricultural Holdings Act 1986 Tenancy)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of an Agricultural Holdings Act 1986 Tenancy)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-007 cont'd			Joanna Burke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) Joanna Green Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (as trustee of the Jiggens Trust)			Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-007 cont'd			John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS (as trustee of the Jiggens Trust) Penny Toleman Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) Robert Church Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
12-008	Temporary Possession	279 square metres of public highway lay-by and verge (B1035)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Foxes Property 3 S.A.R.L. 1 Allee Scheffer L-2520 Luxembourg (in respect of subsoil beneath half width of public highway) James Burke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1) Owners	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
			or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Procedures) Regulations 2009			
12-008 cont'd			Jiggens Trust Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of subsoil beneath half width of public highway) Joanna Burke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway) Joanna Green Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway) Joanna Green Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway)			Tungsten Colchester Limited Gateway House 4 Penman Way Grove Park Enderby Leicester LE19 1SY (in respect of a commercial interest for development) Unknown			

	Land w	hich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii) r	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-008 cont'd			John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway) John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway)			

	Land w	hich is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o Essex and Suffolk	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-008 cont'd			Penny Toleman Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway) Robert Church Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)			

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-009	Temporary Possession	2728 square metres of public highway and verge (B1035)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Georgina Margaret Brown Newhouse Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of subsoil beneath half width of public highway) James Burke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Cont 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-009 cont'd			Jiggens Trust Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of subsoil beneath half width of public highway) Joanna Burke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway) Joanna Green Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway) Joanna Green Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway)						

	Land w	hich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii) r	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1)	i(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-009 cont'd			John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway) John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway)			

	Land w	hich is proposed to be subject t	BOOK OF RE co: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o Essex and Suffolk	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-009 cont'd			Penny Toleman Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway) Robert Church Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)			

	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-009 cont'd			Wix Farms Poultry Ltd 3 Manor Road COLCHESTER Essex CO3 3LU (in respect of subsoil beneath half width of public highway)			
12-010	Temporary Possession	371 square metres of agricultural land, private access track and public footpath (FP 37 138) (east of B1035)	James Burke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as reputed owner and as trustees of the Jiggens Trust) Jiggens Trust Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (as reputed owner)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 37 183)) Jiggens Trust Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of dequisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-010 cont'd			Joanna Burke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as reputed owner and as trustees of the Jiggens Trust) Joanna Green Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as reputed owner and as trustees of the Jiggens Trust) John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (as reputed owner and as trustee of the Jiggens Trust)		Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-010 cont'd			John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS (as reputed owner and as trustee of the Jiggens Trust) Penny Toleman Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as reputed owner and as trustees of the Jiggens Trust) Robert Church Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as reputed owner and as trustees of the Jiggens Trust) Robert Church Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as reputed owner and as trustees of the Jiggens Trust)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-010 cont'd			Unknown (as reputed owner)						
12-011	Temporary Possession	agricultural land and hedgerow (east of B1035)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 12 May 1981) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Charles Jiggens as contained in a Charge dated 11 October 2010 and a Charge dated 23 September 2014) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-011 cont'd						East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)			
12-012	Acquisition of Rights	highway and verges (A120)	John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS (in respect of subsoil beneath full width of public highway) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (as highway authority)	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (as highway authority) Unknown	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-012 cont'd			Unknown (as reputed owner)						
12-013			John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		beschption of hind	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-013 cont'd						Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of CK8 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans       Extent of acquisition or use       Description of land       Qualifying persons under Regular         Owners or Reputed Owners	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
		Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers						
12-013 cont'd						Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-013 cont'd						Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-013 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-013 cont'd						Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 15 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation					
Land Hans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-013 cont'd						Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of Unit 10 Bradfield Lodge) Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of Units 8 and 16 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-013 cont'd						Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of failed	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-013 cont'd						Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	isition or use Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
12-014	Acquisition of Rights	3224 square metres of agricultural land and private access track (north of A120)	John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS (trading as Wix Farms Ltd)	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Horsley Cross MANNINGTREE Essex CO11 2NS (trading as Wix Farms Ltd) John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-014 cont'd						Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of access to CK8 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-014 cont'd						Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-014 cont'd						Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-014 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-014 cont'd						Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 15 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-014 cont'd						Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of Unit 10 Bradfield Lodge) Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of Units 8 and 16 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-014 cont'd						Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-014 cont'd						Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-001	Acquisition of Rights	• · · ·	John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-001 cont'd						Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of access to CK7 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited as contained in a Charge dated 1 September 2016)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-001 cont'd						Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-001 cont'd						Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-001 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
13-001 cont'd						Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 15 Bradfield Lodge)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-001 cont'd						Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of Unit 10 Bradfield Lodge) Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of Units 8 and 16 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-001 cont'd						Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-001 cont'd						Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access in Unit 14 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
13-002	Acquisition of Rights	access track and hardstanding (Bradfield Lodge)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	NONE	Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-002 cont'd						Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of access to CK8 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-002 cont'd						Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 17 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-002 cont'd						Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER COT 8SE (in respect of rights of access) Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-002 cont'd						Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2) Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO12 2NS (in respect of access to Unit 20 Bradfield Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants Occupiers (Applications: Prescribed For	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
13-002 cont'd						Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 15 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge) Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 23 Bradfield Lodge) (in respect of access to Unit 23 Bradfield Lodge)			

# $\lor \Xi \lor$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-002 cont'd						Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 8 Bradfield Lodge) Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) (in respect of access to Unit 21 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	nt of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-002 cont'd						S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge) Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
13-002 cont'd						Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)				
13-003	Acquisition of Rights	299 square metres of private access track and hardstanding (Bradfield Lodge)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-003 cont'd					Horsley Cross MANNINGTREE Essex CO11 2NZ	Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of access to CK8 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-003 cont'd						Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-003 cont'd						Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 14 Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-003 cont'd						Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge) Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	it of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-003 cont'd						Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 15 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
13-003 cont'd						Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 8 and 16 Bradfield Lodge) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-003 cont'd						Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge)			

# $\lor \Xi \lor$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of failed	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-003 cont'd						Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-004	Acquisition of Rights	access track to Bradfield Lodge (south of Clacton Road)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	NONE	Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) (in respect of access to Unit 19 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-004 cont'd						Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of access to CK8 and Scaffold Yard) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-004 cont'd						Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-004 cont'd						Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER COT 8SE (in respect of rights of access) Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-004 cont'd						Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge) Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-004 cont'd						Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 15 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge) Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 23 Bradfield Lodge) (in respect of access to Unit 23 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-004 cont'd						Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 8 and 16 Bradfield Lodge) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-004 cont'd						Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge) Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-004 cont'd						Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)			
13-005	Acquisition of Rights	access track to Bradfield Lodge (off Clacton Road)	John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-005 cont'd					John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of CK8 and Scaffold Yard Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-005 cont'd						Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Jiggens Limited) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-005 cont'd						Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 17 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-005 cont'd						Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER COT 8SE (in respect of rights of access) Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-005 cont'd						Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge) Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-005 cont'd						John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of rights contained in a transfer dated the 22nd February 2011) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 15 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge) (in respect of access to Unit 10 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
13-005 cont'd						Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 8 and 16 Bradfield Lodge) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-005 cont'd						Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-005 cont'd						Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-006	Acquisition of Rights		James Burke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) Jiggens Trust Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ Joanna Burke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Co11 2NZ (as trustee of the Jiggens Trust)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of an Agricultural Holdings Act 1986 Tenancy)		A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	f acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-006 cont'd			Joanna Green Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (as trustee of the Jiggens Trust) John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (as trustee of the Jiggens Trust)			Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of access to CK8 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	nt of acquisition or use Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of bequisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-006 cont'd			Penny Toleman Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) Robert Church Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust)			Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) (in respect of access to Unit 2 Bradfield			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
13-006 cont'd						Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-006 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use	Description of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-006 cont'd						Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 15 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans	Owners     Lessees or Tenants     Occupiers       or Reputed Owners     or Reputed Lessees or Tenants     or Reputed Occupiers		7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
13-006 cont'd						Kelsy Jane Bamford 2 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 2 Jubilee Cottage) Leah Faye Banks 1 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 1 Jubilee Cottage) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
13-006 cont'd						Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 8 and 16 Bradfield Lodge) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-006 cont'd						Peter Anthony Banks 1 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 1 Jubilee cottage) Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	tent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-006 cont'd						Robin Neal Banks 1 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 1 Jubilee Cottage) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge) Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 2 Qualifying persons under Regulation					
Land Plans	extent of acquisition of use	Description of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-006 cont'd						Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Thomas William Wright 2 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 2 Jubilee Cottage) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-007	Acquisition of Rights	25135 square metres of agricultural land (east of Clacton road)	James Burke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) Jiggens Trust Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (as reputed owner) Joanna Burke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Cons Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of an Agricultural Holdings Act 1986 Tenancy)	John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of an Agricultural Holdings Act 1986 Tenancy)	NONE			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-007 cont'd			Joanna Green Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (as trustee of the Jiggens Trust) John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (as trustee of the Jiggens Trust)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-007 cont'd			Penny Toleman Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust) Robert Church Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (as trustee of the Jiggens Trust)						
13-008	Acquisition of Rights	20577 square metres of agricultural land (east of Clacton Road B1035)	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	NONE	Horsley Cross	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
13-008 cont'd						Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Jiggens Limited as contained in a charge dated 10 July 2024) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)		

	Land v	which is proposed to be subject	BOOK OF to: (i) powers of compulsory acquisition,	ind Farm Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights to of Essex and Suffolk	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-009	Temporary Possession	10798 square metres of agricultural land (east of Clacton Road)	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	NONE	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Jiggens Limited as contained in a charge dated 10 July
13-010	Temporary Possession	1594 square metres of agricultural land (east of Clacton Road)	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	NONE	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners or Reputed Owners	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009 Lessees or Tenants or Reputed Lessees or Tenants	ations: Prescribed Forms and Procedures) Occupiers or Reputed Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-010 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Jiggens Limited as contained in a charge dated 10 July 2024)			
13-011	Acquisition of Rights	350 square metres of agricultural land (east of Clacton Road)	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	NONE	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Jiggens Limited as contained in a charge dated 10 July 2024)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
13-012	Temporary Possession	195 square metres of agricultural land and hedgerow (east of Clacton Road)	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	NONE	John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to John Jiggens Limited as contained in a charge dated 10 July 2024)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-013	Temporary Possession	183 square metres of public highway and verge (Clacton Road) (excluding all the interests of the National Highways Limited)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	Essex CM1 1QH (as highway authority)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			
13-014	Temporary Possession	259 square metres of public highway and verge (east of Clacton Road) (excluding all the interests of the National Highways Limited)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE		Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-014 cont'd			John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of subsoil beneath half width of public highway) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of subsoil beneath half width of public highway) Robert Fairley Limited Abbotts Hall Clacton Road Horsley Cross MANNINGTREE CO11 2NX (in respect of subsoil beneath half width of public highway)		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Land w	hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-014 cont'd			Unknown (as reputed owner)			
13-015	Temporary Possession	60 square metres of agricultural land, verge and access splay (west of Clacton Road) (excluding all the interests of the National Highways Limited)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)
13-016	Temporary Possession	3 square metres of agricultural land, verge and access splay (west of Clacton Road) (excluding all the interests of the National Highways Limited)	County Hall Market Road Chelmsford	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-016 cont'd			John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of subsoil beneath half width of public highway) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of subsoil beneath half width of public highway) Robert Fairley Limited Abbotts Hall Clacton Road Horsley Cross MANNINGTREE CO11 2NX (in respect of subsoil beneath half width of public highway)		Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-016 cont'd			Unknown (as reputed owner)						
13-017	Temporary Possession	25799 square metres of agricultural land (west of Clacton Road)	Robert Fairley Limited Abbotts Hall Clacton Road Horsley Cross MANNINGTREE CO11 2NX	NONE	Nigel Graham Dyson Bentley Manor Church Road Little Bentley COLCHESTER Essex (Trading as Horsley Cross Car Boot Sales) Robert Fairley Limited Abbotts Hall Clacton Road Horsley Cross MANNINGTREE CO11 2NX	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights and restrictive covenants as contained in Deed dated 29 April 1936, Deed dated 6 June 1961, Deed dated 8 July 1970 and Deed dated 31 December 1990)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-017 cont'd						HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to Robert Fairley Limited as contained in a Charge dated 15 January 1999) Nigel Graham Dyson Bentley Manor Church Road Little Bentley COLCHESTER Essex CO7 8SE (in respect of right of access to sell goods) Robert Christmas Mulleys Cottage Bentley Road Little Bromley MANNINGTREE CO11 2PL (in respect of shooting rights)			

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-018	Acquisition of Rights	537 square metres of public highway and verges (Clacton Road) (excluding all the interests of the National Highways Limited)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE		Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)
13-019	Acquisition of Rights	3199 square metres of public highway and verges (Clacton Road) (excluding all the interests of the National Highways Limited)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE		Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-019 cont'd			John Jiggens Limited Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of subsoil beneath half width of public highway) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of subsoil beneath half width of public highway) Robert Fairley Limited Abbotts Hall Clacton Road Horsley Cross MANNINGTREE CO11 2NX (in respect of subsoil beneath half width of public highway)		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-019 cont'd			Unknown (as reputed owner)						
13-020	Acquisition of Rights	agricultural land (east of Clacton Road) (excluding all the interests of the National Highways Limited)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	Chelmsford Essex CM1 1QH (as highway authority) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			
13-021	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
13-022	Acquisition of Rights	agricultural land, private access track and verge (west of	Robert Fairley Limited Abbotts Hall Clacton Road Horsley Cross MANNINGTREE CO11 2NX	NONE	Church Road Little Bentley COLCHESTER Essex CO7 8SE (in respect of right of access to sell goods) Robert Fairley Limited Abbotts Hall Clacton Road Horsley Cross MANNINGTREE CO11 2NX	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights and restrictive covenants as contained in Deed dated 29 April 1936, Deed dated 6 June 1961, Deed dated 8 July 1970 and Deed dated 31 December 1990) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to Robert Fairley Limited as contained in a Charge dated 15 January 1999)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-022 cont'd						Nigel Graham Dyson Bentley Manor Church Road Little Bentley COLCHESTER Essex CO7 8SE (in respect of right of access to sell goods) Robert Christmas Mulleys Cottage Bentley Road Little Bromley MANNINGTREE CO11 2PL			
13-023	Acquisition of Rights	32744 square metres of agricultural land and brook (Holland Brook) (west of Clacton Road)	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	NONE	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for T. Fairley and Sons Limited as contained in a Charge dated 15 January 1999)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of failed	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-023 cont'd						Robert Christmas Mulleys Cottage Bentley Road Little Bromley MANNINGTREE CO11 2PL (in respect of shooting rights)			
13-024	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE			
14-001	Acquisition of Rights	377 square metres of agricultural land (west of Clacton Road)	Robert Fairley Limited Abbotts Hall Clacton Road Horsley Cross MANNINGTREE CO11 2NX	NONE	Robert Fairley Limited Abbotts Hall Clacton Road Horsley Cross MANNINGTREE CO11 2NX	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights and restrictive covenants as contained in Deed dated 29 April 1936, Deed dated 6 June 1961, Deed dated 8 July 1970 and Deed dated 31 December 1990) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee to Robert Fairley Limited as contained in a Charge dated 15th January 1999)			

	Land w	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	nd Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to of Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-001 cont'd						Robert Christmas Mulleys Cottage Bentley Road Little Bromley MANNINGTREE CO11 2PL (in respect of shooting rights)
14-002	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE
14-003	Acquisition of Rights	2114 square metres of private access track and brook (Welhams Farm) (east of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)

	Land v	which is proposed to be subject to	BOOK OF F o: (i) powers of compulsory acquisition, (ii	d Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to of Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-003 cont'd						The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for James Andrew Clachan as contained in a Charge dated 31 January 2017)
14-004	Acquisition of Rights	72678 square metres of agricultural land and private access track (east of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for James Andrew Clachan as contained in a Charge dated 31 January 2017)

	Land v	which is proposed to be subject t	BOOK OF f o: (i) powers of compulsory acquisition, (i	nd Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to of Essex and Suffolk	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Qualifying persons under Regulation		
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-005	Freehold Acquisition	709 square metres of agricultural land (east of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for James Andrew Clachan as contained in a Charge dated 31 January 2017)
14-006	Temporary Possession	21539 square metres of agricultural land (east of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for James Andrew Clachan as contained in a Charge dated 31 January 2017)

	Land v	which is proposed to be subject t	BOOK OF I co: (i) powers of compulsory acquisition, (i	nd Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to of Essex and Suffolk	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Qualifying persons under Regulation		
Land Plans		Description of failu	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-007	Freehold Acquisition	204 square metres of agricultural land (east of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for James Andrew Clachan as contained in a Charge dated 31 January 2017)
14-008	Acquisition of Rights	563 square metres of agricultural land (east of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for James Andrew Clachan as contained in a Charge dated 31 January 2017)

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-009	Freehold Acquisition	464 square metres of agricultural land (east of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for James Andrew Clachan as contained in a Charge dated 31 January 2017)
14-010	Acquisition of Rights	24 square metres of private access track (known as Mulberry Lane) and access splay (east of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (as reputed owner) T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (as reputed owner)	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	Unknown

	Land w	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-010 cont'd			Unknown (as reputed owner)		Unknown	
14-011	Freehold Acquisition	220 square metres of agricultural land (east of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	Essex CO7 8SS	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for James Andrew Clachan as contained in a Charge dated 31 January 2017)
14-012	Freehold Acquisition	254 square metres of verge (east of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (as reputed owner) Unknown (as reputed owner)	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS Unknown	Unknown

	Land w	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-013	Freehold Acquisition	701 square metres of agricultural land, public highway and verge (Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (in respect of subsoil beneath half width of public highway) James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans	Execution acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
14-013 cont'd			Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)						
14-014	Temporary Possession	591 square metres of public highway (Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Land Plans		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
14-014 cont'd			James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highway) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)		Unknown				

	Land v	which is proposed to be subject to	BOOK OF o: (i) powers of compulsory acquisition,	ind Farm Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights to of Essex and Suffolk	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-015	Temporary Possession	313 square metres of agricultural land, public highway and verges (Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)

	Land v	vhich is proposed to be subject t	BOOK OF c: (i) powers of compulsory acquisition, (	nd Farm Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) rights to of Essex and Suffolk	o carry out protective works (Regulation 7(1	)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-016	Freehold Acquisition	345 square metres of agricultural land, public highway and verges (Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	

	Land w	which is proposed to be subject	BOOK OF I to: (i) powers of compulsory acquisition, (i	nd Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to of Essex and Suffolk	o carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-017	Freehold Acquisition	26 square metres of agricultural land and verge (west of Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (as reputed owner) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	Unknown

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	i(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-017 cont'd			Unknown (as reputed owner)		T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX Unknown	
14-018	Freehold Acquisition	125 square metres of agricultural land (west of Bentley Road)	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) T&R Fairley	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)

	Land v	vhich is proposed to be subject t	BOOK OF F to: (i) powers of compulsory acquisition, (i	nd Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to of Essex and Suffolk	o carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-019	Freehold Acquisition	10 square metres of agricultural land (west of Bentley Road)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (as reputed owner) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL Unknown (as reputed owner)	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
14-020	Freehold Acquisition	18 square metres of agricultural land and verge (west of Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE		Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)			
14-021	Freehold Acquisition	14 square metres of agricultural land (west of Bentley Road)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)			

	Land v	which is proposed to be subject	BOOK OF to: (i) powers of compulsory acquisition, (i	nd Farm Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to of Essex and Suffolk	o carry out protective works (Regulation 7(1	L)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-022	Temporary Possession	4709 square metres of agricultural land (west of Bentley Road)	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
14-023	Temporary Possession	1438 square metres of agricultural land (west of Bentley Road)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (as reputed owner) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (as reputed owner) Unknown (as reputed owner)	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX Unknown	Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans	extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
14-024	Temporary Possession	, ,	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE			
14-025		, ,	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (as reputed owner) Unknown (as reputed owner)	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS Unknown	Unknown			
14-026	Freehold Acquisition	592 square metres of public highway (Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Land Plans		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
14-026 cont'd			Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (in respect of subsoil beneath half width of public highway) James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)		Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
14-027	Freehold Acquisition	285 square metres of public highway (Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (in respect of subsoil beneath half width of public highway) James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-027 cont'd			Unknown (as reputed owner)			
14-028	Freehold Acquisition	66 square metres of agricultural land and verge (west of Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE		Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)
14-029	Acquisition of Rights	12 square metres of agricultural land and verge (west of Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
14-029 cont'd			Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US		Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)			
14-030	Freehold Acquisition	103 square metres of agricultural land (west of Bentley Road)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			
14-031	Acquisition of Rights	453 square metres of agricultural land (west of Bentley Road)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
14-032	Freehold Acquisition	0 1 0	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for James Andrew Clachan as contained in a Charge dated 31 January 2017)				
14-033	Freehold Acquisition	3495 square metres of public highway (Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
	Land Plans		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
14-033 cont'd			Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (in respect of subsoil beneath half width of public highway) James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7( Owners or Reputed Owners						
14-034	Freehold Acquisition	57 square metres of public highway and verge (Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	Roger Wheatley 1 Hawkins Farm Cottages Paynes Lane Little Bromley MANNINGTREE CO11 2PJ	Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
14-035	Temporary Possession	8996 square metres of agricultural land (west of Bentley Road)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE			
14-036	Freehold Acquisition	838 square metres of agricultural land (west of Bentley Road)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE	Great Bromley COLCHESTER Essex CO7 7US	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Land v	vhich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii	d Farm Development Consent Order REFERENCE - PART 1 ) right to use the land, and/or (iii) rights to of Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-037	Acquisition of Rights	15994 square metres of agricultural land (west of Bentley Road)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE
14-038	Acquisition of Rights	17211 square metres of agricultural land (east of Payne's Lane)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)
14-039	Acquisition of Rights	51 square metres of verge (east of Payne's Lane)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (as reputed owner, pending first registration) Unknown (as reputed owner)	NONE	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US Unknown	Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
14-040	Acquisition of Rights	7 square metres of verge (east of Payne's Lane)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (as reputed owner) Unknown (as reputed owner)	NONE	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US Unknown	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (in respect of right of way) Unknown				
14-041	Acquisition of Rights	2 square metres of verge (east of Payne's Lane)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (as reputed owner) Unknown (as reputed owner)	NONE	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US Unknown	Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
14-042	Acquisition of Rights	64 square metres of public highway and verges (Payne's Lane)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)			
14-043	Acquisition of Rights	715 square metres of public highway (Payne's Lane)	Elspeth Elinor Knott Paynes Cottage Paynes Lane Little Bromley MANNINGTREE CO11 2PJ (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
14-043 cont'd			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (in respect of subsoil beneath half width of public highway) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (in respect of subsoil beneath half width of public highway)		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Land v	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1)	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-043 cont'd			Roland Alan Knott Paynes Cottage Paynes Lane Little Bromley MANNINGTREE CO11 2PJ (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)			
14-044	Acquisition of Rights	146 square metres of agricultural land (west of Payne's Lane)	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (as reputed owner) Unknown (as reputed owner)	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
14-044 cont'd					T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX Unknown				
14-045	Acquisition of Rights	agricultural land and grass land (west of Payne's Lane)	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Wivenhoe Road Crockleford Heath	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
14-046	Acquisition of Rights	agricultural land and grass land (west of Payne's Lane)	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			
					T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			
						Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)			

	Land v	vhich is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-001	Freehold Acquisition	1085 square metres of agricultural land and hedgerow (east of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)
15-002	Freehold Acquisition	953 square metres of public highway and verges (Bentley Road)	Davina Andreena Templeton Oakwood Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highway) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-002 cont'd			Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (in respect of subsoil beneath half width of public highway) James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highway) Paul John Dimond Orchard Cottage Bentley Road Little Bentley COLCHESTER CO7 8SS (in respect of subsoil beneath half width of rublic highway)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-002 cont'd			Shane James Templeton Oakwood Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highway) The Executor of the Estate of the Late Kerstin Jane Dimond Orchard Cottage Bentley Road Little Bentley COLCHESTER CO7 8SS (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-003	Freehold Acquisition	10 square metres of verge (east of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (as reputed owner) Unknown (as reputed owner)	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS Unknown	Unknown			
15-004	Freehold Acquisition	20 square metres of public highway and access splay (Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Market Road Chelmsford Essex CM1 1QH (as highway authority)	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for James Andrew Clachan as contained in a Charge dated 31 January 2017)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-004 cont'd			James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS		James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS				
15-005	Freehold Acquisition	62 square metres of public highway and verges (Bentley Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners or Reputed Owners	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
15-006	Freehold Acquisition	3089 square metres of agricultural land (west of Bentley Road)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	NONE	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights contained in a Deed dated 22 February 2008) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			
15-007	Freehold Acquisition	8381 square metres of public highway and verges (Bentley Road) (excluding all the interests of the National Highways Limited)	David Brinley Lifton Pellens Cottage Pelhams Corner Bentley Road Little Bentley COLCHESTER CO7 8SS (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
15-007 cont'd			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (in respect of subsoil beneath half width of public highway) Holly Marie Florence Johnson The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 CUS		National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of subsoil beneath half width of public highway) Unknown	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-007 cont'd			James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highway) John Paul Jeffery Traveller The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of subsoil beneath half width of public highway)						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-007 cont'd			Pauline Margaret Lifton Pellens Cottage Pelhams Corner Bentley Road Little Bentley COLCHESTER CO7 8SS (in respect of subsoil beneath half width of public highway) Rachael Donna Thackery The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highway) Richard John Clachan Craigus Bentley Road Little Bentley COLCHESTER CO7 8SS (in respect of subsoil beneath half width of public highway) Richard John Clachan Craigus Bentley Road						

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-007 cont'd			Russell Albert Johnson The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)	The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS In respect of subsoil beneath half width of public highway) Jnknown					
15-008	(as reputed owner)								

Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans Extent of acquisition or use		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
15-009 Freehold Acquisition	73 square metres of garden (The Nook)	Holly Marie Florence Johnson The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS John Paul Jeffery Traveller The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS Rachael Donna Thackery The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	Holly Marie Florence Johnson The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS John Paul Jeffery Traveller The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS Rachael Donna Thackery The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	Skipton Building Society The Bailey SKIPTON North Yorkshire BD23 1DN (as mortgagee for Holly Marie Florence Johnson, John Paul Jeffery Traveller, Rachael Donna Thackery and Russell Albert Johnson as contained in a Charge dated 12 December 2023)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
15-009 cont'd			Russell Albert Johnson The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS The Executor of the Estate of the Late Stephen James McFarlane The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS		Russell Albert Johnson The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS The Executor of the Estate of the Late Stephen James McFarlane The Nook Bentley Road Little Bentley COLCHESTER Essex CO7 8SS			
15-010		578 square metres of agricultural land (north of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	CO7 8SS	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for James Andrew Clachan as contained in a Charge dated 31 January 2017)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
15-011	Freehold Acquisition	646 square metres of agricultural land (north of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS Linda Maureen Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS Linda Maureen Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee to James Andrew Clachan, Linda Maureen Clachan and Richard John Clachan as dated 6 April 2023) Unknown				
			Craigus Bentley Road Little Bentley COLCHESTER CO7 8SS		Craigus Bentley Road Little Bentley COLCHESTER CO7 8SS	(in respect of rights contained in a Conveyance dated 26 June 1968)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of raild	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-012	Temporary Possession	Road) (excluding all the	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE			
15-013	Freehold Acquisition	348 square metres of agricultural land and hedgerow (north of Bentley Road)	James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS	NONE		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
15-013 cont'd			Linda Maureen Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			
			Richard John Clachan Craigus Bentley Road Little Bentley COLCHESTER CO7 8SS			The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee to James Andrew Clachan, Linda Maureen Clachan and Richard John Clachan as dated 6 April 2023)			
						Unknown (in respect of rights contained in a Conveyance dated 26 June 1968)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning						
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
15-014	Freehold Acquisition	74 square metres of public highway and verges (Bentley Road) (excluding all the interests of the National Highways Limited)	David Brinley Lifton Pellens Cottage Pelhams Corner Bentley Road Little Bentley COLCHESTER CO7 8SS (in respect of subsoil beneath half width of public highway) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) James Andrew Clachan Welhams Farm Bentley Road Little Bentley COLCHESTER Essex CO7 8SS (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of subsoil beneath half width of public highway) Unknown	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-014 cont'd			National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of subsoil beneath half width of public highway) Pauline Margaret Lifton Pellens Cottage Pelhams Corner Bentley Road Little Bentley COLCHESTER CO7 8SS (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)						
15-015	Temporary Possession	18 square metres of public highway and verge (Bentley Road) (excluding all the interests of the National Highways Limited)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
15-015 cont'd			National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ		National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ					
15-016	Temporary Possession	2142 square metres of public highway and verge (Harwich Road, A120) (excluding all the interests of the National Highways Limited)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)				
15-017	Temporary Possession	2857 square metres of public highway (A120) (excluding all the interests of the National Highways Limited)	Christopher Burgoyne 35 Foster Road ABINGDON OX14 1YN (in respect of subsoil beneath half width of public highway)	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (as highway authority)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-017 cont'd			James Tweed 35 Foster Road ABINGDON OX14 1YN (in respect of subsoil beneath half width of public highway) Jim Clifton Castle Byeways Pellens Corner Little Bentley Essex CO7 8SR (in respect of subsoil beneath half width of public highway) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (as highway authority) Unknown (as reputed owner)			Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Land w	/hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-018	Temporary Possession	200 square metres of public highway and verge (A120) (excluding all the interests of the National Highways Limited)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
15-019	Temporary Possession	1959 square metres of public highway and verges (A120) (excluding all the interests of the National Highways Limited)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE
15-020	Temporary Possession	334 square metres of public highway and verge (A120) (excluding all the interests of the National Highways Limited)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
15-021	Temporary Possession	219 square metres of public highway and verges (A120) (excluding all the interests of the National Highways Limited)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

	Land w	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (i	nd Farm Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to of Essex and Suffolk	o carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans		e Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-022	Temporary Possession	hedgerow and verge (south of A120) (excluding all the interests of the National	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)
16-001	Acquisition of Rights	336 square metres of agricultural land (east of Spratts Lane)	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-001 cont'd					T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX				
16-002	Acquisition of Rights	42 square metres of public highway and verge (Spratts Lane)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-002 cont'd					T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX				
16-003	Acquisition of Rights	10 square metres of public highway and verge (Spratts Lane)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-003 cont'd					T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX				
16-004	Acquisition of Rights	18 square metres of agricultural land (east of Spratts lane)	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
	Acquisition of Rights	1559 square metres of public highway and verges (Spratts Lane)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (in respect of subsoil beneath full width of public highway) Unknown (as reputed owner)	NONE	Essex CM1 1QH (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			
16-006	Acquisition of Rights	15 square metres of public highway and verges (Spratts Lane)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-006 cont'd			Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (in respect of subsoil beneath full width of public highway) Unknown (as reputed owner)		Unknown				
16-007	Acquisition of Rights	41 square metres of public highway and verge (Spratts Lane)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-007 cont'd			Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (in respect of subsoil beneath full width of public highway) Unknown (as reputed owner)		Unknown				
16-008	Acquisition of Rights	agricultural land and hedgerow (west of Spratts Lane)	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (as reputed owner) Unknown (as reputed owner)	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-008 cont'd					T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX Unknown				
16-009	Acquisition of Rights	144 square metres of agricultural land (west of Spratts Lane)	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (as reputed owner) Unknown (as reputed owner)	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-009 cont'd					T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX Unknown				
16-010	Acquisition of Rights	198 square metres of agricultural land (west of Spratts Lane)	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (as reputed owner) Unknown (as reputed owner)	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-010 cont'd					T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX Unknown				
16-011	Acquisition of Rights	146 square metres of agricultural land (west of Spratts Lane)	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-012	Acquisition of Rights	· ·	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 17 172)) T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
16-013	Acquisition of Rights	33129 square metres of agricultural land (east of Barlon Road) and public footpath (FP 17 172)	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 17 172)) T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers					
16-014	Acquisition of Rights	'	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (as reputed owner) Unknown (as reputed owner)	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement) T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX Unknown	Unknown				
16-015	Acquisition of Rights	41 square metres of public highway and agricultural land (Barlon Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX	Allens Farm Partners Allens Farm Wivenhoe Road Crockleford Heath COLCHESTER Essex CO7 7BN (in respect of a cropping agreement)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
16-015 cont'd			Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL		Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T&R Fairley Abbotts Hall Clacton Road Horsley Cross MANNGINGTREE Essex CO11 2NX					
16-016	Acquisition of Rights	710 square metres of public highway and verges (Barlon Road)	David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP (in respect of subsoil beneath half width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers				
16-016 cont'd			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)		Unknown				
16-017	Temporary Possession	20 square metres of agricultural land and verge (west of Barlon Road)	David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP	NONE	David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP	NONE			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-018	Acquisition of Rights		David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE			
16-019	Acquisition of Rights	50 square metres of public highway and agricultural (Barlon Road)	David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP	NONE	David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP	NONE			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-019 cont'd			Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)		Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)				
16-020	Acquisition of Rights	agricultural land, private access track and public footpath (FP	David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP	NONE	David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 16 172))	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-021	Acquisition of Rights	agricultural land, private access track and public footpath (FP 16 172) (west of Barlon Road)	David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP	NONE	David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 16 172))	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)			
16-022	Temporary Possession	agricultural land, private access tack and public footpath (FP 16 172) (west of Barlon Road)		NONE	David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-022 cont'd					Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 16 172))				
16-023	Acquisition of Rights	public footpath (FP 16 172)	David William Salmon Hall Farm Church Road Little Bromley MANNINGTREE CO11 2PP	NONE	County Hall Market Road Chelmsford	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners or Reputed Owners	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
17-001	Acquisition of Rights	agricultural land and hedgerow (south of Ardleigh Road)	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	NONE	Horsley Cross MANNINGTREE Essex CO11 2NU	Apatura Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Exclusivity Agreement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

# $\lor \Xi \lor$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-001 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012)			
17-002	Acquisition of Rights	agricultural land and hedgerow (south of Ardleigh Road)	,		T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	Apatura Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Exclusivity Agreement)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-002 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-002 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012)			
17-003	Temporary Possession	2442 square metres of agricultural land and drain (south of Ardleigh Road)	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	NONE	Clacton Road Horsley Cross MANNINGTREE Essex	Apatura Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Exclusivity Agreement)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-003 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-003 cont'd						The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012)			
17-004	Acquisition of Rights	agricultural land and hedgerow (south of Ardleigh Road)	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	NONE	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	Apatura Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Exclusivity Agreement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
17-004 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-004 cont'd						The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012)			
17-005	Temporary Possession	12584 square metres of agricultural land (south of Ardleigh Road)	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	NONE	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	Apatura Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Exclusivity Agreement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
17-005 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012)		
17-006	Acquisition of Rights	Farm (off Ardleigh Road), hedgerow and public footpath	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 15 172))	Apatura Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Exclusivity Agreement)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and		
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Procedures) Regulations 2009		
17-006 cont'd					T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-006 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
17-007	Acquisition of Rights	access track to Catts Green Farm (off Ardleigh Road)	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (as reputed owner) Unknown (as reputed owner)	NONE	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown				
17-008	Acquisition of Rights	highway, verges (Ardleigh Road) and public footpath (FP 15 172)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (in respect of subsoil beneath full width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public footpath (FP 15 172))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-008 cont'd			Unknown (as reputed owner)		Unknown	Unknown			
17-009	Acquisition of Rights	10 square metres of agricultural land and verge (south of Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	Apatura Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Exclusivity Agreement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
17-009 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-010	Acquisition of Rights	16 square metres of agricultural land (south of Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as reputed owner) T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (as reputed owner) Unknown (as reputed owner)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU Unknown	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown			
17-011	Acquisition of Rights	51 square metres of agricultural land (south of Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as reputed owner)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-011 cont'd			T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (as reputed owner) Unknown (as reputed owner)		T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU Unknown	Unknown			
17-012	Temporary Possession	411 square metres of agricultural land (south of Ardleigh Road)	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	NONE		Apatura Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Exclusivity Agreement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-012 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Lawford Solar Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Option Agreement) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-012 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012)			
17-013	Temporary Possession	61 square metres of agricultural land and verge (south of Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as reputed owner)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-013 cont'd			T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (as reputed owner) Unknown (as reputed owner)		T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU Unknown	Unknown			
17-014	Temporary Possession	489 square metres of public highway and verges (Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (in respect of subsoil beneath full width of public highway)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Qualifying persons under Regulation			
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-014 cont'd			Unknown (as reputed owner)						
17-015	Acquisition of Rights	30 square metres of agricultural land and verge (north of Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965)			
			T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU		T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	Lawford Solar Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Option Agreement)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-015 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
	Acquisition of Rights	175 square metres of agricultural land and verge (north of Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (in respect of subsoil beneath full width of public highway) Unknown (as reputed owner)	NONE	Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown				
17-017	Acquisition of Rights	107 square metres of agricultural land and verge (north of Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-017 cont'd			T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (in respect of subsoil beneath full width of public highway) Unknown (as reputed owner)		Unknown				
17-018	Acquisition of Rights	428 square metres of agricultural land and verge (north of Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Market Road Chelmsford Essex CM1 1QH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-018 cont'd			T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU		T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Lawford Solar Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Option Agreement) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Qualifying persons under Regulation			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-018 cont'd						The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)			
17-019	Acquisition of Rights	35 square metres of agricultural land and verge (north of Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	t of acquisition or use Description of land	Qualifying persons under Regulation 7(1 Owners	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
17-019 cont'd			or Reputed Owners T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	or Reputed Lessees or Tenants	or Reputed Occupiers T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	Lawford Solar Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Option Agreement) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
17-019 cont'd						Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)				
17-020	Acquisition of Rights	407 square metres of public highway and verges (Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (in respect of subsoil beneath full width of public highway) Unknown (as reputed owner)	NONE	CM1 1QH (as highway authority) Unknown	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	use Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			Category 2 Qualifying persons under Regulation			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
	Acquisition of Rights	342 square metres of public highway and verges (Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (in respect of subsoil beneath full width of public highway) Unknown (as reputed owner)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			
17-022	Acquisition of Rights	933 square metres of public highway and verges (Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-022 cont'd			T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (in respect of subsoil beneath full width of public highway) Unknown (as reputed owner)		Unknown	Unknown			
17-023	Freehold Acquisition	120 square metres of agricultural land (west of Ardleigh Road)	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (as reputed owner) Unknown (as reputed owner)	NONE	MANNINGTREE	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-024	Freehold Acquisition	296427 square metres of agricultural land and drain (north of Ardleigh Road)	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	NONE	T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Lawford Solar Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Option Agreement)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
17-024 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-024 cont'd						Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)			
17-025	Freehold Acquisition	182196 square metres of agricultural land and verge (east of Grange Road)	Michael Hughes Whirledge & Nott Ltd The Black Barn Lubards Lodge Hullbridge Road RAYLEIGH Essex SS6 9QG (as the executor of the estate for the Late Charles Tabor) Rebecca Mason Holmes & Hills LLP A12 Commercial Hub 86 London Road Marks Tey COLCHESTER COG 1ED (as the executor of the estate for the Late Charles Tabor)	NONE		Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Charles James Tabor as contained in a Charge dated 3 March 2016) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Land v	which is proposed to be subject t	BOOK OF R to: (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-025 cont'd			The Executor of The Estate of the Late Charles James Tabor Sutton Hall Shopland Road ROCHFORD SS4 1LH			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights contained in a Deed dated 10 December 1942)
17-026	Acquisition of Rights	2436 square metres of agricultural land and verge (north of Ardleigh Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation				
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
17-026 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Lawford Solar Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Option Agreement) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
17-026 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-027	Acquisition of Rights	64 square metres of agricultural land and verge (east of Grange Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	NONE	Essex CO11 2NU	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Lawford Solar Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (as beneficiary in respect of an Option Agreement) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
17-027 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-028	Acquisition of Rights	463 square metres of agricultural land (east of Grange Road)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965)			
			T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU		T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Lawford Solar Ltd 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT			
						(as beneficiary in respect of an Option Agreement)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-028 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for T. Fairley & Sons Limited as contained in a Charge dated 28 December 2012)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Qualifying persons under Regulation			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-028 cont'd						Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)			
17-029	Acquisition of Rights		Elizabeth Birgitta Harris Badley Hall Little Bromley Road Ardleigh COLCHESTER CO7 7NF (in respect of subsoil beneath half width of public highway) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Unknown			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Jest provi of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-029 cont'd			T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)						
17-030	Acquisition of Rights	2838 square metres of public highway and verges (Grange Road)	Elizabeth Birgitta Harris Badley Hall Little Bromley Road Ardleigh COLCHESTER CO7 7NF (in respect of subsoil beneath half width of public highway) Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority)	NONE	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (as highway authority) Unknown	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans		Description of failed	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-030 cont'd			T. Fairley & Sons Limited New Hall Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NU (in respect of subsoil beneath half width of public highway) Unknown (as reputed owner)			Unknown
17-031	Acquisition of Rights	338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)	Elizabeth Birgitta Harris Badley Hall Little Bromley Road Ardleigh COLCHESTER CO7 7NF	NONE	Elizabeth Birgitta Harris Badley Hall Little Bromley Road Ardleigh COLCHESTER CO7 7NF	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-031 cont'd						Moorhouse Farms Limited Begbies Traynor Town Wall House 4 Balkerne Hill COLCHESTER CO3 3AD (in respect of rights granted by a Deed dated 19 July 2018) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Philip Douglas Reeve Waterhouse Farm Waterhouse Farm Waterhouse Lane Ardleigh COLCHESTER Essex CO7 7NB (in respect of rights contained in a Conveyance dated 4 July 1985)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-031 cont'd						Unknown (in respect of restrictive covenants and rights reserved by a Transfer dated 3 April 1995)			
18-001	Acquisition of Rights	34524 square metres of agricultural land and hedgerow (east of Hungerdown Lane)	Elizabeth Birgitta Harris Badley Hall Little Bromley Road Ardleigh COLCHESTER CO7 7NF	NONE	Elizabeth Birgitta Harris Badley Hall Little Bromley Road Ardleigh COLCHESTER CO7 7NF	NONE			
			Peter Leslie Harris Badley Hall Little Bromley Road Ardleigh COLCHESTER CO7 7NF		Peter Leslie Harris Badley Hall Little Bromley Road Ardleigh COLCHESTER CO7 7NF				
18-002	Acquisition of Rights	agricultural land and hedgerow	Timothy Simon Ecott Holly Tree Nursery Hungerdown Lane Ardleigh COLCHESTER CO7 7LZ	Moorhouse Farms Limited Begbies Traynor Town Wall House 4 Balkerne Hill COLCHESTER CO3 3AD	Moorhouse Farms Limited Begbies Traynor Town Wall House 4 Balkerne Hill COLCHESTER CO3 3AD	NONE			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	Category 1 )(a) of the Infrastructure Planning (Applica Regulations 2009 Lessees or Tenants	ations: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and			
			or Reputed Owners	or Reputed Lessees or Tenants	or Reputed Occupiers	Procedures) Regulations 2009			
19-001	Acquisition of Rights	506 square metres of river (River Ore) and jetty (south of Orford Quay) (excluding all the interests of the Crown)	Orford and Gedgrave Parish Council The Town Hall Market Hill Orford Woodbridge IP12 2NZ (as trustee of the New Orford Town Trust)	NONE		Orford and Gedgrave Parish Council The Town Hall Market Hill Orford Woodbridge IP12 2NZ (in respect of restrictive covenants as contained in a Deed dated 5 February 1987) Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987) The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA (in respect of rights of access)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1 Owners	Category 1 nder Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procee Regulations 2009 ers Lessees or Tenants Occupiers		Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			or Reputed Owners	or Reputed Lessees or Tenants	or Reputed Occupiers				
19-002		Ore) (excluding all the interests of the Crown)	The Old Vicarage	NONE	KING'S LYNN PE32 2AA	East Anglia One North Limited 1 Tudor Street LONDON EC4Y 0AH (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) East Anglia Two Limited 1 Tudor Street LONDON EC4Y 0AH (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
19-002 cont'd						Norfolk Boreas Limited 70 St Mary Axe LONDON EC3A 8BE (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Norfolk Vanguard West Limited 70 St Mary Axe LONDON EC3A 8BE (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
19-002 cont'd						Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987, in a Conveyance dated 26 August 1913 and in a Conveyance dated 20 January 1914) The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA (in respect of rights of way and access to maintain services as contained in Conveyance dated 31 March 1993)			

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation				
Land Plans	Extent of dequisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
19-002 cont'd						The United Kingdom Atomic Energy Authority Culham Centre for Fusion Energy Culham Science Centre ABINGDON OX14 3DB (in respect of rights and restrictive covenants a contained a Deed dated 25 July 1962) Unknown (in respect of rights as stated in Conveyance dated 26 August 1913) Unknown (in respect of rights as stated in Conveyance dated 20 January 1914) Unknown (in respect of right of way as contained in Conveyance dated 30 June 1925)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
19-003	Acquisition of Rights	Ore) (excluding all the interests of the Crown)	Orford and Gedgrave Parish Council The Town Hall Market Hill Orford Woodbridge IP12 2NZ (as trustee of the New Orford Town Trust)	NONE	Orford Woodbridge IP12 2NZ (as trustee of the New Orford Town Trust)	Orford and Gedgrave Parish Council The Town Hall Market Hill Orford Woodbridge IP12 2NZ (in respect of restrictive covenants as contained in a Deed dated 5 February 1987) Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987)			

	Land w	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-004	Acquisition of Rights	14 square metres of private access track and scrubland (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of mines and minerals)	NONE	KING'S LYNN	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993)
			The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA		The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA	Unknown (in respect of rights as stated in Conveyance dated 31 March 1993)
						Unknown (in respect of rights as stated in Conveyance dated 26 August 1913)

	Land w	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	I Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
19-005	Acquisition of Rights	78 square metres of private access track (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of mines and minerals)	NONE	Castle Acre KING'S LYNN	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993)
			The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA		The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993) Unknown (in respect of right of way as stated in
			(in respect of mines and minerals) The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire		(in respect of mines and minerals) The National Trust For Places Of Histo Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire	oric

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk						
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-006	Acquisition of Rights	86 square metres of private access track (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of mines and minerals)	NONE	Castle Acre KING'S LYNN	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993)	
			The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA		Wiltshire SN2 2NA	G Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993) Unknown	
						(in respect of right of way as stated in Conveyance dated 30 June 1925)	

	Land w	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	Farm Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-007	Acquisition of Rights	97 square metres of private access track (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of mines and minerals)	NONE		Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993)
			The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA		Kemble Drive SWINDON Wiltshire	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk						
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
20-001	Acquisition of Rights	60 square metres of private access track (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of mines and minerals) The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA	NONE	Castle Acre KING'S LYNN PE32 2AA (in respect of mines and minerals) The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993)	
20-002	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	

	Land w	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	d Farm Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to f Essex and Suffolk	carry out protective works (Regulation 7(1	)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(	Category 2 Qualifying persons under Regulation		
Land Plans			Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
20-003	Acquisition of Rights	radio masts (east of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA	NONE	The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA	East Anglia One North Limited 1 Tudor Street LONDON EC4Y 0AH (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) East Anglia Two Limited 1 Tudor Street LONDON EC4Y 0AH (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Norfolk Boreas Limited 70 St Mary Axe LONDON EC3A 8BE (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk					
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans	Land Plans Extent of acquisition or use Description of land		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
20-003 cont'd						Norfolk Vanguard West Limited 70 St Mary Axe LONDON EC3A 8BE (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022)
						Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987, in a Conveyance dated 26 August 1913 and dated 20 January 1914)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Essex and Suffolk						
Number on	Extent of acquisition or use	uisition or use Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
20-003 cont'd						The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA (in respect of rights of way and access to maintain services as contained in Conveyance dated 31 March 1993) The United Kingdom Atomic Energy Authority Culham Centre for Fusion Energy Culham Science Centre ABINGDON OX14 3DB (in respect of rights and restrictive covenants a contained a Deed dated 25 July 1962) Unknown (in respect of rights as stated in Conveyance dated 20 January 1914)	
20-004	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	NOT IN USE	

	Five Estuaries	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
	1	Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-001	52 square metres of verges, copse and access splay (east of Frinton Road, B1032)	brian Leonard Cross2 Sluice CottagesManor WayCLACTON-ON-SEACO15 STZ(in respect of rights of access)Gunfleet Boating ClubManor WayCLACTON-ON-SEAEssexCO15 STZ(in respect of rights of access)Irene Cross2 Sluice CottagesManor WayCLACTON-ON-SEAEssex(in respect of rights of access)Irene Cross2 Sluice CottagesManor WayCLACTON-ON-SEAEssexCO15 STZ(in respect of rights of access)Manor WayCLACTON-ON-SEACO15 STZ(in respect of rights of access)Maria Ann Martin1 Sluice CottagesManor WayCLACTON-ON-SEAEssexCO15 STZ(in respect of rights of access)Mount VayCLACTON-ON-SEAEssexCO15 STZ(in respect of rights of access)Mountview Estates PLCMountview Estates PLCMountview House151 High StreetLONDONN14 6EW(in respect of rights of access)

	Five Estuaries	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-001 cont'd		Port Of London Authority London River House Royal Pier Road GRAVESEND Kent DA12 2BG (in respect of rights of access) Stephen Philip Martin 1 Sluice Cottages Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access) Unknown
01-002	6416 square metres of access way (off Frinton Road) leading to public road (Manor Way) verges and access tracks (north of The Esplanade, Holland-on-Sea) and public footpath (FP 29 167)	Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of rights granted by a Conveyance dated 21 October 1930, in Conveyance dated 3 August 1932, reserved in a Conveyance dated 9 August 1977 and in respect of right of way and easement) Brian Leonard Cross 2 Sluice Cottages Manor Way CLACTON-ON-SEA CO15 5TZ (in respect of rights of access)

	Five Estuarie	s Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
	1	Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-002 cont'd		Environment Agency Horizon House Dearnery Road BRISTOL Avon BSI SAH (In respect of the legal easements granted contained in a Deed of Easement dated 11 October 2019, in a Deed of Grant dated 26 February 1986, in a Deed of Grant dated 4 April 1989 and in a Deed of Grant dated 16 January 1998) Gunfleet Boating Club Manor Way CLACTON-ON-SEA Essex CO15 STZ (In respect of rights of access) Irene Cross 2 Sluce Cottages Manor Way CLACTON-ON-SEA CO15 STZ (In respect of rights of access) Maria Ann Martin 1 Sluce Cottages Manor Way CLACTON-ON-SEA Essex CO15 STZ (In respect of rights of access) Maria Ann Martin 1 Sluce Cottages Manor Way CLACTON-ON-SEA Essex CO15 STZ (In respect of rights of access) Mountview Estates PLC Mountview House 1 S1 High Street LONDON N14 GFW (In respect of rights of access)

	Five Estuaries	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-002 cont'd		Port Of London Authority London River House Royal Pier Road GRAVESEND Kent DA12 2BG (in respect of rights of access) Stephen Philip Martin 1 Sluice Cottages Manor Way CLACTON-ON-SEA Essex CO15 STZ (in respect of rights of access) Unknown (in respect of covenants contained in Conveyance dated 16 May 1902) Unknown (in respect of covenants contained in a Conveyance dated 7 October 1930) Unknown (in respect of covenants contained in a Conveyance dated 23 April 1938)
01-003	10490 square metres of access way off Frinton Road leading to public road (Manor Way) grassland, verges, access tracks and buildings (south of The Esplanade, Holland-on-Sea) and public footpath (FP 29 167)	Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of rights granted by a Conveyance dated 21 October 1930, in Conveyance dated 3 August 1932, reserved in a Conveyance dated 9 August 1977 and in respect of right of way and easement)

r	Five Estuaries	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-003 cont'd		Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of the legal easements granted contained in Deed of Easement dated 11 October 2019, in a Deed of Grant dated 26 February 1986, in a Deed of Grant dated 4 April 1989 and in a Deed of Grant dated 16 January 1998) Port Of London Authority London River House Royal Pier Road GRAVESEND Kent DA12 2BG (in respect of rights of access) Unknown (in respect of covenants contained in Conveyance dated 16 May 1902) Unknown (in respect of covenants contained in a Conveyance dated 27 October 1930)
01-004	63 square metres of hardstanding (south of Holland Brook) and public footpath (FP 29 167)	Unknown
01-005	93 square metres of hardstanding (south of Holland Brook) and public footpath (FP 29 167)	Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of rights as contained in a transfer dated 1 September 1989)

	Five Estuaries	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-005 cont'd		Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of rights granted by a Deed of Grant dated 12 April 1938) The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP (in respect of rights granted by a Deed of Grant dated 12 December 1938) Unknown (in respect of restrictive covenants contained in a Conveyance dated 6 November 1900) Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929)
01-006	3218 square metres of foreshore, sea wall and access track (east of Manor Way, Holland-or Sea) and public footpath (FP 29 167)	Unknown
01-007	21703 square metres of foreshore, sea wall, rock armour, grassland and sloping masonry (west of Second Avenue, Frinton-on-Sea)	Unknown (in respect of an Admitted Claim for C W Hayne dated 28 September 1933 and an Admitted claim for the executors of G F Beaumont dated 1961)
01-008	29586 square metres of foreshore, grassland, access track, sea wall, rock armour, sloping masonry and drains (west of Second Avenue, Frinton-on-Sea) public bridleway (BR 2 164) and public footpaths (FP 3 164, FP 41 164 and FP 29 167)	Unknown

	Five Estuaries	Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-009	9511 square metres of agricultural land (east of Manor Way, Holland-on-Sea)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights granted by a Deed of Grant dated 12 April 1938 and Deed of Grant dated 12 December 1938) Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX (in respect of rights reserved by a Conveyance dated 30 March 1990) Unknown (in respect of restrictive covenants as contained in Conveyance dated 6 November 1900) Unknown (in respect of restrictive covenants as contained in Conveyance dated 14 November 1929)
01-010	183349 square metres of golf course, ponds, public footbridges, drains, watercourse (Kirby Brook), public bridleway (BR 2 164) and public footpath (FP 3 164) (south of Short Lane, Frinton-on-Sea)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights granted by the Deed of Grant dated 12 December 1938 and Conveyance dated 16 May 1951) Unknown (in respect of covenants contained in a Conveyance dated 24 April 1920)
01-011	2423 square metres of watercourse (Kirby Brook)	Unknown

1	Five Estuaries	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-012	82161 square metres of agricultural land, footbridge and drain (west of Long Lane, Frinton- on-Sea)	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of rights granted by a Conveyance dated 30 March 1990) Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929)
02-002	937 square metres of access track (west of Second Avenue, Frinton-on-Sea) and public bridleway (BR 2 164)	Unknown
02-003	1231 square metres of access track, verges and private car park (west of Second Avenue, Frinton-on-Sea)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by a Transfer dated 5 December 2000) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 SAH (in respect of rights granted by the Deed of Grant dated 12 December 1938 and Conveyance dated 16 May 1951) Unknown (in respect of covenants contained in a Conveyance dated 16 March 1905) Unknown (in respect of restrictive covenants contained in a Conveyance dated 16 March 1905) Unknown (in respect of covenants contained in a Conveyance dated 24 April 1920)

	Five Estuaries	Offshore Wind Farm Development Consent Order	
	BOOK OF REFERENCE - PART 2		
		Counties of Essex and Suffolk	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973	
Land Plans		or 152 (3) of the Planning Act 2008	
02-004	55938 square metres of agricultural land and drains (west of Long Lane, Frinton-on-Sea)	Tendring District Council	
		Town Hall	
		Station Road	
		CLACTON-ON-SEA	
		Essex	
		C015 1SE	
		(in respect of rights granted by a Conveyance dated 30 March 1990)	
		Unknown	
		(in respect of restrictive covenants contained in a Conveyance dated 14 November 1929)	
02-005	122 square metres of footbridge and drain (south of Short Lane, Frinton-on-Sea)	Unknown	
02-006		Andrea Woods	
02-006	561 square metres of access track, agricultural land and hedgerow (west of Long Lane, Frinton-on-Sea)	9 Beaumont Close	
	FIIItoli-oli-sea)	WALTON ON THE NAZE	
		CO14 8TX	
		(in respect of a right of way)	
		(in respect of a fight of way)	
		Andrew Robinson	
		4 Polley Close	
		Kirby Cross	
		FRINTON-ON-SEA	
		C013 0UF	
		(in respect of a right of way)	
		Barry Cullum	
		56 Crown Road	
		CLACTON-ON-SEA	
		CO15 1AT	
		(in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-006 cont'd		Courtney Byrne S3 Cornflower Road Jaywick CLACTON-ON-SEA C015 25A (in respect of a right of way) Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH C012 5BY (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FRINTON-ON-SEA C013 0JT (in respect of a right of way) Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE NAZE C014 8RN (in respect of a right of way) Louisa Thatcher 104 Salisburg Road Holland-on-Sea CLACTON-ON-SEA CLACTON-ON-SEA CLACTON-ON-SEA CLACTON-ON-SEA	

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
02-006 cont'd		May King 2 Manor Road Great Holland FRINTON-ON-SEA C013 0JT (in respect of a right of way) Michelle Miller 2 Homerton Close CLACTON-ON-SEA C015 4U (in respect of a right of way) Pat Watson 25 Hillerest CLACTON-ON-SEA CLACTON-ON-SEA CO15 4HU (in respect of a right of way) Tanya Wheeler 2 St. Andrew Close Weeley CLACTON-ON-SEA CO16 9FX (in respect of a right of way) Fendring District Council Town Hall Station Road CLACTON-ON-SEA CSIS 5E (in respect of rights granted by a Conveyance dated 14 November 1929)

	Five Estuaries	Offshore Wind Farm Development Consent Order
	Five Estuaries	BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
02-006 cont'd		Victoria Oxland 8 Colthorpe Road CLACTON-ON-SEA CO15 4PU (in respect of a right of way) Wendy Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way) Yvonne Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)
02-007	5537 square metres of private roads (Short Lane and Long Lane), and public footpaths (FP 1 164 and FP 3 164) and public bridleway (BR 2 164) (south of Church Lane, Great Holland)	Andrea Woods 9 Beaumont Close WALTON ON THE NAZE CO14 8TX (in respect of a right of way) Andrew Robinson 4 Polley Close Kirby Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way) Barry Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-007 cont'd		Courtney Byrne S3 Corrflower Road Jaywick CLACTON-ON-SEA CO15 25A (in respect of a right of way) Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH CO12 5BY (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FRINTON-ON-SEA CO13 OJT (in respect of a right of way) Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE NAZE CO14 8RN (in respect of a right of way) Louisa Thatcher 104 Salisburg Road Hollandn-Sea CLACTON-ON-SEA CLACTON-ON-SEA CLACTON-ON-SEA	

	Five Estuaries Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-007 cont'd		Mary King         2 Manor Road         Great Holland         FRINTON-ON-SEA         C013 0JT         (in respect of a right of way)         Michelle Miller         22 Homerton Close         CLACTON-ON-SEA         C015 4UJ         (in respect of right of way)         Pat Watson         25 Hillcrest         CLACTON-ON-SEA         C015 4UJ         (in respect of a right of way)         Pat Watson         25 Hillcrest         CLACTON-ON-SEA         C015 4HU         (in respect of a right of way)         Tanya Wheeler         25 t. Andrews Close         Weeley         CLACTON-ON-SEA         C015 4FU         (in respect of a right of way)         Victoria Oxland         8 Colthorpe Road         CLACTON-ON-SEA         C015 4FU         (in respect of a right of way)         Victoria Oxland         8 Colthorpe Road         CLACTON-ON-SEA         C015 4FU         (in respect of a right of way)         Victoria Oxland         8 Colthorpe Road         CLACTON-ON-SEA <t< td=""></t<>	

r	Five Estuaries (	Dffshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk			
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973		
Land Plans		or 152 (3) of the Planning Act 2008		
02-007		Yvonne Cullum		
cont'd		56 Crown Road		
		CLACTON-ON-SEA		
		CO15 1AT		
		(in respect of a right of way)		
02-008	8503 square metres of agricultural land, drains and access track (west of Long Lane, Frinton	Andrea Woods		
		9 Beaumont Close		
	,	WALTON ON THE NAZE		
		CO14 8TX		
		(in respect of a right of way)		
		Andrew Robinson		
		4 Polley Close		
		Kirby Cross		
		FRINTON-ON-SEA		
		CO13 0UF		
		(in respect of a right of way)		
		Barry Cullum		
		56 Crown Road		
		CLACTON-ON-SEA		
		CO15 1AT		
		(in respect of a right of way)		
		Courtney Byrne		
		53 Cornflower Road		
		Jaywick		
		CLACTON-ON-SEA		
		CO15 2SA		
		(in respect of a right of way)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-008 cont'd		Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH C012 5P Y (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FRINTON-ON-SEA C013 0JT (in respect of a right of way) Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE NAZE C014 8RN (in respect of a right of way) Louisa Thatcher 104 Salisbury Road Holland-on-Sea CLACTON-ON-SEA C015 SLT (in respect of a right of way) Mazy King 2 Manor Road Great Holland FRINTON-ON-SEA C013 0JT (in respect of a right of way)	

Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
02-008 cont'd		Michelle Miller 22 Homerton Close CLACTON-ON-SEA C015 4UJ (in respect of a right of way) Pat Watson 25 Hillcrest CLACTON-ON-SEA C015 4HU (in respect of a right of way) Tanya Wheeler 2 St. Andrews Close Weeley CLACTON-ON-SEA C016 9FX (in respect of a right of way) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex C015 15E (in respect of rights granted by a Conveyance dated 30 March 1990) Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929) Victoria Oxland 8 Colthorpe Road CLACTON-ON-SEA CO15 4PU (in respect of a right of way)

	Five Estuaries Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-008		Wendy Robinson	
cont'd		4 Polley Close	
		Kirby Cross	
		FRINTON-ON-SEA	
		CO13 0UF (in respect of a right of way)	
		Yvonne Cullum	
		56 Crown Road CLACTON-ON-SEA	
		CO15 1AT	
		(in respect of a right of way)	
	56235 square metres of agricultural land, access track, hedgerow and drain (south of	Andrea Woods 9 Beaumont Close	
	Church Lane, Great Holland)	WALTON ON THE NAZE	
		CO14 8TX	
		(in respect of a right of way)	
		Andrew Robinson	
		4 Polley Close	
		Kirby Cross	
		FRINTON-ON-SEA	
		CO13 OUF	
		(in respect of a right of way)	
		Barry Cullum	
		56 Crown Road	
		CLACTON-ON-SEA CO15 1AT	
		(in respect of a right of way)	
		Courtney Byrne	
		53 Cornflower Road	
		Jaywick CLACTON-ON-SEA	
		CO15 2SA	
		(in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
. [		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-009 cont'd		Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARW/CH CO12 SPV (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way) Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE NAZE CO14 8RN (in respect of a right of way) Louisa Thatcher 104 Salisbury Road Holland-on-Sea CLACTON-ON-SEA CO15 SLT (in respect of a right of way)	

r		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
02-009 cont'd		Michelle Miller 22 Homerton Close CLACTON-ON-SEA CO15 4UJ (in respect of right of way) Pat Watson 25 Hillcrest CLACTON-ON-SEA CO15 4HU (in respect of a right of way) Tanya Wheeler 2 St. Andrews Close Weeley CLACTON-ON-SEA CO16 9FX (in respect of a right of way) Victoria Oxland 8 Cotthorpe Road CLACTON-ON-SEA CO15 4PU (in respect of a right of way) Wendy Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0JF (in respect of a right of way) Yvonne Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)

	Five Estuaries (	Dffshore Wind Farm Development Consent Order	
	BOOK OF REFERENCE - PART 2		
		Counties of Essex and Suffolk	
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973	
Land Plans		or 152 (3) of the Planning Act 2008	
02-010		Andrea Woods	
	of Clacton Road, B1032)	9 Beaumont Close	
		WALTON ON THE NAZE	
		CO14 8TX	
		(in respect of a right of way)	
		Andrew Robinson	
		4 Polley Close	
		Kirby Cross	
		FRINTON-ON-SEA	
		CO13 0UF	
		(in respect of a right of way)	
		Barry Cullum	
		56 Crown Road	
		CLACTON-ON-SEA	
		CO15 1AT	
		(in respect of a right of way)	
		Courtney Byrne	
		53 Cornflower Road	
		Jaywick	
		CLACTON-ON-SEA	
		CO15 2SA	
		(in respect of a right of waγ)	
		Julie Watson	
		3 Skighaugh Cottages	
		Clacton Road	
		Stones Green	
		HARWICH	
		CO12 5BY	
		(in respect of a right of way)	

	Five Estuaries	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
02-010 cont'd		Katy O'Donnell 2 Manor Road
cont d		Great Holland
		FRINTON-ON-SEA
		CO13 OJT
		(in respect of a right of way)
		Lindsey-Cher Johnson
		4 Rochford Way
		WALTON ON THE NAZE CO14 8RN
		(in respect of a right of way)
		Louisa Thatcher
		104 Salisbury Road
		Holland-on-Sea
		CLACTON-ON-SEA
		(in respect of a right of way)
		Mazy King
		2 Manor Road
		Great Holland
		FRINTON-ON-SEA CO13 0JT
		(in respect of a right of way)
		Michelle Miller
		22 Homerton Close
		CLACTON-ON-SEA
		CO15 4UJ (in respect of a right of way)
		(in respect of a fight of way)
		Pat Watson
		25 Hillcrest
		CLACTON-ON-SEA CO15 4HU
		(in respect of a right of way)

		Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
02-010 cont'd		Tanya Wheeler 2 St. Andrews Close Weeley CLACTON-ON-SEA CO16 9FX (in respect of a right of way) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 15E (in respect of rights granted by a Conveyance dated 30 March 1990) Unknown (in respect of rights granted by a Conveyance dated 14 November 1929) Victoria Oxland 8 Cothorpe Road CLACTON-ON-SEA CO15 4PU (in respect of a right of way) Wendy Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 UF (in respect of a right of way) Vicone Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-002	43609 square metres of agricultural land, copse and hedgerow (east of Clacton Road, B1032)	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019)	
03-003	14204 square metres of agricultural land and hedgerow (east of Clacton Road, B1032)	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019)	
03-004	1568 square metres of public highway, verges and field accessway (Clacton Road, B1032)	Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-004A	297 square metres of public highway, verges and field accessway (Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of rights of access) Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 0NG (in respect of rights of access) Unknown	
03-005	92 square metres of agricultural land (east of Clacton Road, B1032)	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019)	

l	Five Estuaries	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973
Land Plans		or 152 (3) of the Planning Act 2008
03-006	16 square metres of agricultural land and hedgerow (east of Clacton Road, B1032)	Alison Margaret Brown
		Dairy House Farm
		Little Clacton Road
		Great Holland
		FRINTON-ON-SEA
		Essex
		CO13 0EX
		(in respect of rights contained in a Transfer dated 4 February 2019)
		Douglas Kenneth Brown
		Dairy House Farm
		Little Clacton Road
		Great Holland
		FRINTON-ON-SEA
		Essex
		C013 0EX
		(in respect of rights contained in a Transfer dated 4 February 2019)
03-007	1340 square metres of public highway, verges, hedgerow and access splay (Clacton Road,	Unknown
	B1032)	
03-007A	1640 square metres of public highway, verges, hedgerow and access splay (Clacton Road,	Unknown
	B1032)	
03-008	23 square metres of agricultural land and hedgerow (west of Clacton Road, B1032)	John William Glover
		Great Holland Lodge
		Clacton Road
		Great Holland
		FRINTON-ON-SEA
		Essex
		C013 0JU
		(in respect of rights reserved as contained in a Deed dated 27 March 1951)

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-008 cont'd		Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)	
03-009	740 square metres of public highway and verges (Clacton Road, B1032)	Unknown	
03-011	12346 square metres of agricultural land and hedgerow (west of Clacton Road, B1032)	John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 OJU (in respect of rights reserved as contained in a Deed dated 27 March 1951) Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 OJU (in respect of rights reserved as contained in a Deed dated 27 March 1951)	
03-012	10870 square metres of agricultural land and hedgerows (west of Clacton Road, B1032)	John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)	

1	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-012 cont'd		Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)	
03-013	58 square metres of agricultural land and hedgerow (west of Clacton Road, B1032)	John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951) Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)	
03-014	56162 square metres of agricultural land hedgerow (Little Clacton Road, Great Holland)	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights as contained in a Transfer dated 4 February 2019 and in a Conveyance dated 22 December 1982)	

I	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-014 cont'd		Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respects of rights contained in a Transfer dated 4 February 2019)	
03-015	3759 square metres of agricultural land, access track, hardstanding, hedgerow, access splay and public footpath (FP 5 164) (Dairy House Farm, Little Clacton Road, Great Holland)		
03-016	940 square metres of access track and public footpath (FP 7 164 and FP 10 164) (north of Little Clacton Road, Great Holland)	Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access)	

1	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-017	27 square metres of access track (north of Little Clacton Road, Great Holland) and public footpath (FP 11 164)	Brenda Jane Gibson-Wynes Flat 11 Linkswood Compton Place Road EASTBOURNE BN21 1EE (in respect of covenants by a Transfer dated 5 November 1982) Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access) Unknown (in respect of rights granted by a Deed 23 August 1968)	
03-018	249 square metres of access track (north of Little Clacton Road, Great Holland) and public footpath (FP 10 164)	Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access)	
04-002	1004 square metres of public highway, verges, access splay (Little Clacton Road) and public footpath (FP 6 164)	Unknown	
04-003	96 square metres of agricultural land (east of Mill Lane, Little Clacton Road)	Unknown	
04-011	19 square metres of agricultural land (east of Great Holland Mill, Little Clacton Road) and public footpath (FP 38 164)	Unknown	
04-015	3 square metres of agricultural land (east of Great Holland Mill, Little Clacton Road)	Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-016	185 square metres of access track (north of Little Clacton Road, Great Holland) and public footpath (FP 10 164)	Brenda Jane Gibson-Wynes Flat 11 Linkswood Compton Place Road EASTBOURNE BN21 1EE (in respect of covenants by a Transfer dated 5 November 1982) Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access) Unknown (in respect of rights granted by a Deed 23 August 1968)	
04-017	1893 square metres of agricultural land, access track and public footpaths (FP 10 164, FP 38 164 and FP 11 164) (east of Great Holland Mill, Little Clacton Road)	Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access)	
04-018	17 square metres of access track (west of Pork Lane, Great Holland) and public footpath (FF 10 164)	Unknown	
04-019	552 square metres of agricultural land, copse, hedgerow (west of Pork Lane, Great Holland) and public footpath (FP 10 164)	Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of statutory rights as contained in a Deed dated 19 December 1966) Unknown (in respect of right to enter to maintain a water main and in respect of restrictive covenants as stated in a Deed dated 16 July 1968)	

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-020	38061 square metres of agricultural land, copse and hedgerow (west of Pork Lane, Great Holland)	Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of statutory rights as contained in a Deed dated 19 December 1966) Unknown (in respect of right to enter to maintain a water main and in respect of restrictive covenants as stated in a Deed dated 16 July 1968)	
05-001	8773 square metres of agricultural land and hedgerow (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Unknown	
05-002	43670 square metres of agricultural land, woodland, brook and access track at Birch Hoe Farm (west of Pork Lane, Great Holland)	Unknown (in respect of rights as stated in Conveyance dated 31 March 1982)	
05-003	129 square metres of agricultural land and hedgerow (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-004	460 square metres of access track (west of Pork Lane, Great Holland)	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of right of access as stated in a Conveyance dated 16 July 1968) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of right of access as stated in a Conveyance dated 16 July 1968) Unknown (in respect of rights reserved as contained in a Conveyance dated 16 Huly 1968)	
05-005	290 square metres of access track and access splay (west of Pork Lane, Great Holland)	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of right of access as stated in a Conveyance dated 16 July 1968) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of right of access as stated in a Conveyance dated 16 July 1968)	

r	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-006	3 square metres of agricultural land and access splay (west of Pork Lane, Great Holland)	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of rights of access) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of rights of access) Unknown	
05-007	5689 square metres of agricultural land, access track, brook and hedgerow (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Unknown	
05-008	5609 square metres of agricultural land, brook and hedgerow (west of Pork Lane, Great Holland)	Unknown (in respect of rights granted by a Conveyance dated 29 September 1954) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	

	Five Estuaries Offshore Wind Farm Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
05-008 cont'd		Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)		
05-009	982 square metres of agricultural land, brook and hedgerow (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Unknown		
05-010	20 square metres of agricultural land, brook and hedgerow (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Unknown		
05-011	25 square metres of agricultural land and brook (west of Pork Lane, Great Holland)	Unknown (in respect of rights granted by a Conveyance dated 29 September 1954) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998) Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)		
	889 square metres of agricultural land and brook (south of Thorpe Park Lane, Thorpe-le- Soken)	Unknown (in respect of rights granted by a Conveyance dated 29 September 1954) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)		

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-012 cont'd		Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)	
05-013	30555 square metres of agricultural land and woodland (south of Thorpe Park Lane, Thorpe le-Soken)	Unknown (in respect of rights granted by a Conveyance dated 29 September 1954) Unknown	
		(in respect of rights reserved by a Conveyance dated 14 March 1998)	
		Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)	
05-014	3558 square metres of railway (Thorpe-le-Soken and Kirby Cross) and scrubland (south of Thorpe Park Lane)	Unknown	
05-015	1021 square metres of agricultural land (south of Thorpe Park Lane, Thorpe-le-Soken)	Unknown (in respect of rights granted by a Conveyance dated 29 September 1954)	
		Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
		Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)	
05-016	103 square metres of railway (Thorpe-le-Soken and Kirby Cross) access track and scrubland (south of Thorpe Park Lane)	Unknown	
05-017	5373 square metres of access track (south of Thorpe Park Lane)	Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
05-018	408 square metres of private road (Thorpe Park Lane)	Andrzej Tomasz Wiecek 1 Thorpe Park Cottages Thorpe Park Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 OHN (in respect of right of access)	

		Five Estuaries Offshore Wind Farm Development Consent Order
BOOK OF REFERENCE - PART 2		
		Counties of Essex and Suffolk
umber on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973
and Plans	Description of Land	or 152 (3) of the Planning Act 2008
05-018		Annis Roberts
cont'd		Thorpe Park House
		Thorpe Park Lane
		Thorpe-le-Soken
		CLACTON-ON-SEA
		C016 0HN
		(in respect of rights of access)
		Benjamin Mark Worrallo
		3 Thorpe Park Cottages
		Thorpe Park Lane
		Thorpe-le-Soken
		CLACTON-ON-SEA
		C016 0HN
		(in respect of rights of access)
		Georgios Tsaousellis
		4 Thorpe Park Cottages
		Thorpe Park Lane
		Thorpe-le-Soken
		CLACTON-ON-SEA
		C016 0HN
		(in respect of rights of access)
		John Whiten
		2 Thorpe Park Cottages
		Thorpe Park Lane
		Thorpe-le-Soken
		CLACTON-ON-SEA
		CO16 0HN
		(in respect of rights of access)

r	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-018 cont'd		Justyna Magdalena Wiecek 1 Thorpe Park Lane Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access) Rhiannon Wheeler 4 Thorpe Park Cottages Thorpe Park Cottages Thorpe Park Cottages Thorpe Park Lane Thorpe Park Lane Thorpe Park Cottages Sam Worrallo 3 Thorpe Park Cottages Thorpe Park Lane Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access) Saint Worrallo 3 Thorpe Park Cottages Thorpe Park Lane Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access) Shirley Whiten 2 Thorpe Park Cottages Thorpe Infights of access) Shirley Whiten 2 Thorpe Park Cottages Thorpe Infights of access) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
05-019	56801 square metres of agricultural land (south of Thorpe Park Lane)	Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-022	61 square metres of agricultural land (west of Pork Lane, Great Holland)	Unknown	
05-023	3249 square metres of agricultural land and hedgerow (south of Thorpe Road, B1035)	Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
05-024	38017 square metres of agricultural land, woodland (Parklane Grove) and hedgerow (south of Thorpe Road, B1035)	Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
05-025	232 square metres of agricultural land and hedgerow (south of Thorpe Road, B1035)	Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
06-001	3067 square metres of public highway, verges and lay-by (Thorpe Road, B1033)	Unknown	
06-005	23298 square metres of agricultural land and hedgerow (south of Thorpe Road, B1035)	Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
06-006	249 square metres of agricultural land (south of Thorpe Road, B1035)	Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
06-007	224 square metres of agricultural land (west of Thorpe Road, B1033)	Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
06-008	361 square metres of verge and lay-by (Thorpe Road, B1033)	Unknown	
06-014	2785 square metres of public highway, verges, hardstanding and drain (Sneating Hall Lane, B1034)	Unknown	
06-018	2430 square metres of public highway, verges and access splay (Damants Farm Lane, Thorpe-le-Soken)	Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-007	751 square metres of woodland (south of Walton Road, Thorpe-Le-Soken)	M Scott Property Group Limited Suite 5 Oyster House Severalls Lane Colchester CO4 9PD (in respect of a restriction dated 22 March 2021)	
07-011	128000 square metres of agricultural land and hedgerow (south of Walton Road) and public footpath (FP 13 180)	M Scott Property Group Limited Suite 5 Oyster House Severalls Lane Colchester CO4 9PD (in respect of a restriction dated 22 March 2021)	
07-013	3120 square metres of public highway (Landermere Road, B1414)	Unknown	
08-001	7 square metres of access splay (east of Landermere Road)	Unknown	
08-004	70 square metres of hedgerow, verge and access splay (Landermere Road, B1414)	Unknown	
08-009	1029 square metres of agricultural land and public footpath (FP 7 180)	Unknown (in respect of covenants contained in various Conveyances)	
08-010	418 square metres of private road and verges (Lonsdale Road, Thorpe-Le-Soken)	Andrew William Bacon Green Lane Farm Colchester Road Weeley CLACTON-ON-SEA Essex CO16 9AD (in respect of rights of way and maintenance)	

	Five Estuar	ies Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
08-010 cont'd		Benjamin Furness         7 Crownfields         Crown Street         Dedham         COLCHESTER         CO7 6AT         (in respect of rights of way and maintenance)         Brendan Malachy Gormley         Gunfleet         Lonsdale Road         Thorpe-le-Soken         CLACTON-ON-SEA         Colo 0.F         (in respect of rights of way and maintenance)         Colin Ernest Anson         Framble Barn         Lonsdale Road         Thorpe-le-Soken         CLACTON-ON-SEA         Colin Ernest Anson         Framble Barn         Lonsdale Road         Thorpe-le-Soken         CLACTON-ON-SEA         Colin O.F         (in respect of rights of way and maintenance)         Delicia Maria Ransom         The Old Cottage         The Street         Copdock         IPSWICH         IPS 3HS         (In respect of rights of way and maintenance)         Dominic Furness         Portuniemntie 21         Hameenlinna         1320         FINLAND         (In respect of rights of way and maintenance)

	Five Estuaries	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
08-010 cont'd		Gwendolyn Rose Batley 87 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LW (in respect of rights of way and maintenance) Jane El/zabeth Anson Framble Barn Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance) John Frederick Peirson Blacksmiths Farm Harwich Road Beaumont CLACTON-ON-SEA CO16 0AS (in respect of rights of way and maintenance) Julian Furness Clambridge House Amberfield Drive Naton IPSVICH IP10 0GQ (in respect of rights of way and maintenance) Linda Draper Hamford Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance) Linda Draper Hamford Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance)

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
08-010 cont'd		Margaret Shona Gormley         Gunfleet         Lonsdale Road         Thorpe-le-Soken         CLACTON-ON-SEA         C016 0LF         (in respect of rights of way and maintenance)         Mark Terry Sangwine         87 Landermere Road         Thorpe-le-Soken         CLACTON-ON-SEA         C016 0LF         (in respect of rights of way and maintenance)         Michael David Pollard         89 Landermere Road         Thorpe-le-Soken         CLACTON-ON-SEA         C016 0LF         (in respect of rights of way and maintenance)         Michael David Pollard         89 Landermere Road         Thorpe-le-Soken         CLACTON-ON-SEA         C016 0LF         (in respect of rights of way and maintenance)         Michael William Gay         Trinity         Lonsdale Road         Thorpe-le-Soken         CLACTON-ON-SEA         C016 0LF         (in respect of rights of way and maintenance)         Michael William Gay         Trinity         Lonsdale Road         Thorpe-le-Soken         CLACTON-ON-SEA         C016 0LF         (i

	Fivo Fctu:	aries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
08-010 cont'd		Principal Homes Limited Oak House Church Lane Great Holland FRINTON-ON-SEA Essex CO13 OUS (in respect of rights of way and maintenance) Sharon Susan Gay Trinity Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 OUF (in respect of rights of way and maintenance) Simon Stone Jimilda Lonsdale Road Thorpe-ie-Soken CLACTON-ON-SEA CO16 OUF (in respect of rights of way and maintenance) Susan Mary Pollard 89 Landermere Road Thorpe-ie-Soken CLACTON-ON-SEA CO16 OUF (in respect of rights of way and maintenance) Susan Mary Pollard 89 Landermere Road Thorpe-ie-Soken CLACTON-ON-SEA CO16 ONA (in respect of rights of way and maintenance) Unknown
08-011	17889 square metres of agricultural land (west of Landermere Road)	Unknown (in respect of covenants contained in various Conveyances)

<b></b>	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-013	31370 square metres of agricultural land and hedgerows (south of Golden Lane)	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 OLE (in respect of rights reserved by the Transfer dated 21 July 2010) Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 OLE (in respect of rights reserved by a Transfer dated 21 July 2010)	
08-014	2138 square metres of agricultural land and hedgerow (south of Golden Lane)	Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN (in respect of rights granted by a Deed dated 11 August 1967) Unknown (in respect of rights contained in a Conveyance dated 18 February 1952)	
08-015	405 square metres of agricultural land (east of Golden Lane)	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by the Transfer dated 21 July 2010)	

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-015 cont'd		Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by a Transfer dated 21 July 2010)	
08-016	176 square metres of agricultural land (east of Golden Lane)	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by the Transfer dated 21 July 2010) Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by a Transfer dated 21 July 2010)	
08-017	56 square metres of agricultural land (east of Golden Lane)	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by the Transfer dated 21 July 2010)	

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-017 cont'd		Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 OLE (in respect of rights reserved by a Transfer dated 21 July 2010)	
08-018	1801 square metres of public highway and verges (Golden Lane)	Unknown	
08-020	4361 square metres of paddock and copse (east of Tendring Road, B1035)	Julie Wendy Thompson 92 Station Road CLACTON-ON-SEA Essex CO15 1SG (in respect of rights granted by a Transfer dated 28 May 2015)	
08-021	777 square metres of paddock, hedgerow and public footpath (FP 3 180) (east of Tendring Road, B1035)	Unknown	
08-022	2452 square metres of paddock and hedgerow (east of Tendring Road, B1035)	Julie Wendy Thompson 92 Station Road CLACTON-ON-SEA Essex CO15 1SG (in respect of rights granted by a Transfer dated 28 May 2015)	
09-002	1910 square metres of public highway and verges (Tendring Road, B1035)	Unknown	
09-004	62 square metres of public highway and verges (Tendring Road, B1035)	Unknown	
09-007	852 square metres of public highway and verges (Tendring Road, B1035)	Unknown	

Number on Land Plans 09-008 13	Description of Land	Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973
Land Plans	Description of Land	Counties of Essex and Suffolk
Land Plans	Description of Land	
00.000 17		or 152 (3) of the Planning Act 2008
	13526 square metres of agricultural land (south of Thorpe Road and west of Tendring Road, B1035)	Laura Lilian Lord The Barn Bradley Hall Farm Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH (in respect of rights reserved as contained in a Transfer dated 25 October 2017)
09-009 17	1798 square metres of public highway and verges (Thorpe Road)	Unknown
09-012 73	737 square metres of public highway and verges (Swan Road)	Unknown
09-013 6	5 square metres of hedgerow (west of Swan Road)	Unknown
	4271 square metres of garden (2 Barkers Hall Cottages), hedgerow and paddock (west of Swan Road)	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN (in respect of rights as granted by a Transfer dated 15 June 2006)
09-015 33	339 square metres of agricultural land (north of Thorpe Road, B1035)	Unknown
09-017 91	9105 square metres of agricultural land (north of Thorpe Road)	Unknown
09-018 22	22937 square metres of agricultural land (north of Thorpe Road)	Unknown

r	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-023	581 square metres of agricultural land (north of Thorpe Road, B1035)	Rob Long Jasmine Cottage Tendring Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0AA (in respect of rights of access) Unknown	
09-024	42538 square metres of agricultural land and hedgerow (north of Thorpe Road, B1035)	David Milburn Cannell The Old Barn Thorpe Road Tendring CLACTON-ON-SEA Essex CO16 9AR (in respect of sporting rights granted by a Transfer dated 7 May 2004)	
10-002	515 square metres of agricultural land (north of Thorpe Road, B1035)	David Milburn Cannell The Old Barn Thorpe Road Tendring CLACTON-ON-SEA Essex CO16 9AR (in respect of sporting rights granted by a Transfer dated 7 May 2004)	
10-006	693 square metres of agricultural land, woodland and private access track (north of Thorpe Road, B1035)	lan Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)	

<b></b>	Five Estuaries Offshore Wind Farm Development Consent Order	
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
10-006 cont'd		Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)
10-007	4639 square metres of agricultural land (north of Thorpe Road, B1035)	lan Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 OBS (in respect of rights granted by a Transfer dated 9 June 2011) Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 OBS (in respect of rights granted by a Transfer dated 9 June 2011)
	2056 square metres of private road (known as Lodge Lane), private access track and public footpath (FP 8 179)	lan Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 OBS (in respect of rights granted by a Transfer dated 9 June 2011)

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
10-008 cont'd		Joanna Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA C016 0BS (in respect of rights of access) Stephen Nigel Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA C016 0BS (in respect of rights of access) Susan Beverley Patch Tanzara Lodge Lane Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex C016 0BS (in respect of rights of access)
10-009	126 square metres of private road (known as Lodge Lane)	Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 OBS (in respect of rights granted by a Transfer dated 9 June 2011)

	Eivo Ectuarios	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973
Land Plans		or 152 (3) of the Planning Act 2008
10-009		Joanna Mangham
cont'd		The Lodge
		Lodge Lane
		Tendring
		CLACTON-ON-SEA
		CO16 OBS
		(in respect of rights of access)
		Stephen Nigel Mangham
		The Lodge
		Lodge Lane
		Tendring
		CLACTON-ON-SEA
		C016 0BS
		(in respect of rights of access)
		Susan Beverley Patch
		Tanzara
		Lodge Lane
		Tendring
		CLACTON-ON-SEA
		Essex
		CO16 OBS
		(in respect of rights granted by a Transfer dated 9 June 2011)
10-011	22499 square metres of private road (known as Lodge Lane) and agricultural land (east of	lan Martin Patch
	Lodge Lane)	Tanzara
		Lodge Lane
		Tendring
		CLACTON-ON-SEA
		Essex
		CO16 OBS
		(in respect of rights granted by a Transfer dated 9 June 2011)
		(in respect of rights granted by a Transfer dated 9 June 2011)

		Five Estuaries Offshore Wind Farm Development Consent Order			
		BOOK OF REFERENCE - PART 2			
	Counties of Essex and Suffolk				
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
10-011 cont'd		Joanna Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 OBS (in respect of rights of access) Stephen Nigel Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 OBS (in respect of rights of access) Susan Beverley Patch Tanzara Lodge Lane Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 OBS (in respect of rights of access)			
10-012	175 square metres of private road (known as Lodge Lane)	Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)			

	Five Ectuaries	Offshore Wind Farm Development Consent Order
	Five Estuaries	BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
10-012 cont'd		Joanna Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 085 (in respect of rights of access) Stephen Nigel Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access) Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)
10-013	120236 square metres of agricultural land (east of Heath Road) and public footpaths (FP 3 179 and FP 8 179)	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of the unknown covenants contained in Conveyance dated 29 November 1982) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 15 October 1954) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 6 October 1956)

r	Five Estuaries	Offshore Wind Farm Development Consent Order			
		BOOK OF REFERENCE - PART 2			
	Counties of Essex and Suffolk				
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
11-003	578 square metres of private access track (south of Wolves Hall Lane)	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of the unknown covenants contained in Conveyance dated 29 November 1982) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 15 October 1954) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 6 October 1956)			
11-004	10 square metres of private access track (south of Wolves Hall Lane)	Unknown			
11-007	1879 square metres of public highway and verge (Wolves Hall Lane)	Unknown			
11-012	51258 square metres of agricultural land (south of Stones Green Road) and public footpath (FP 1 179)	Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 15 October 1954)			
11-013	2959 square metres of agricultural land (south of Stones Green Road)	Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 15 October 1954)			
11-015	39 square metres of public highway and verge (Stones Green Road)	Unknown			
11-016	244 square metres of agricultural land (north of Stones Green Road)	Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)			
11-017	205 square metres of agricultural land and hedgerow (north of Stones Green Road)	Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)			
11-018	3001 square metres of public highway and verges (Stones Green Road)	Unknown			

	Five Estuaries	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-019	28 square metres of agricultural land (north of Stones Green Road)	Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)
11-020	1069 square metres of agricultural land and public footpath (FP 31 183) (north of Stones Green Road)	Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)
12-002	680 square metres of agricultural land (north of Stones Green Road)	Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)
12-003	1825 square metres of agricultural land (north of Stones Green Road)	Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)
12-004	103132 square metres of agricultural land and public footpaths (FP 32 183 and FP 31 183) (north of Stones Green Road)	Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)
12-005	2021 square metres of agricultural land, private track and public footpaths (FP 32 183, FP 14 183 and FP 31 183) (north of Stones Green Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)
12-006	2123 square metres of agricultural land, private access track to Hempstalls Farm and public footpath (FP 15 183) (south of A120)	Aurlius Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-006 cont'd		Chris Driver Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE C011 2NZ (in respect of rights of access to Hempstalls Farm) Chris Mudd Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE C011 2NZ (in respect of rights of access to Hempstalls Farm) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER C07 8SE (in respect of rights of access) Harry Weavers Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE C12 XZ (in respect of rights of access to Hempstalls Farm)	

Five Estuaries Offshore Wind Farm Development Consent Order			
	BOOK OF REFERENCE - PART 2		
Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-006 cont'd		Jay McDermot Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Lewis Clarke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Lisa Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS (in respect of rights of access to Hempstalls Farm) Mark Taylor Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS (in respect of rights of access to Hempstalls Farm) Mark Taylor Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE ClatUR (in respect of rights of access to Hempstalls Farm)	

	Five Estuaries (	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
12-006 cont'd		Nick Bowingcoke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)
12-007	92705 square metres of agricultural land and hedgerow (south of A120)	Unknown
12-008	279 square metres of public highway lay-by and verge (B1035)	Tungsten Colchester Limited Gateway House 4 Penman Way Grove Park Enderby Leicester LE19 ISY (in respect of a commercial interest for development) Unknown
12-009	2728 square metres of public highway and verge (B1035)	Unknown
12-010	371 square metres of agricultural land, private access track and public footpath (FP 37 138) (east of B1035)	Unknown

	Five Estuaries	Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
12-011	1061 square metres of agricultural land and hedgerow (east of B1035)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)
12-012	1178 square metres of public highway and verges (A120) (excluding all the interests of the National Highways Limited)	Unknown
12-013	63308 square metres of agricultural land, hedgerow and private access track (north of A120)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge)

Five Estuaries Offshore Wind Farm Development Consent Order		
BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
12-013 cont'd		Arron CK8 and Scaffold Yard Bradifield Lodge Clacton Road MANNGINGTREE Essex C011 2NX (In respect of CK8 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree C011 2NS (In respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree C011 2NS (In respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Blake Birchill Unit 12 Bradfield Lodge (In respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge (In respect of access to Unit 12 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge (In respect of access to Unit 12 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge (In respect of access to Unit 12 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge (In respect of access to Unit 12 Bradfield Lodge)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-013 cont'd		Carl TarrantUnit 2Bradifield LodgeClacton RoadManningtreeCO11 2SN(in respect of access to Unit 2 Bradfield Lodge)Darren SmithUnits 17 and 18aBradfield LodgeClacton RoadMANNGINGTREECO11 2NS(in respect of access to Units 17 and 18a Bradfield Lodge)Dave SayerUnits 1 and 4Bradfield LodgeClacton RoadMANNGINGTREECO11 2NS(in respect of access to Units 17 and 18a Bradfield Lodge)Dave SayerUnits 1 and 4Bradfield LodgeClacton RoadMANNGINGTREECO11 2NS(in respect of access to Units 1 and 4 Bradfield Lodge)East Anglian Farm RidesManor BungalowChurch RoadLittle BentieyCOLCHESTERCO7 8E(in respect of rights of access)	

Five Estuaries Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
12-013 cont'd		Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge) Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Clacton Road Manningtree Clacton Road

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-013 cont'd		Kanye Ault         Unit 15         Bradfield Lodge         MANNINGTREE         Essex         C011 2NS         (in respect of Unit 15 Bradfield Lodge)         Les Troller         Unit 10         Bradfield Lodge         Clacton Road         Manningtree         C011 2NS         (in respect of Unit 10 Bradfield Lodge)         Kinke Cummins         Unit 23         Bradfield Lodge         Clacton Road         Mick Cummins         Unit 23         Bradfield Lodge         Clacton Road         MANNIGNETEE         C011 2NS         (in respect of Unit 10 Bradfield Lodge)         Mick Cummins         Unit 23         Bradfield Lodge         Clacton Road         MANNGINGTREE         C011 2NS         (in respect of Unit 23 Bradfield Lodge)         Mick Innis         Units 8 and 16         Bradfield Lodge         Clacton Road         MANNGINGTREE         C011 2NS         (in respect of Unit 23 Bradfield Lodge)         Mick Innis         Units 8 and 16	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-013 cont'd		Peter Barrat         Unit 3         Bradifield Lodge         MANNINGTREE         Essex         C011 2NS         (in respect of Unit 3 Bradfield Lodge)         Rickman         Unit 21         Bradfield Lodge         Clacton Road         Manningtree         C011 2NS         (in respect of access to Unit 21 Bradfield Lodge)         S Smith         Unit 7         Bradfield Lodge         Clacton Road         Manningtree         C011 2NS         (in respect of access to Unit 21 Bradfield Lodge)         S Smith         Unit 7         Bradfield Lodge         Claton Road         Manningtree         C011 2NS         (in respect of access to Unit 7 Bradfield Lodge)         Scott Adams         CK7         Bradfield Lodge         MANNINGTREE         Essex         CN1 2NS         (in respect of access to CK7 Bradfield Lodge)         (Cot1 2NS         (in respect of access to CK7 Bradfield Lodge)	

	Five Estuarie:	s Offshore Wind Farm Development Consent Order			
		BOOK OF REFERENCE - PART 2			
	Counties of Essex and Suffolk				
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
12-013 cont'd		StairformFeed Store 1Bradfield LodgeClacton RoadManningtreeCO11 2NS(in respect of access to Feed Store 1 Bradfield Lodge)Tony RobinsonUnit 14Bradfield LodgeClacton RoadManningtreeClacton RoadManningtree(in respect of access to Unit 14 Bradfield Lodge)			
12-014	3224 square metres of agricultural land and private access track (north of A120)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge)			

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
12-014		Albert Du Plessis
cont'd		Unit 19
some a		Bradfield Lodge
		Clacton Road
		Manningtree
		C011 2NS
		(in respect of access to Unit 19 Bradfield Lodge)
		A
		Arron
		CK8 and Scaffold Yard Bradfield Lodge
		Clacton Road
		MANNGINGTREE
		Essex
		CO11 2NX
		(in respect of access to CK8 and Scaffold Yard Bradfield Lodge)
		Arron Ingram
		Unit 18b
		Bradfield Lodge
		Clacton Road
		Manningtree
		CO11 2NS
		(in respect of access to Unit 18b Bradfield Lodge)
		Ben Ross
		Unit 5
		Bradfield Lodge
		Clacton Road
		Manningtree
		CO11 2NS
		(in respect of access to Unit 5 Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
12-014 cont'd		Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
umber on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973
and Plans	Description of Land	or 152 (3) of the Planning Act 2008
12-014		East Anglian Farm Rides
cont'd		Manor Bungalow
		Church Road
		Little Bentley
		COLCHESTER
		CO7 8SE
		(in respect of rights of access)
		Jason Childs
		Units 9 and 13
		Bradfield Lodge
		MANNINGTRE
		Essex
		C011 2NS
		(in respect of access to Units 9 and 13 Bradfield Lodge)
		Jason Hinsen
		Feed Store 2
		Bradfield Lodge
		Clacton Road
		Manningtree
		C011 2NS
		(in respect of access to Feed Store 2 Bradfield Lodge)
		Jim Storey
		Unit 6
		Bradfield Lodge
		Clacton Road
		Manningtree
		CO11 2NS
		(in respect of access to Unit 6 Bradfield Lodge)
		· · · · · · · · · · · · · · · · · · ·

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
12-014 cont'd		Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 15 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of Unit 10 Bradfield Lodge) Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNIGINGTREE CO11 2NS (in respect of Unit 10 Bradfield Lodge) Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNIGINGTREE Clacton Road

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
lumber on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973
and Plans	Description of Land	or 152 (3) of the Planning Act 2008
12-014		Mick Innis
cont'd		Units 8 and 16
		Bradfield Lodge
		Clacton Road
		MANNGINGTREE
		C011 2NS
		(in respect of Units 8 and 16 Bradfield Lodge)
		Peter Barrat
		Unit 3
		Bradfield Lodge
		MANNINGTREE
		Essex
		C011 2NS
		(in respect of Unit 3 Bradfield Lodge)
		Rickman
		Unit 21
		Bradfield Lodge
		Clacton Road
		Manningtree
		C011 2NS
		(in respect of access to Unit 21 Bradfield Lodge)
		S Smith
		Unit 7
		Bradfield Lodge
		Clacton Road
		Manningtree
		CO11 2NS
		(in respect of access to Unit 7 Bradfield Lodge)
		· · · · · · · · · · · · · · · · · · ·

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
-		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
12-014 cont'd		Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)
13-001 230 square m	etres of private access track (north of A120)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-001 cont'd		Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of access to CK7 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road MANNGINGTEE Essex CO11 2NX (in respect of access to CK7 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-001 cont'd		Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2	
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-001 cont'd		Dave SayerUnits 1 and 4Bradfield LodgeClacton RoadMANNGINGTREECO11 2NS(in respect of access to Units 1 and 4 Bradfield Lodge)East Anglian Farm RidesManor BungalowChurch RoadLittle BentleyCOLCHESTERCO7 85E(in respect of rights of access)Jason ChildsUnits 9 and 13Bradfield LodgeEssexCO11 2NS(in respect of access to Units 9 and 13 Bradfield Lodge)Jason HinsenFeed Store 2Bradfield LodgeClacton RoadManningtreeCO11 2NS(in respect of access to Feed Store 2 Bradfield Lodge)	

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-001 cont'd		Jim Storey Unit 6 Bradfield Loge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of unit 20 Bradfield Lodge) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 15 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree Clacton Road

		Five Estuaries Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk			
lumber on and Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
13-001		Mick Cummins		
cont'd		Unit 23		
		Bradfield Lodge		
		Clacton Road		
		MANNGINGTREE		
		CO11 2NS		
		(in respect of Unit 23 Bradfield Lodge)		
		Mick Innis		
		Units 8 and 16		
		Bradfield Lodge		
		Clacton Road		
		MANNGINGTREE		
		C011 2NS		
		(in respect of Units 8 and 16 Bradfield Lodge)		
		Peter Barrat		
		Unit 3		
		Bradfield Lodge		
		MANNINGTREE		
		Essex		
		CO11 2NS		
		(in respect of Unit 3 Bradfield Lodge)		
		Rickman		
		Unit 21		
		Bradfield Lodge		
		Clacton Road		
		Manningtree		
		CO11 2NS		
		(in respect of access to Unit 21 Bradfield Lodge)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-001 cont'd		S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge) Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access in Unit 14 Bradfield Lodge)	

r	Five Estuaries Offshore Wind Farm Development Consent Order				
		BOOK OF REFERENCE - PART 2			
	Counties of Essex and Suffolk				
Number					
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973			
Land Plans		or 152 (3) of the Planning Act 2008			
13-002	106 square metres of private access track and hardstanding (Bradfield Lodge)	A Johns			
		Unit 22			
		Bradfield Lodge			
		Clacton Road			
		Manningtree			
		C011 2NS			
		(in respect of access to Unit 22 Bradfield Lodge)			
		Alan Pole			
		Unit 11			
		Bradfield Lodge			
		Clacton Road			
		Manningtree			
		C011 2NS			
		(in respect of access to Unit 11 Bradfield Lodge)			
		Albert Du Plessis			
		Unit 19			
		Bradfield Lodge			
		Clacton Road			
		Manningtree			
		CO11 2NS			
		(in respect of access to Unit 19 Bradfield Lodge)			
		Arron			
		CK8 and Scaffold Yard			
		Bradfield Lodge			
		Clacton Road			
		MANNGINGTREE			
		Essex			
		CO11 2NX			
		(in respect of access to CK8 and Scaffold Yard Bradfield Lodge)			
		4			

		Five Estuaries Offshore Wind Farm Development Consent Order	
		BOOK OF REFERENCE - PART 2	
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-002 cont'd		Arron IngramUnit 18bBradifield LodgeClacton RoadManningtreeC011 2NS(in respect of access to Unit 18b Bradfield Lodge)Ben RossUnit 5Bradfield LodgeClacton RoadManningtreeC011 2NS(in respect of access to Unit 5 Bradfield Lodge)Blake BirchillUnit 12Bardfield LodgeMANNINGTREEEssexC011 2NS(in respect of access to Unit 12 Bradfield Lodge)Cati TarrantUnit 2Bradfield LodgeClatton RoadManningtreeC011 2NS(in respect of access to Unit 12 Bradfield Lodge)Cati TarrantUnit 2Bradfield LodgeClatton RoadManningtreeC11 2NS(in respect of access to Unit 12 Bradfield Lodge)Cati TarrantUnit 2Bradfield LodgeClatton RoadManningtreeC11 2SN(in respect of access to Unit 2 Bradfield Lodge)	

		Five Estuaries Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
13-002 cont'd		Darren Smith         Units 17 and 18a         Bradfield Lodge         Clacton Road         MANNGINGTREE         CO11 2NS         (in respect of access to Unit 17 Bradfield Lodge)         Dave Sayer         Units 1 and 4         Bradfield Lodge         Clacton Road         MANNGINGTREE         CO11 2NS         (in respect of access to Units 1 and 4 Bradfield Lodge)         East Anglian Farm Rides         Manor Bungalow         Church Road         Little Bentley         COCHESTER         CO7 8SE         (in respect of rights of access)         Jason Childs         Units 3 and 13         Bradfield Lodge         MANNINGTREE         Essex         CO11 2NS         (in respect of rights of access)         Jason Childs         Units 9 and 13         Bradfield Lodge         MANNINGTREE         Essex         CO11 2NS         (in respect of access to Units 9 and 13 Bradfield Lodge)		

	Five Estuaries Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
Counties of Essex and Suffolk			
lumber on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-002 cont'd		Jason Hinsen Feed Store 2	
cont d		Bradfield Lodge	
		Clacton Road	
		Manningtree	
		CO11 2NS	
		(in respect of access to Feed Store 2)	
		Jim Storey	
		Unit 6	
		Bradfield Lodge	
		Clacton Road	
		Manningtree CO11 2NS	
		(in respect of access to Unit 6 Bradfield Lodge)	
		Joe Piggot	
		Unit 20	
		Bradfield Lodge	
		Clacton Road	
		Manningtree	
		CO11 2NS	
		(in respect of access to Unit 20 Bradfield Lodge)	
		Kanye Ault	
		Unit 15	
		Bradfield Lodge	
		MANNINGTREE	
		Essex	
		CO11 2NS	
		(in respect of access to Unit 15 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-002 cont'd		Les Troller         Unit 10         Bradifield Lodge         Clacton Road         Manningtree         CO11 2NS         (in respect of access to Unit 10 Bradfield Lodge)         Mick Cummins         Unit 23         Bradifield Lodge         Clacton Road         MANNGINGTREE         CO11 2NS         (in respect of access to Unit 23 Bradfield Lodge)         Mick Innis         Unit 3         Bradifield Lodge         Clacton Road         MANNGINGTREE         CO11 2NS         (in respect of access to Unit 23 Bradfield Lodge)         Mick Innis         Units 8 and 16         Bradfield Lodge         Claton Road         MANNGINGTREE         CO11 2NS         (in respect of access to Unit 8 Bradfield Lodge)         Peter Barrat         Unit 3         Bradfield Lodge         MANNINGTREE         Essex         CO11 2NS         (in respect of access to Unit 3 Bradfield Lodge)         Heter Barrat         Unit 3         Bradfield Lodge         MANNINGTREE         Essex	

		Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-002 cont'd		Rickman         Unit 21         Bradifield Lodge         Clacton Road         Manningtree         CO11 2NS         (in respect of access to Unit 21 Bradfield Lodge)         S Smith         Unit 7         Bradfield Lodge         Clacton Road         Manningtree         C011 2NS         (in respect of access to Unit 7 Bradfield Lodge)         Starfore         Clacton Road         Manningtree         C011 2NS         (in respect of access to Unit 7 Bradfield Lodge)         Scott Adams         CK7         Bradfield Lodge         MANNINGTREE         Essex         C011 2NS         (in respect of access to CK7 Bradfield Lodge)         Stairform         Feed Store 1         Bradfield Lodge         Clacton Road         Manningtree         Clacton Road         Manningtree         Clatt Road         Manningtree         Clatt Road         Manningtree         Clatt Road         Manningtree         Clatt Poly         Kin respect of access to Feed Store 1 Bradfield Lodge)

	Five Estuaries Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-002 cont'd		Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)	
13-003	299 square metres of private access track and hardstanding (Bradfield Lodge)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree Clacton Road	

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973
Land Plans	·	or 152 (3) of the Planning Act 2008
13-003		Arron
cont'd		CK8 and Scaffold Yard
		Bradfield Lodge
		Clacton Road
		MANNGINGTREE
		Essex
		CO11 2NX
		(in respect of access to CK8 and Scaffold Yard Bradfield Lodge)
		Arron Ingram
		Unit 18b
		Bradfield Lodge
		Clacton Road
		Manningtree
		CO11 2NS
		(in respect of access to Unit 18b Bradfield Lodge)
		Ben Ross
		Unit 5
		Bradfield Lodge
		Clacton Road
		Manningtree
		CO11 2NS
		(in respect of access to Unit 5 Bradfield Lodge)
		Blake Birchill
		Unit 12
		Bradfield Lodge
		MANNINGTREE
		Essex
		CO11 2NS
		(in respect of access to Unit 12 Bradfield Lodge)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-003 cont'd		Carl TarrantUnit 2Bradifield LodgeClacton RoadManningtreeCO11 2SN(in respect of access to Unit 2 Bradfield Lodge)Darren SmithUnits 17 and 18aBradfield LodgeClacton RoadMANNGINGTREECO11 2NS(in respect of access to Unit 14 Bradfield Lodge)Dave SayerUnits 1 and 4Bradfield LodgeClacton RoadMANNGINGTREECO11 2NS(in respect of access to Unit 14 Bradfield Lodge)Dave SayerUnits 1 and 4Bradfield LodgeClacton RoadMANNGINGTREECO11 2NS(in respect of access to Units 1 and 4 Bradfield Lodge)East Anglian Farm RidesManor BungalowChurch RoadLittle BentieyCOLCHESTERCO7 8E(in respect of rights of access)	

		Five Estuaries Offshore Wind Farm Development Consent Order	
		BOOK OF REFERENCE - PART 2	
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-003 cont'd		Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Claton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge) Jim Storey Unit 6 Bradfield Lodge Claton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Claton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) (in respect of access to Unit 6 Bradfield Lodge) (acton Road Manningtree Claton Road Manningtree Claton Road Manningtree Claton Road	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-003 cont'd		Kanye Ault         Unit 15         Bradifield Lodge         MANNINGTREE         Essex         C011 2NS         (in respect of access to Unit 15 Bradfield Lodge)         Les Troller         Unit 10         Bradifield Lodge         Clatcn Road         Manningtree         C011 2NS         (in respect of access to Unit 10 Bradfield Lodge)         Mick Cummins         Unit 23         Bradfield Lodge         Clatcon Road         MANNGINGTREE         C011 2NS         (in respect of access to Unit 10 Bradfield Lodge)         Mick Cummins         Unit 23         Bradfield Lodge         Clatcon Road         MANNGINGTREE         C011 2NS         (in respect of access to Unit 23 Bradfield Lodge)         Mick Innis         Unit 28 and 16         Bradfield Lodge         Clatcon Road         MANNGINGTREE         C011 2NS         (in respect of access to Unit 23 Bradfield Lodge)         Mick Innis         Units 8 and 16         Bradfield Lodge         Clatcon Road         MANNGINGTREE	

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-003 cont'd		Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge) Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge)

	Five Estuarie	es Offshore Wind Farm Development Consent Order			
		BOOK OF REFERENCE - PART 2			
	Counties of Essex and Suffolk				
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
13-003 cont'd		Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)			
13-004	78 square metres of private access track to Bradfield Lodge (south of Clacton Road)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge)			

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on		Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973
and Plans	Description of Land	or 152 (3) of the Planning Act 2008
13-004		Albert Du Plessis
cont'd		Unit 19
		Bradfield Lodge
		Clacton Road
		Manningtree
		CO11 2NS
		(in respect of access to Unit 19 Bradfield Lodge)
		(in respect of access to onit 15 bradnend Louge)
		Arron
		CK8 and Scaffold Yard
		Bradfield Lodge
		Clacton Road
		MANNGINGTREE
		Essex
		CO11 2NX
		(in respect of access to CK8 and Scaffold Yard)
		Arron Ingram
		Unit 18b
		Bradfield Lodge
		Clacton Road
		Manningtree
		CO11 2NS
		(in respect of access to Unit 18b Bradfield Lodge)
		Ben Ross
		Unit 5
		Bradfield Lodge
		Clacton Road
		Manningtree
		CO11 2NS
		(in respect of access to Unit 5 Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-004 cont'd		Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order	
BOOK OF REFERENCE - PART 2			
Counties of Essex and Suffolk			
umber on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973	
and Plans		or 152 (3) of the Planning Act 2008	
13-004		East Anglian Farm Rides	
cont'd		Manor Bungalow	
		Church Road	
		Little Bentley	
		COLCHESTER	
		CO7 8SE	
		(in respect of rights of access)	
		Jason Childs	
		Units 9 and 13	
		Bradfield Lodge	
		MANNINGTREE	
		Essex	
		C011 2NS	
		(in respect of access to Units 9 and 13 Bradfield Lodge)	
		Jason Hinsen	
		Feed Store 2	
		Bradfield Lodge	
		Clacton Road	
		Manningtree	
		C011 2NS	
		(in respect of access to Feed Store 2 Bradfield Lodge)	
		Jim Storey	
		Unit 6	
		Bradfield Lodge	
		Clacton Road	
		Manningtree	
		CO11 2NS	
		(in respect of access to Unit 6 Bradfield Lodge)	

		Five Estuaries Offshore Wind Farm Development Consent Order	
	BOOK OF REFERENCE - PART 2		
Counties of Essex and Suffolk			
lumber on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973	
and Plans		or 152 (3) of the Planning Act 2008	
13-004		Joe Piggot	
cont'd		Unit 20	
		Bradfield Lodge	
		Clacton Road	
		Manningtree	
		C011 2NS	
		(in respect of access to Unit 20 Bradfield Lodge)	
		Kanye Ault	
		Unit 15	
		Bradfield Lodge	
		MANNINGTEE	
		Essex	
		C011 2NS	
		(in respect of access to Unit 15 Bradfield Lodge)	
		Les Troller	
		Unit 10	
		Bradfield Lodge	
		Clacton Road	
		Manningtree	
		C011 2NS	
		(in respect of access to Unit 10 Bradfield Lodge)	
		Mick Cummins	
		Unit 23	
		Bradfield Lodge	
		Clacton Road	
		MANNGINGTREE	
		CO11 2NS	
		(in respect of access to Unit 23 Bradfield Lodge)	

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-004 cont'd		Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 8 and 16 Bradfield Lodge) Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clatton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) (in respect of access to Unit 7 Bradfield Lodge) (in respect of access to Unit 7 Bradfield Lodge)

	Five Estuar	ries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-004 cont'd		Scott Adams         CK7         Bradfield Lodge         MANNINGTREE         Essex         C011 2NS         (in respect of access to CK7 Bradfield Lodge)         Stairform         Feed Store 1         Bradfield Lodge         Clacton Road         Manningtree         C011 2NS         (in respect of access to Feed Store 1 Bradfield Lodge)         Tony Robinson         Unit 14         Bradfield Lodge         Clacton Road         Manningtree         C011 2NS         (in respect of access to Feed Store 1 Bradfield Lodge)
13-005 2	235 square metres of private access track to Bradfield Lodge (off Clacton Road)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on		Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973
Land Plans	Description of Land	or 152 (3) of the Planning Act 2008
13-005		Alan Pole
cont'd		Unit 11
		Bradfield Lodge
		Clacton Road
		Manningtree
		C011 2NS
		(in respect of access to Unit 11 Bradfield Lodge)
		Albert Du Plessis
		Unit 19
		Bradfield Lodge
		Clacton Road
		Manningtree
		CO11 2NS
		(in respect of access to Unit 19 Bradfield Lodge)
		Arron
		CK8 and Scaffold Yard
		Bradfield Lodge
		Clacton Road
		MANNGINGTREE
		Essex
		CO11 2NX
		(in respect of CK8 and Scaffold Yard Bradfield Lodge)
		Arron Ingram
		Unit 18b
		Bradfield Lodge
		Clacton Road
		Manningtree
		CO11 2NS
		(in respect of access to Unit 18b Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-005 cont'd		Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 17 Bradfield Lodge)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2		
Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-005 cont'd		Dave Sayer         Units 1 and 4         Bradfield Lodge         Clacton Road         MANNGINGTREE         CO11 2NS         (in respect of access to Units 1 and 4 Bradfield Lodge)         East Anglian Farm Rides         Manor Bungalow         Church Road         Little Bentley         COLCHESTER         CO7 85E         (in respect of rights of access)         Jason Childs         Units 9 and 13         Bradfield Lodge         MANNINTREE         Essex         C011 2NS         (in respect of access to Units 9 and 13 Bradfield Lodge)         Jason Hinsen         Feed Store 2         Bradfield Lodge         Clacton Road         Manningtree         CO12 1NS         (in respect of access to Feed Store 2 Bradfield Lodge)	

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-005 cont'd		Jim Storey Unit 6 Bradifield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge) John Charles Jiggens HempStalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of rights contained in a transfer dated the 22nd February 2011) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NZ (in respect of rights contained in a transfer dated the 22nd February 2011) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 15 Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-005 cont'd		Les Troller Unit 10 Bradifield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge) Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 8 and 16 Bradfield Lodge) Pter Barrat Units 8 and 16 Bradfield Lodge Clacton Road MANNGISTEE CO11 2NS (in respect of access to Units 8 and 16 Bradfield Lodge) Pter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2	
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-005 cont'd		Rickman         Unit 21         Bradifield Lodge         Clacton Road         Manningtree         C011 2NS         (in respect of access to Unit 21 Bradfield Lodge)         S Smith         Unit 7         Bradifield Lodge         Clacton Road         Manningtree         C011 2NS         Clacton Road         Manningtree         C011 2NS         (in respect of access to Unit 7 Bradfield Lodge)         Scott Adams         CK7         Bradfield Lodge         MANNINGTREE         Essex         C011 2NS         (in respect of access to CK7 Bradfield Lodge)         Stairform         Feed Store 1         Bradfield Lodge         Clatton Road         Manningtree         C011 2NS         (in respect of access to CK7 Bradfield Lodge)         Stairform         Feed Store 1         Bradfield Lodge         Clacton Road         Manningtree         C011 2NS         (in respect of access to Feed Store 1 Bradfield Lodge)	

r	Five Estuaries	Offshore Wind Farm Development Consent Order			
1	BOOK OF REFERENCE - PART 2				
	Counties of Essex and Suffolk				
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
13-005 cont'd		Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)			
13-006	829 square metres of private access track and verges (south of Clacton Road)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge)			

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-006 cont'd		ArronCK8 and Scaffold YardBradifield LodgeClacton RoadMANNGINGTREEEssexC011 2NX(in respect of access to CK8 and Scaffold Yard Bradfield Lodge)Arron IngramUnit 18bBradifield LodgeClacton RoadManningtreeC011 2NS(in respect of access to Unit 18b Bradfield Lodge)Ben RossUnit 5Bradfield LodgeClacton RoadManningtreeC011 2NS(in respect of access to Unit 18b Bradfield Lodge)Ben RossUnit 5Bradfield LodgeClaton RoadManningtreeC011 2NS(in respect of access to Unit 5 Bradfield Lodge)Blake BirchillUnit 12Bradfield LodgeBlake BirchillUnit 12Bradfield LodgeKANNINGTREEEssexC011 2NS(in respect of access to Unit 12 Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-006 cont'd		Carl Tarrant Unit 2 Bradifield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 85E (in respect of faccess)

		Five Estuaries Offshore Wind Farm Development Consent Order	
		BOOK OF REFERENCE - PART 2	
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-006 cont'd		Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge) Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree Clacton Road	

		Five Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-006 cont'd		Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 15 Bradfield Lodge) Kelsy Jane Bamford 2 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 2 Jubilee Cottage) Leah Faye Banks 1 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 1 Jubilee Cottage) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge)

		Five Estuaries Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk			
lumber on and Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
13-006 cont'd		Mick Cummins Unit 23		
		Bradfield Lodge		
		Clacton Road		
		MANNGINGTREE		
		CO11 2NS		
		(in respect of access to Unit 23 Bradfield Lodge)		
		Mick Innis		
		Units 8 and 16		
		Bradfield Lodge		
		Clacton Road		
		MANNGINGTREE		
		CO11 2NS		
		(in respect of access to Units 8 and 16 Bradfield Lodge)		
		Peter Anthony Banks		
		1 Jubilee Cottages		
		Clacton Road		
		Horsley Cross		
		MANNINGTREE		
		C011 2NS		
		(in respect of a right of access to 1 Jubilee cottage)		
		Peter Barrat		
		Unit 3		
		Bradfield Lodge		
		MANNINGTREE		
		Essex		
		CO11 2NS		
		(in respect of access to Unit 3 Bradfield Lodge)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-006 cont'd		Rickman         Unit 21         Bradifield Lodge         Clacton Road         Manningtree         CO11 2NS         (in respect of access to Unit 21 Bradfield Lodge)         Robin Neal Banks         1 Jubilee Cottages         Clacton Road         Horsley Cross         MANNINGTREE         CO11 2NS         (in respect of a right of access to 1 Jubilee Cottage)         S smith         Unit 7         Bradfield Lodge         Clacton Road         Manningtree         CO11 2NS         (in respect of a right of access to 1 Jubilee Cottage)         S smith         Unit 7         Bradfield Lodge         Clacton Road         Manningtree         CO11 2NS         (in respect of access to Unit 7 Bradfield Lodge)         Scott Adams         CK7         Bradfield Lodge         MANNINGTREE         Essex         CO11 2NS         (in respect of access to CK7 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order				
		BOOK OF REFERENCE - PART 2			
		Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
13-006 cont'd		Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Thomas William Wright 2 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 2 Jubilee Cottage) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree Clatton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)			
13-008	20577 square metres of agricultural land (east of Clacton Road B1035)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970)			
13-009	10798 square metres of agricultural land (east of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970)			

	Five Estuaries	Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
13-010	1594 square metres of agricultural land (east of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970)
13-011	350 square metres of agricultural land (east of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970)
13-012	195 square metres of agricultural land and hedgerow (east of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970)
13-014	259 square metres of public highway and verge (east of Clacton Road) (excluding all the interests of the National Highways Limited)	Unknown
13-016	3 square metres of agricultural land, verge and access splay (west of Clacton Road) (excluding all the interests of the National Highways Limited)	Unknown
13-017	25799 square metres of agricultural land (west of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights and restrictive covenants as contained in Deed dated 29 April 1936, Deed dated 6 June 1961, Deed dated 8 July 1970 and Deed dated 31 December 1990)

r	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-017 cont'd		Nigel Graham Dyson Bentley Manor Church Road Little Bentley COLCHESTER Essex CO7 8SE (in respect of right of access to sell goods) Robert Christmas Mulleys Cottage Bentley Road Little Bromley MANNINGTREE CO11 2PL (in respect of shooting rights)	
13-019	3199 square metres of public highway and verges (Clacton Road) (excluding all the interests of the National Highways Limited)	Unknown	
13-022	39078 square metres of agricultural land, private access track and verge (west of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights and restrictive covenants as contained in Deed dated 29 April 1936, Deed dated 6 June 1961, Deed dated 8 July 1970 and Deed dated 31 December 1990) Nigel Graham Dyson Bentley Manor Church Road Little Bentley COLCHESTER Essex CO7 8SE (in respect of right of access to sell goods)	

1	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Northan			
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973	
Land Plans		or 152 (3) of the Planning Act 2008	
13-022		Robert Christmas	
cont'd		Mulleys Cottage	
		Bentley Road	
		Little Bromley	
		MANNINGTREE	
		CO11 2PL	
		(in respect of shooting rights)	
13-023	32744 square metres of agricultural land and brook (Holland Brook) (west of Clacton Road)		
		Mulleys Cottage	
		Bentley Road	
		Little Bromley	
		MANNINGTREE	
		CO11 2PL	
		(in respect of shooting rights)	
14-001	377 square metres of agricultural land (west of Clacton Road)	Affinity Water Limited	
		Tamblin Way	
		HATFIELD	
		Hertfordshire	
		AL10 9EZ	
		(in respect of the rights and restrictive covenants as contained in Deed dated 29 April 1936, Deed dated 6 June 1961, Deed dated 8 July	
		1970 and Deed dated 31 December 1990)	
		Robert Christmas	
		Mulleys Cottage	
		Bentley Road	
		Little Bromley	
		MANNINGTREE	
		CO11 2PL	
		(in respect of shooting rights)	
14-010	24 square metres of private access track (known as Mulberry Lane) and access splay (east	Unknown	
	of Bentley Road)		
14-012	254 square metres of verge (east of Bentley Road)	Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
14-013	701 square metres of agricultural land, public highway and verge (Bentley Road)	Unknown	
14-014	591 square metres of public highway (Bentley Road)	Unknown	
14-015	313 square metres of agricultural land, public highway and verges (Bentley Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
14-016	345 square metres of agricultural land, public highway and verges (Bentley Road)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
14-017	26 square metres of agricultural land and verge (west of Bentley Road)	Unknown	
14-018	125 square metres of agricultural land (west of Bentley Road)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
14-019	10 square metres of agricultural land (west of Bentley Road)	Unknown	
14-020	18 square metres of agricultural land and verge (west of Bentley Road)	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)	

	Five	Estuaries Offshore Wind Farm Development Consent Order
		BOOK OF REFERENCE - PART 2
		Counties of Essex and Suffolk
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
14-022	4709 square metres of agricultural land (west of Bentley Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)
14-023	1438 square metres of agricultural land (west of Bentley Road)	Unknown
14-025	99 square metres of verge (east of Bentley Road)	Unknown
14-026	592 square metres of public highway (Bentley Road)	Unknown
14-027	285 square metres of public highway (Bentley Road)	Unknown
14-028	66 square metres of agricultural land and verge (west of Bentley Road)	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)
14-029	12 square metres of agricultural land and verge (west of Bentley Road)	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)
14-033	3495 square metres of public highway (Bentley Road)	Unknown
14-034	57 square metres of public highway and verge (Bentley Road)	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)
14-039	51 square metres of verge (east of Payne's Lane)	Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
14-040	7 square metres of verge (east of Payne's Lane)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (in respect of right of way) Unknown	
14-041	2 square metres of verge (east of Payne's Lane)	Unknown	
14-042	64 square metres of public highway and verges (Payne's Lane)	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)	
14-043	715 square metres of public highway (Payne's Lane)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Unknown	
14-044	146 square metres of agricultural land (west of Payne's Lane)	Unknown	
14-045	23636 square metres of agricultural land and grass land (west of Payne's Lane)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
14-046	23615 square metres of agricultural land and grass land (west of Payne's Lane)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-002	953 square metres of public highway and verges (Bentley Road)	Unknown	
15-003	10 square metres of verge (east of Bentley Road)	Unknown	
15-005	62 square metres of public highway and verges (Bentley Road)	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)	
15-006	3089 square metres of agricultural land (west of Bentley Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights contained in a Deed dated 22 February 2008)	
15-007	8381 square metres of public highway and verges (Bentley Road) (excluding all the interests of the National Highways Limited)	Unknown	
15-011	646 square metres of agricultural land (north of Bentley Road)	Unknown (in respect of rights contained in a Conveyance dated 26 June 1968)	
15-013	348 square metres of agricultural land and hedgerow (north of Bentley Road)	Unknown (in respect of rights contained in a Conveyance dated 26 June 1968)	
15-014	74 square metres of public highway and verges (Bentley Road) (excluding all the interests of the National Highways Limited)	Unknown	
15-017	2857 square metres of public highway (A120) (excluding all the interests of the National Highways Limited)	Unknown	
16-001	336 square metres of agricultural land (east of Spratts Lane)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-002	42 square metres of public highway and verge (Spratts Lane)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-003	10 square metres of public highway and verge (Spratts Lane)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-004	18 square metres of agricultural land (east of Spratts lane)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-005	1559 square metres of public highway and verges (Spratts Lane)	Unknown	
16-006	15 square metres of public highway and verges (Spratts Lane)	Unknown	
16-007	41 square metres of public highway and verge (Spratts Lane)	Unknown	
16-008	274 square metres of agricultural land and hedgerow (west of Spratts Lane)	Unknown	
16-009	144 square metres of agricultural land (west of Spratts Lane)	Unknown	
16-010	198 square metres of agricultural land (west of Spratts Lane)	Unknown	
16-011	146 square metres of agricultural land (west of Spratts Lane)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-012	30979 square metres of agricultural land and public footpath (FP 17 172) (east of Barlon Road)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-013	33129 square metres of agricultural land (east of Barlon Road) and public footpath (FP 17 172)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-014	85 square metres of verge (east of Barlon Road)	Unknown	
16-015	41 square metres of public highway and agricultural land (Barlon Road)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	

r	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-016	710 square metres of public highway and verges (Barlon Road)	Unknown	
16-020	32985 square metres of agricultural land, private access track and public footpath (FP 16 172) (west of Barlon Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)	
16-021	33719 square metres of agricultural land, private access track and public footpath (FP 16 172) (west of Barlon Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)	
16-022	7974 square metres of agricultural land, private access tack and public footpath (FP 16 172) (west of Barlon Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)	
16-023	1494 square metres of agricultural land, private access track (part of Hall Farm) and public footpath (FP 16 172) (west of Barlon Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)	

<b>I</b>	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-001	17377 square metres of agricultural land and hedgerow (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	
17-002	15087 square metres of agricultural land and hedgerow (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	
17-003	2442 square metres of agricultural land and drain (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-004	9600 square metres of agricultural land and hedgerow (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	
17-005	12584 square metres of agricultural land (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	
17-006	12405 square metres of private access track to Catts Green Farm (off Ardleigh Road), hedgerow and public footpath (FP 15 172)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)	

<b></b>	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-006 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	
17-007	14 square metres of private access track to Catts Green Farm (off Ardleigh Road)	Unknown	
17-008	1623 square metres of public highway, verges (Ardleigh Road) and public footpath (FP 15 172)	Unknown	
17-009	10 square metres of agricultural land and verge (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Unknown (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	
17-010	16 square metres of agricultural land (south of Ardleigh Road)	Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
17-011	51 square metres of agricultural land (south of Ardleigh Road)	Unknown		
17-012	411 square metres of agricultural land (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)		
17-013	61 square metres of agricultural land and verge (south of Ardleigh Road)	Unknown		
17-014	489 square metres of public highway and verges (Ardleigh Road)	Unknown		
17-015	30 square metres of agricultural land and verge (north of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Unknown (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)		

	Five Estuaries Offshore Wind Farm Development Consent Order		
	The L3	BOOK OF REFERENCE - PART 2	
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-016	175 square metres of agricultural land and verge (north of Ardleigh Road)	Unknown	
17-017	107 square metres of agricultural land and verge (north of Ardleigh Road)	Unknown	
17-018	428 square metres of agricultural land and verge (north of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Unknown (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	
17-019	35 square metres of agricultural land and verge (north of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Unknown (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	

	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-020	407 square metres of public highway and verges (Ardleigh Road)	Unknown	
17-021	342 square metres of public highway and verges (Ardleigh Road)	Unknown	
17-022	933 square metres of public highway and verges (Ardleigh Road)	Unknown	
17-023	120 square metres of agricultural land (west of Ardleigh Road)	Unknown	
17-024	296427 square metres of agricultural land and drain (north of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Unknown (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	
17-025	182196 square metres of agricultural land and verge (east of Grange Road)	Unknown (in respect of rights contained in a Deed dated 10 December 1942)	
17-026	2436 square metres of agricultural land and verge (north of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965)	

	Five Estuaries Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-026 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)	
17-027	64 square metres of agricultural land and verge (east of Grange Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)	
17-028	463 square metres of agricultural land (east of Grange Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N SEH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	

Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk         Number on Land Plans       Description of Land       Potential claims under S10 Compulsory Purchase or 152 (3) of the Pla 0 r 152 (3	inning Act 2008
Image: Counties of Essex and Suffolk         Number on Land Plans       Description of Land       Potential claims under S10 Compulsory Purchase or 152 (3) of the Pla or 152 (3) of the Pla (in respect of right of way contained in a Conveyance dated 11 January         17-028       Unknown         17-029       6 square metres of verge (east of Grange Road)       Unknown         17-030       2838 square metres of public highway and verges (Grange Road)       Unknown         17-031       338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)       Moorhouse Farms Limited Begbies Traynor Town Wall House         4       Balkerne Hill COLCHESTER CO3 3AD       COL HESTER	inning Act 2008
Number on Land PlansDescription of LandPotential claims under S10 Compulsory Purchase or 152 (3) of the Plance17-028 cont'dUnknown (in respect of right of way contained in a Conveyance dated 11 January17-0296 square metres of verge (east of Grange Road)Unknown17-0302838 square metres of public highway and verges (Grange Road)Unknown17-031338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)Moorhouse Farms Limited Begbies Traynor Town Wall House 4 Balkerne Hill COLCHESTER CO3 3AD	nning Act 2008
Land PlansDescription of Landor 152 (3) of the Plance17-028 cont'dUnknown (in respect of right of way contained in a Conveyance dated 11 January17-0296 square metres of verge (east of Grange Road)Unknown17-0302838 square metres of public highway and verges (Grange Road)Unknown17-031338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)Moorhouse Farms Limited Begbies Traynor Town Wall House 4 Balkerne Hill COLCHESTER CO3 3AD	nning Act 2008
Land Plans       or 152 (3) of the Plans         17-028       Unknown         cont'd       Unknown         17-029       6 square metres of verge (east of Grange Road)       Unknown         17-030       2838 square metres of public highway and verges (Grange Road)       Unknown         17-031       338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)       Moorhouse Farms Limited         Begbies Traynor       Town Wall House       4 Balkerne Hill         COLCHESTER       CO3 3AD	
cont'd(in respect of right of way contained in a Conveyance dated 11 January17-0296 square metres of verge (east of Grange Road)Unknown17-0302838 square metres of public highway and verges (Grange Road)Unknown17-031338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)Moorhouse Farms Limited Begbies Traynor Town Wall House 4 Balkerne Hill COLCHESTER CO3 3AD	1962)
cont'd(in respect of right of way contained in a Conveyance dated 11 January17-0296 square metres of verge (east of Grange Road)Unknown17-0302838 square metres of public highway and verges (Grange Road)Unknown17-031338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)Moorhouse Farms Limited Begbies Traynor Town Wall House 4 Balkerne Hill COLCHESTER CO3 3AD	1962)
17-029       6 square metres of verge (east of Grange Road)       Unknown         17-030       2838 square metres of public highway and verges (Grange Road)       Unknown         17-031       338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)       Moorhouse Farms Limited Begbies Traynor Town Wall House 4 Balkerne Hill COLCHESTER CO3 3AD	
17-030       2838 square metres of public highway and verges (Grange Road)       Unknown         17-031       338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)       Moorhouse Farms Limited Begbies Traynor Town Wall House         4       Balkerne Hill COLCHESTER CO3 3AD	
17-030       2838 square metres of public highway and verges (Grange Road)       Unknown         17-031       338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)       Moorhouse Farms Limited Begbies Traynor Town Wall House         4       Balkerne Hill COLCHESTER CO3 3AD	
17-031       338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)       Moorhouse Farms Limited Begbies Traynor         17-031       Town Wall House         4 Balkerne Hill       COLCHESTER         COLCHESTER       CO3 3AD	
17-031       338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)       Moorhouse Farms Limited Begbies Traynor         17-031       Town Wall House         4 Balkerne Hill       COLCHESTER         COLCHESTER       CO3 3AD	
Road) Begbies Traynor Town Wall House 4 Balkerne Hill COLCHESTER CO3 3AD	
Road) Begbies Traynor Town Wall House 4 Balkerne Hill COLCHESTER CO3 3AD	
Town Wall House 4 Balkerne Hill COLCHESTER CO3 3AD	
4 Balkerne Hill COLCHESTER CO3 3AD	
COLCHESTER CO3 3AD	
CO3 3AD	
(in respect of rights granted by a beed dated 19 July 2018)	
Philip Douglas Reeve	
Waterhouse Farm	
Waterhouse Lane	
Ardleigh	
COLCHESTER	
Essex	
CO7 7NB	
(in respect of rights contained in a Conveyance dated 4 July 1985)	
Unknown	
(in respect of restrictive covenants and rights reserved by a Transfer da	ed 3 April 1995)
19-001 506 square metres of river (River Ore) and jetty (south of Orford Quay) (excluding all the Orford and Gedgrave Parish Council	
interests of the Crown) The Town Hall	
Market Hill	
Orford	
Woodbridge	
IP12 2NZ	
(in respect of restrictive covenants as contained in a Deed dated 5 Febr	
(in respect of restrictive covenants as contained in a beed dated 5 Febr	1007)

I	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973	
Land Plans	-	or 152 (3) of the Planning Act 2008	
19-001		Secretary of State for Defence	
cont'd		Property Legal Team	
		Defence Infrastructure Organisation Mailpoint 2216	
		Poplar 2	
		Abbey Wood	
		Bristol	
		BS34 8JH	
		(in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987)	
		The National Trust For Places Of Historic Interest Or Natural Beauty	
		Kemble Drive	
		SWINDON	
		Wiltshire	
		SN2 2NA	
		(in respect of rights of access)	
19-002	14323 square metres of private road and jetty (south of River Ore) (excluding all the	East Anglia One North Limited	
	interests of the Crown)	1 Tudor Street	
		LONDON	
		ЕС4У ОАН	
		(in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022)	
		East Anglia Two Limited	
		1 Tudor Street	
		LONDON	
		ЕС4У ОАН	
		(in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022)	
		Norfolk Boreas Limited	
		70 St Mary Axe	
		LONDON	
		EC3A 8BE	
		(in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022)	

	Five Estuaries Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
		Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
19-002 cont'd		Norfolk Vanguard West Limited 70 St Mary Axe LONDON EC3A 8BE (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mallpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987, in a Conveyance dated 26 August 1913 and in a Conveyance dated 20 January 1914) The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Witshire SN2 2NA (in respect of rights of way and access to maintain services as contained in Conveyance dated 31 March 1993) The United Kingdom Atomic Energy Authority Culham Centre for Fusion Energy Culham Science Centre ABINGDON (in respect of rights as stated in Conveyance dated 26 August 1913) Unknown (in respect of rights as stated in Conveyance dated 26 August 1913) Unknown (in respect of rights as stated in Conveyance dated 20 January 1914) Unknown (in respect of right of way as contained in Conveyance 1925)	

1	Five Estuaries Offshore Wind Farm Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
19-003	16 square metres of river (River Ore) (excluding all the interests of the Crown)	Orford and Gedgrave Parish Council The Town Hall Market Hill Orford Woodbridge IP12 2NZ (in respect of restrictive covenants as contained in a Deed dated 5 February 1987) Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987)	
19-004	14 square metres of private access track and scrubland (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993) Unknown (in respect of rights as stated in Conveyance dated 26 August 1913)	
19-005	78 square metres of private access track (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993)	

	Five Estuaries Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2	
	Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
19-005 cont'd		Unknown (in respect of rights as stated in Conveyance dated 31 March 1993)	
		Unknown (in respect of right of way as stated in Conveyance dated 30 June 1925)	
19-006	86 square metres of private access track (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993) Unknown (in respect of right of way as stated in Conveyance dated 30 June 1925)	
19-007	97 square metres of private access track (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993)	
20-001	60 square metres of private access track (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993)	

	Five Estuari	es Offshore Wind Farm Development Consent Order		
		BOOK OF REFERENCE - PART 2		
	Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
20-001 cont'd		Unknown (in respect of rights as stated in Conveyance dated 31 March 1993)		
20-003	69153 square metres of scrubland, drains, tracks and radio masts (east of River Ore) (excluding all the interests of the Crown)	East Anglia One North Limited 1 Tudor Street LONDON EC4Y OAH (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) East Anglia Two Limited 1 Tudor Street LONDON EC4Y OAH (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Norfolk Boreas Limited 70 St Mary Axe LONDON EC3A BBE (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Norfolk Vanguard West Limited 70 St Mary Axe LONDON EC3A BBE (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Norfolk Vanguard West Limited 70 St Mary Axe LONDON EC3A BBE (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Propert 2 Abbey Wood Bristol BS34 B/H (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987, in a Conveyance dated 26 August 1913 and dated 20 January 1914)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
20-003 cont'd		The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA (in respect of rights of way and access to maintain services as contained in Conveyance dated 31 March 1993) The United Kingdom Atomic Energy Authority Culham Centre for Fusion Energy Culham Science Centre ABINGDON OX14 3DB (in respect of rights and restrictive covenants a contained a Deed dated 25 July 1962) Unknown (in respect of rights as stated in Conveyance dated 20 January 1914)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Great Holland Pits Nature Reserve, Little Clacton Road, Great Holland, Frinton-On-Sea	Essex Wildlife Trust Limited Joan Elliot Visitor Centre Abbotts Hall Farm Maldon Road Great Wigborough Colchester CO5 7RZ	
N/A	Walnut House, Lodge Lane, Clacton-on-Sea, CO16 0BS	Veronica Mary Patten Walnut House Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (trading as Vrouslambs)	
N/A	Tanzara, Lodge Lane, Clacton-on-Sea, CO16 0BS	Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Hannams Hall, Tendring, Clacton-On-Sea, CO16 9AR	William Francis Henry Gibbon Hannams Hall Thorpe Road Tendring CLACTON-ON-SEA Essex CO16 9AR	
N/A	Bradley Hall Farm, Whitehall Lane, Thorpe-Le-Soken, Clacton-On-Sea, CO16 0AH	Richard John Lord The Barn Bradley Hall Farm Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH	
N/A	Simons Wood, Thorpe Road, Thorpe-Le-Soken, Clacton-On-Sea	David Milburn Cannell The Old Barn Thorpe Road Tendring CLACTON-ON-SEA Essex CO16 9AR	
N/A	White House, Whitehall Lane, Thorpe-Le-Soken, Clacton-On-Sea, CO16 0AH	Anthony Ronald Winter White House Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH (trading as Winter Family Property Trust)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Daphne Joyce Winter White House Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH (trading as Winter Family Property Trust) Paul Winter White House Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH (trading as Winter Family Property Trust) Rachael Winter White House Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH (LACTON-ON-SEA Essex CO16 0AH (trading as Winter Family Property Trust)	
N/A	Barkers Hall, Whitehall Lane, Thorpe-Le-Soken, Clacton-On-Sea, CO16 0AH	Karl Reuben Lord The Gig House Bradley Hall Farm Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
N/A	Scenefelda Farm, 103 Landermere Road, Thorpe Le Soken, Clacton-on-Sea, CO16 0NG	David Todd Scenefelda Farm 103 Landermere Road Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0NG
N/A	Mayfields Farm, Hungerdown Lane, Ardleigh, CO7 7LZ	Edward James Fairey Mayfield Farm Hungerdown Lane Ardleigh COLCHESTER Essex CO7 7LZ
N/A	Bounds Farm, Hungerdown Lane, Ardleigh, Colchester, CO7 7LZ	Gillian Ann Whittle Bounds Farm Hungerdown Lane Ardleigh COLCHESTER Essex CO7 7LZ Paul John Whittle Bounds Farm Hungerdown Lane Ardleigh COLCHESTER Essex CO7 7LZ
N/A	An electricity sub station, Ardleigh Road, Little Bromley, Manningtree	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Wormsey Wood Farm, Wood Barn Lane, Lawford, Manningtree	Michael Hughes Whirledge & Nott Ltd The Black Barn Lubards Lodge Hullbridge Road RAYLEIGH Essex SS6 9QG (as the executor of the estate for the Late Charles Tabor) Rebecca Mason Holmes & Hills LLP A12 Commercial Hub 86 London Road Marks Tey COLCHESTER CO6 1ED (as the executor of the estate for the Late Charles Tabor) The Executor of the estate for the Late Charles Tabor) The Executor of the Estate of the Late Charles Tabor Sutton Hall Shopland Road ROCHFORD SS4 1LH	
N/A	Jennings Farm, Ardleigh Road, Little Bromley, Manningtree, CO11 2QB	James Richard Sadler Jennings Farm Ardleigh Road Little Bromley MANNINGTREE Essex CO11 2QB	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Badley Hall Farm, Badley Hall Road, Great Bromley, Colchester, CO7 7UU	Kenneth William Robinson Badley Hall Farm Badley Hall Road Great Bromley COLCHESTER Essex CO7 7UU	
N/A	Grange Farm, Grange Road, Lawford	Michael George Harris Richardson Farm Gaston Street East Bergholt COLCHESTER Essex CO7 6SB	
N/A	Little Bromley Hall, Church Road, Little Bromley, Manningtree, CO11 2PP	Natalie Louise Smith Little Bromley Hall Church Road Little Bromley MANNINGTREE Essex CO11 2PP	
N/A	Hiskeys Farm, Spratts Lane, Little Bromley, Manningtree, CO11 2PR	Clare Louise Stow Stow Farm Kennels Spratts Lane Little Bromley MANNINGTREE CO11 2PR	
		Paul Graham Stow Stow Farm Kennels Spratts Lane Little Bromley MANNINGTREE CO11 2PR	

Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
lumber on .and Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
N/A	Crabtrees, Paynes Lane, Little Bromley, Manningtree, CO11 2PJ	Graeme John Knott Crabtrees Paynes Lane Little Bromley MANNINGTREE Essex CO11 2PJ Kate Hodgkiss Crabtrees Paynes Lane Little Bromley MANNINGTREE Essex CO11 2PJ
N/A	Mulleys Farm, Bentley Road, Little Bromley, Manningtree, CO11 2PL	Mary Ann Cooper Mulleys Farm Bentley Road Little Bromley MANNINGTREE Essex CO11 2PL
N/A	Touchwood House, Little Bromley Road, Pelhams Corner, Little Bentley, CO7 8SR	Patricia Maestrani Spring Hall Little Bromley Road Little Bentley COLCHESTER CO7 8SR

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		The Executor of the Estate of the Late Nicholas Paul Maestrani Touchwood House Little Bromley Road Little Bentley COLCHESTER Essex CO7 8SR	
N/A	Touchwood House, Little Bromley Road, Pelhams Corner, Little Bentley, CO7 8SR	Patricia Maestrani Spring Hall Little Bromley Road Little Bentley COLCHESTER CO7 8SR The Executor of the Estate of the Late Nicholas Paul Maestrani Touchwood House Little Bromley Road Little Bentley COLCHESTER Essex CO7 8SR	
N/A	Kellys Farm, Clacton Road, Horsley Cross, Manningtree, CO11 2NZ	Wix Farms Poultry Ltd 3 Manor Road COLCHESTER Essex CO3 3LU	
N/A	1 and 2 Arch Cottages, Clacton Road, Horsley Cross, Manningtree, CO11 2NS	John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Lisa Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	
N/A	Burnt Ash Farm, Colchester Road, Wix, CO11 2PD	Ann Elizabeth Watkinson Burnt Ash Farm Colchester Road Wix MANNINGTREE Essex CO11 2PD Nicholas Martin Watkinson Burnt Ash Farm Colchester Road Wix MANNINGTREE Essex CO11 2PD	
N/A	Castle Byways, Pelhams Corner, Little Bentley, Colchester, CO7 8SS	Jim Clifton Castle Byeways Pellens Corner Little Bentley Essex CO7 8SR	
N/A	Water Works, to the north of A120 and B1035, Horsley Cross, Tendring, Essex, CO11 2PH	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ	

Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
N/A	Hempstall's Farm, Clacton Road, Tendring Heath, Essex, CO11 2PB	John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS Lisa Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS
N/A	Gooses Farm, Parsonage Lane, Tendring, CO16 0DE	Sharon Cheryl Brown Gooses Farm Parsonage Lane Tendring CLACTON-ON-SEA CO16 0DE Simon Bernard Brown Gooses Farm Parsonage Lane Tendring CLACTON-ON-SEA CO16 0DE
N/A	Gooses Farm, Parsonage Lane, Tendring, CO16 0DE	Sharon Cheryl Brown Gooses Farm Parsonage Lane Tendring CLACTON-ON-SEA CO16 0DE

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
N/A cont'd		Simon Bernard Brown Gooses Farm Parsonage Lane Tendring CLACTON-ON-SEA CO16 0DE
N/A	Beckwith Farm, Little Clacton Road, Great Holland, FRINTON-ON-SEA, CO13 0ET	Mark Timothy Borrett Beckwith Farm Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET Trude Borrett Beckwith Farm Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET
N/A	Birch Hoe Farm, Pork Lane, Great Holland, Frinton-On-Sea CO13 0ER	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
N/A	1, Barkers Hall Cottages, Thorpe Road, Beaumont, Clacton-On-Sea, CO16 0AJ	Lesley Elizabeth Mclean Smith 1 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA CO16 0AJ
N/A	2 Barkers Hall Cottages, Thorpe Road, Beaumont, Clacton-on-Sea, CO16 0AJ	Cherie Ann Bomando 2 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA Essex CO16 0AJ Ricky Gerard Bomando 2 Barkers Hall Cottages Thorpe Road Beaumont CLACTON-ON-SEA Essex CO16 0AJ
N/A	1 and 2 Little Clacton Road, Great Holland, Frinton-on-Sea, CO13 0ET	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE
N/A	5 and 6 Little Clacton Road, Great Holland, Frinton-on-Sea, CO13 0ET	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Meadowend, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0ET	Jacqueline Eileen Mills Meadow End Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0ET Stanley Albert Mills Meadow End Little Clacton Road Great Holland	
		FRINTON-ON-SEA Essex CO13 0ET	
N/A	Holly Tree, Little Clacton Road, Great Holland, Frinton-on-Sea, CO13 0ET	Maxine Wiggins Holly Tree Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0ET	
N/A	Paynes Cottage, Paynes Lane, Little Bromley, Manningtree, CO11 2PJ	Elspeth Elinor Knott Paynes Cottage Paynes Lane Little Bromley MANNINGTREE CO11 2PJ Roland Alan Knott	
		Paynes Cottage Paynes Lane Little Bromley MANNINGTREE CO11 2PJ	

Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
N/A	Richmond Cottage, Paynes Lane, Little Bromley, Manningtree, CO11 2PJ	Andrew James Fairley Richmond House Paynes Lane Little Bromley Manningtree Essex CO11 2PJ Helen Davina Fairley Richmond House Paynes Lane Little Bromley Manningtree Essex CO11 2PJ
N/A	Brambles, Barkers Lane, Beaumont, Clacton-On-Sea, CO16 0AL	Barbara Jane Roscoe Brambles Barkers Lane Beaumont CLACTON-ON-SEA CO16 0AL David Graham Roscoe Brambles Barkers Lane Beaumont CLACTON-ON-SEA CO16 0AL
N/A	Molecatchers Cottage, Swan Road, Beaumont, Clacton-On-Sea, CO16 0AN	Hilary Vernon Molecatchers Cottage Swan Road Beaumont CLACTON-ON-SEA CO16 0AN

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	The Nursery, Lodge Lane, Tendring, Clacton-On-Sea, CO16 0BS	Christopher John Opperman 16 Winchester Road Frinton On Sea Colchester CO13 9SB Susan Kathleen Opperman 16 Winchester Road Frinton On Sea Colchester CO13 9SB	
N/A	4 West End Cottages, Swan Road, Beaumont, Clacton-On-Sea, CO16 0AN	Amanda Jayne Greenwood 4 West End Cottages Swan Road Beaumont CLACTON-ON-SEA CO16 0AN	
N/A	Bricklayers Cottage, Swan Road, Beaumont, Clacton-On-Sea, CO16 0AN	Ronald Pierce Traynor Bricklayers Cottage Swan Road Beaumont CLACTON-ON-SEA CO16 0AN Susan Carol Traynor Bricklayers Cottage Swan Road Beaumont CLACTON-ON-SEA CO16 0AN	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Great Holland Mill, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0EU	David William Rampling Great Holland Mill Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU Tracy Jane Rampling Great Holland Mill Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU	
N/A	The Firs, Thorpe Road, Kirby Cross, Frinton-on-Sea, CO13 0NJ	Sheik Kemal Kadar The Firs Thorpe Road Kirby Cross FRINTON-ON-SEA Essex CO13 0NJ	
N/A	4 Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0ET	Ginette Ann Scott 4 Council Houses Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0ET	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Linda Meloney Scott 4 Council Houses Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0ET	
N/A	3 Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0ET	Claire Shandley 3 Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0ET David William Knappett 3 Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0ET	
N/A	Dankeer, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0EU	Martin Terence Clary Dankeer Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU Tracey Jane Clary Dankeer Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	White Cottage, Little Clacton Road, Great Holland, Frinton-on-Sea, CO13 0ET	Cheryl June Crowe White Cottage Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0ET David John Crowe White Cottage Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0ET	
N/A	The Willows, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0ET	Graham Stevens The Willows Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET	
N/A	Yonder House, Little Clacton Road, Great Holland, Frinton-on-Sea, CO13 0ET	Barry Kelvin Reid Yonder House Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0ET	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Elizabeth Kathleen Reid Yonder House Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0ET	
N/A	Barkers Hall, Whitehall Lane, Thorpe-Le-Soken, Clacton-On-Sea, CO16 0AH	Karen Anne Hull Barkers Hall Whitehall Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0AH Richard Walter Hull Barkers Hall Whitehall Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0AH	
N/A	Sunny Days, Little Clacton Road, Great Holland, FRINTON-ON-SEA, CO13 0ET	Joanne Burrell Sunny Days Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET Richard Burrell Sunny Days Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET	

Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
N/A	Sunny Dreams, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0ET	Lesley Susan Clarke Sunny Dreams Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET Leslie Francis Clarke Sunny Dreams Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET
N/A	Frost Farm, Golden Lane, Thorpe-le-Soken, Clacton-on-Sea, CO16 0LE	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE
N/A	Land to the west of Tendring Lane	Sally McAteer Cyprus Cottage Tendring Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0AA

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A		Andrew Charles Furzer Kendor Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET Julie Furzer Kendor Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET	
N/A		Graham Clive Gilbert The Lions Den Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET Jean Gilbert The Lions Den Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET Russell Gilbert The Lions Den Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	1 Valley Farm Cottages, Golden Lane, Thorpe Le Soken, Clacton-On-Sea, CO16 0LE	Gillian Bennington Valley Farm Cottages 1 Golden Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0LE	
N/A	1 Tudor Cottage, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0EU	Jennifer Welsby 1 Tudor Cottage Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU	
N/A	White Lodge, Clacton Road, Great Holland, Frinton-On-Sea, CO13 0JU	Martin Paul Acres White Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU	
N/A	80 Landermere Road, Thorpe-Le-Soken, Clacton-On-Sea, CO16 0NF	Ian Douglas Garner 80 Landermere Road Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0NF Mandy Kathleen Garner 80 Landermere Road Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0NF	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	82 Landermere Road, Thorpele-Soken, CO16 0NF	Anita Higgs 82 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NF James Leonard George Higgs 82 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NF	
N/A	Lodge Farm Bungalow, Clacton Road, Frinton-on-Sea, CO13 OJU	Marian Sarah Reynolds Lodge Farm Bungalow Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU	
N/A	Oakley House, Wolves Hall Lane, Tendring, Clacton-On-Sea, CO16 0DG	Christine Barber Sparkes Farm Cottage Bury Road Thorpe Morieux BURY ST. EDMUNDS IP30 ONT (as the executor of the estate for the late James Hendrie Fairley and the late June Mary Fairley) James Francis Fairley Wolves Hall Wolves Hall Lane Tendring CLACTON-ON-SEA Essex CO16 0DG (as the executor of the estate for the late James Hendrie Fairley and the late June Mary Fairley)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Ring Cottage, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0EU	Patricia Kathleen Hooper 20 Conisboro Avenue Caversham READING RG4 7JB	
N/A	Seawinds, 2 Little Clacton Road, Great Holland, Frinton-on-Sea, CO13 0EU	Barnaby Charles Parker Seawinds Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU Emma Jane Parker Seawinds Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU Louis Parker Seawinds Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU Louis Parker Seawinds Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Valley Barns, Golden Lane, Thorpe Le Soken, Clacton-On-Sea, CO16 0LE	Spencer Leigh Brown Brown Roofing Valley Barns Golden Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0LE (trading as Brown Roofing Contractors Ltd)	
N/A	Land and buildings at New Hall Cottages, Horsley Cross, Manningtree	Bentley Photographic Limited New Hall Barn Clacton Road Horsley Cross Manningtree CO11 2NU	
N/A	101 Landermere Road, Thorpe Le Soken, Clacton-On-Sea, CO16 0NG	Michael George Robert Goosetree 101 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NG	
N/A	Rainbows End, Clacton Road, Great Holland, Frinton-On-Sea, CO13 0JU	Derek Robert Bursey Rainbows End Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU Jean Rosemary Bursey Rainbows End Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Unknown (in respect of mines and minerals)	
N/A	Valley Farm, Golden Lane, Thorpe Le Soken, Clacton-On-Sea, CO16 OLE	Martin Roy Bowers Charity Cottage Thorpe Road Preston St. Mary SUDBURY CO10 9NA Sharon Joy Bowers Charity Cottage Thorpe Road Preston St. Mary SUDBURY CO10 9NA	
N/A	Lodge Farm, Clacton Road, Great Holland, Frinton-on-Sea, CO13 0JU	John George Bellingham Gladwyn House 180 Thorpe Road Kirby Cross FRINTON-ON-SEA Essex CO13 0NH	
N/A	Dairy House Farmhouse and 1 and 2 Dairy House Farm Cottages, Little Clacton Road, Great Holland, Frinton- On-Sea, CO13 0EX	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX	
N/A	The Veldt, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0ET	Richard Edward Smith 19 Cotman Road CLACTON-ON-SEA Essex CO16 8YP	
N/A	Great Holland Lodge, Clacton Road, Great Holland, Frinton-On-Sea, CO13 OJU	John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 OJU Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 OJU	
N/A	Sunny Skies, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0ET	Edward Frederick Pryde Sunny Skies Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Wendy Caroline Pryde Sunny Skies Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET	
N/A	Sunny Fields, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0ET	Bernard Jameson Coverdale Sunny Fields Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET Janis Susan Coverdale Sunny Fields Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET	
N/A	Sunny Meadows, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0ET	David Hinds Sunny Meadows Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET Geraldine Hinds Sunny Meadows Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Sunny Views, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0ET	Linda Ann Chapman Sunny Views Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET Paul Chapman Sunny Views Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET	
N/A	99 Landermere Road, Thorpe Le Soken, CO16 0NG	Pauline Jarrold 99 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NG (as reputed owner)	
N/A	Newhouse Farm, Clacton Road, Horsley Cross, Manningtree, CO11 2NZ	Georgina Margaret Brown Newhouse Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ	
N/A	105 Landermere Road, Thorpe Le Soken, Clacton-On-Sea, CO16 0NG	Mary Patricia Macaulay 105 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NG	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Robert John Macaulay 105 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NG	
N/A	Part of Tendring Road, CO16	Sally McAteer Cyprus Cottage Tendring Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0AA	
N/A	Land at Bounds Farm, Hungerdown Lane, Ardleigh, Colchester, CO7 7LZ	Gillian Ann Whittle Bounds Farm Hungerdown Lane Ardleigh COLCHESTER Essex CO7 7LZ Paul John Whittle Bounds Farm Hungerdown Lane Ardleigh COLCHESTER Essex CO7 7LZ	
N/A	Unit 2, Valley Barns, Golden Lane, Thorpe-Le-Soken, Clacton-On-Sea, CO16 0LE	D A Phillips & Company Limited Bridewell House Bridewell Lane TENTERDEN Kent TN30 6FA	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		David Philip Lewis Bridewell House Bridewell Lane TENTERDEN Kent TN30 6FA	
N/A	The Old Barn, Thorpe Road, Tendring, Clacton-On-Sea, CO16 9AR	David Milburn Cannell The Old Barn Thorpe Road Tendring CLACTON-ON-SEA Essex CO16 9AR	
N/A	2 Frost Farm Cottages, Golden Lane, Thorpe-Le-Soken, Clacton-On-Sea CO16 0LE	Alfie James Davies 2 Frost Farm Cottages Golden Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0LE	
N/A	Damonts Farm, Damants Farm Lane, Thorpe-Le-Soken, Clacton-On-Sea, CO16 ONP	Bobby Swift Damonts Farm Damants Farm Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0NP Penelope Swift Damonts Farm Damants Farm Damants Farm Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0NP	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Land at Abbotts Farm Hall, Mistley, Manningtree	Foxes Property 3 S.A.R.L. 1 Allee Scheffer L-2520 Luxembourg Tungsten Colchester Limited Gateway House 4 Penman Way Grove Park Enderby Leicester LE19 1SY (in respect of a conditional contract to purchase subject to planning)	
N/A	Bradfield Lodge, Clacton Road, Horsley Cross, Manningtree, CO11 2NS	John Harvey Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS	
N/A	Mulberry Lodge, Ardleigh Road, Little Bromley, Manning Tree, CO11 2QB	Graham Peter Lucas Mulberry Lodge Ardleigh Road Little Bromley MANNINGTREE Essex CO11 2QB Sarah Kate Lucas Mulberry Lodge Ardleigh Road Little Bromley MANNINGTREE Essex CO11 2QB	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Rosewood House, Ardleigh Road, Little Bromley, Manningtree, CO11 2QA	Christine Pamela Barrett Rosewood House Ardleigh Road Little Bromley MANNINGTREE CO11 2QA Robert Terence Barrett Rosewood House Ardleigh Road Little Bromley MANNINGTREE CO11 2QA	
N/A	Harvest House, Ardleigh Road, Little Bromley, Manningtree, CO11 2QA	Jens Gerd Thomas Duffy Harvest House Ardleigh Road Little Bromley MANNINGTREE CO11 2QA Mimi Sofia Curran Harvest House Ardleigh Road Little Bromley MANNINGTREE CO11 2QA	
N/A	Jubilee Villa, Ardleigh Road, Little Bromley, Manningtree, CO11 2QA	Lynda Mary Blackburn Jubilee Villa Ardleigh Road Little Bromley MANNINGTREE CO11 2QA	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Land lying to the north of Landermere Road, Thorpe-le-Soken	John Frederick Peirson Blacksmiths Farm Harwich Road Beaumont CLACTON-ON-SEA CO16 0AS	
N/A	Lonsdale Road, Thorpe-Le-Soken, Clacton-On-Sea	Michael William Gay Trinity Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (as reputed owner) Principal Homes Limited Oak House Church Lane Great Holland FRINTON-ON-SEA Essex CO13 0JS Sharon Susan Gay Trinity Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (as reputed owner)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Hamford, Lonsdale Road, Thorpe-le-Soken, CLACTON-ON-SEA, CO16 0LF	Linda Draper Hamford Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (as reputed owner) Principal Homes Limited Oak House Church Lane Great Holland FRINTON-ON-SEA Essex CO13 0JS	
N/A	Jimilda, Lonsdale Road, Thorpe Le Soken, Clacton-On-Sea, CO16 0LF	Orwell Housing Association Limited Crane Hill Lodge 325 London Road IPSWICH Suffolk IP2 0BE Simon Stone Jimilda Lonsdale Road Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0LF (as reputed owner)	
N/A	87 Landermere Road, Thorpe-Le-Soken, Clacton-On-Sea, CO16 0LW	Gwendolyn Rose Batley 87 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LW	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Mark Terry Sangwine 87 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LW	
N/A	Framble Barn, Lonsdale Road, Thorpe-le-Soken, CLACTON-ON-SEA, CO16 0LF	Colin Ernest Anson Framble Barn Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF Jane Elizabeth Anson Framble Barn Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF	
N/A	Millstone Farm, Swan Road, Beaumont, Clacton-On-Sea, CO16 0AN	Stephen Frederick Charles Mills Millstone Farm Swan Road Beaumont CLACTON-ON-SEA CO16 0AN	
N/A	Millstone Farm, Swan Road, Beaumont-cum-Moze, Clacton-on-Sea, CO16 0AN	Stephen Frederick Charles Mills Millstone Farm Swan Road Beaumont CLACTON-ON-SEA CO16 0AN	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Land and buildings Thorpe Park Farm, Thorpe Park Lane, Thorpe Le Soken, Clacton-on-sea	James Roberts Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 OHN Valerie Joan Roberts Thorpe Park Farm Thorpe Park Farm Thorpe Park Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 OHN	
N/A	Green-Acre, Clacton Road, Horsley Cross, Manningtree, CO11 2NS	Deborah Kay Williams Green-Acre Clacton Road Horsley Cross MANNINGTREE CO11 2NS Simon Ronald Williams Green-Acre Clacton Road Horsley Cross MANNINGTREE CO11 2NS	
N/A	Cyprus Cottage, Tendring Road, B1035	Sally McAteer Cyprus Cottage Tendring Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0AA	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Unknown	
N/A	Sunny Days, Little Clacton Road, Great Holland, FRINTON-ON-SEA, CO13 0ET	Joanne Burrell Sunny Days Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET Richard Burrell Sunny Days Little Clacton Road Great Holland FRINTON-ON-SEA CO13 0ET	
N/A	1, Jubilee Cottages, Clacton Road, Horsley Cross, Manningtree, CO11 2NS	Peter Anthony Banks 1 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS Robin Neal Banks 1 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS	
N/A	2, Jubilee Cottages, Clacton Road, Horsley Cross, Manningtree, CO11 2NS	Kelsy Jane Bamford 2 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Thomas William Wright 2 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS	
N/A	Appledene, Clacton Road, Horsley Cross, Manningtree, CO11 2NT	Michaela Jane Partner Appledene Clacton Road Horsley Cross MANNINGTREE CO11 2NT	
N/A	Pinewood, Clacton Road, Horsley Cross, Manningtree, CO11 2NT	David Graham Rider Pinewood Clacton Road Horsley Cross MANNINGTREE CO11 2NT	
N/A	The Lost Willow, 2 Council Houses, Clacton Road, Horsley Cross, Manningtree, CO11 2NT	Georgia Claire Wadling The Lost Willow 2 Clacton Road Horsley Cross MANNINGTREE CO11 2NT Matthew Andrew Wadling The Lost Willow 2 Clacton Road Horsley Cross MANNINGTREE CO11 2NT	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	1 Council Houses, Clacton Road, Horsley Cross, Manningtree, CO11 2NT	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE	
N/A	Burnt Ash Cottage, Colchester Road, Wix, CO11 2PD	Carol Lesley Sinclair Burnt Ash Cottage Colchester Road Wix MANNINGTREE CO11 2PD Stephen William Sinclair Burnt Ash Cottage Colchester Road Wix MANNINGTREE CO11 2PD	
N/A	Oakley House, Lodge Lane, Tendring, Clacton-On-Sea, CO16 0BS	David Anthony White Oakley House Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS Helen White Oakley House Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	The Rondavaal, Lodge Lane, Tendring, Clacton-On-Sea, CO16 0BS	Jacqueline Innes The Rondavaal Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS Stewart Peter Innes The Rondavaal Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS	
N/A	Hawthorn Cottage, Lodge Lane, Tendring, Clacton-On-Sea, CO16 OBS	Rachel Dawn Edwards         Hawthorn Cottage         Lodge Lane         Tendring         CLACTON-ON-SEA         CO16 0BS         Trevor Michael Edwards         Hawthorn Cottage         Lodge Lane         Tendring         CLACTON-ON-SEA         C016 0BS         Trevor Michael Edwards         Hawthorn Cottage         Lodge Lane         Tendring         CLACTON-ON-SEA         C016 0BS	
N/A	Millstone Farm, Swan Road, Beaumont-cum-Moze, Clacton-on-Sea, CO16 0AN	Stephen Frederick Charles Mills Millstone Farm Swan Road Beaumont CLACTON-ON-SEA CO16 0AN	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Wesley Cottage, Little Clacton Road, Great Holland, Frinton-On-Sea, CO13 0EU	Carol Freda White Wesley Cottage Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU Jemma White Wesley Cottage Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EU	
N/A	Gunfleet, Lonsdale Road, Thorpe-Le-Soken, Clacton-On-Sea, CO16 0LF	Brendan Malachy Gormley Gunfleet Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF Margaret Shona Gormley Gunfleet Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF	
N/A	Park Farm, Chase Road, Great Bromley	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	The Lodge, Birch Hoe Farm, Pork Lane, Great Holland, Frinton-On-Sea, CO13 0ER	Jayne Louise Hutley The Lodge Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER Simon James Hutley The Lodge Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-001	52 square metres of verges, copse and access splay (east of Frinton Road, B1032)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Brian Leonard Cross 2 Sluice Cottages Manor Way CLACTON-ON-SEA CO15 5TZ (in respect of rights of access) Gunfleet Boating Club Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access) Irene Cross 2 Sluice Cottages Manor Way CLACTON-ON-SEA CO15 5TZ (in respect of rights of access) Maria Ann Martin 1 Sluice Cottages Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access) Maria Ann Martin 1 Sluice Cottages Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-001 cont'd		Mountview House 151 High Street LONDON N14 6EW (in respect of rights of access) Orsted Energy Solutions (UK) Limited 5 Howick Place LONDON SW1P 1WG (in respect of electricity apparatus) Port Of London Authority London River House Royal Pier Road GRAVESEND Kent DA12 2BG (in respect of rights of access) Stephen Philip Martin 1 Sluice Cottages Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access) TC Gunfleet Sands OFTO Limited 3 More London Riverside LONDON SE1 2AQ (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
01-001 cont'd		Unknown		
01-002		Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of rights granted by a Conveyance dated 21 October 1930, in Conveyance dated 3 August 1932, reserved in a Conveyance dated 9 August 1977 and in respect of right of way and easement) Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of sewerage apparatus) Brian Leonard Cross 2 Sluice Cottages Manor Way CLACTON-ON-SEA CO15 5TZ (in respect of rights of access) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)		

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-002 cont'd		Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 SAH (in respect of the legal easements granted contained in a Deed of Easement dated 11 October 2019, in a Deed of Grant dated 26 February 1986, in a Deed of Grant dated 4 April 1989 and in a Deed of Grant dated 16 January 1998) Gunfleet Boating Club Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access) Irene Cross 2 Sluice Cottages Manor Way CLACTON-ON-SEA CO15 5TZ (in respect of rights of access) Maria Ann Martin 1 Sluice Cottages Manor Way CLACTON-ON-SEA Essex CO15 5TZ (in respect of rights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-002 cont'd		Mountview Estates PLC Mountview House 151 High Street LONDON N14 6FW (In respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (In respect of telecommunication apparatus) Orsted Energy Solutions (UK) Limited S Howick Place LONDON SW1P 1WG (In respect of electricity apparatus) Port Of London Authority London River House Royal Pier Road GRAVESEND Kent DA12 2BG (In respect of rights of access) Stephen Philip Martin 1 Sluice Cottages Manor Way CLACTON-ON-SEA Essex CO15 STZ (In respect of rights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-002 cont'd		TC Gunfleet Sands OFTO Limited 3 More London Riverside LONDON SE1 2AQ (in respect of electricity apparatus) Unknown (in respect of covenants contained in Conveyance dated 16 May 1902) Unknown (in respect of covenants contained in a Conveyance dated 7 October 1930) Unknown (in respect of covenants contained in a Conveyance dated 23 April 1938)	
01-003	10490 square metres of access way off Frinton Road leading to public road (Manor Way) grassland, verges, access tracks and buildings (south of The Esplanade, Holland-on-Sea) and public footpath (FP 29 167)	Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of sewerage apparatus) Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of rights granted by a Conveyance dated 21 October 1930, in Conveyance dated 3 August 1932, reserved in a Conveyance dated 9 August 1977 and in respect of right of way and easement)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-003 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of the legal easements granted contained in Deed of Easement dated 11 October 2019, in a Deed of Grant dated 26 February 1986, in a Deed of Grant dated 4 April 1989 and in a Deed of Grant dated 16 January 1998) Port Of London Authority London River House Royal Pier Road GRAVESEND Kent DA12 2BG (in respect of rights of access) Unknown (in respect of covenants contained in Conveyance dated 16 May 1902) Unknown (in respect of covenants contained in a Conveyance dated 7 October 1930) Unknown (in respect of covenants contained in a Conveyance dated 23 April 1938)	

#### $\bigvee \Xi$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-004	63 square metres of hardstanding (south of Holland Brook) and public footpath (FP 29 167)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
01-005	93 square metres of hardstanding (south of Holland Brook) and public footpath (FP 29 167)	Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of rights as contained in a transfer dated 1 September 1989) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of rights granted by a Deed of Grant dated 12 April 1938)	

#### $\bigvee \Xi$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-005 cont'd 01-006	3218 square metres of foreshore, sea wall and access track (east of Manor Way, Holland-on Sea) and public footpath (FP 29 167)	The Frinton-On-Sea Golf Trust Limited 1 The Esplanade FRINTON-ON-SEA Essex CO13 9EP (in respect of rights granted by a Deed of Grant dated 12 December 1938) Unknown (in respect of restrictive covenants contained in a Conveyance dated 6 November 1900) Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
01-007	21703 square metres of foreshore, sea wall, rock armour, grassland and sloping masonry (west of Second Avenue, Frinton-on-Sea)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Unknown (in respect of an Admitted Claim for C W Hayne dated 28 September 1933 and an Admitted claim for the executors of G F Beaumont dated 1961)	
01-008	29586 square metres of foreshore, grassland, access track, sea wall, rock armour, sloping masonry and drains (west of Second Avenue, Frinton-on-Sea) public bridleway (BR 2 164) and public footpaths (FP 3 164, FP 41 164 and FP 29 167)	Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-009	9511 square metres of agricultural land (east of Manor Way, Holland-on-Sea)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights granted by a Deed of Grant dated 12 April 1938 and Deed of Grant dated 12 December 1938) Great Holland Hall Limited 27 Old Gloucester Street London WC1N 3AX (in respect of rights reserved by a Conveyance dated 30 March 1990) Unknown (in respect of restrictive covenants as contained in Conveyance dated 6 November 1900) Unknown (in respect of restrictive covenants as contained in Conveyance dated 14 November 1929)	
01-010	183349 square metres of golf course, ponds, public footbridges, drains, watercourse (Kirby Brook), public bridleway (BR 2 164) and public footpath (FP 3 164) (south of Short Lane, Frinton-on-Sea)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights granted by the Deed of Grant dated 12 December 1938 and Conveyance dated 16 May 1951) Unknown (in respect of covenants contained in a Conveyance dated 24 April 1920)	
01-011	2423 square metres of watercourse (Kirby Brook)	Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-012	82161 square metres of agricultural land, footbridge and drain (west of Long Lane, Frinton- on-Sea)	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of rights granted by a Conveyance dated 30 March 1990) Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929)	
02-001	16315 square metres of beach, rock armour, sloping masonry, groynes and foreshore at Frinton-On-Sea (south of Frinton Golf Course)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	
02-002	937 square metres of access track (west of Second Avenue, Frinton-on-Sea) and public bridleway (BR 2 164)	Unknown	
02-003	1231 square metres of access track, verges and private car park (west of Second Avenue, Frinton-on-Sea)	Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of sewerage apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by a Transfer dated 5 December 2000)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-003 cont'd		Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights granted by the Deed of Grant dated 12 December 1938 and Conveyance dated 16 May 1951) Unknown (in respect of covenants contained in a Conveyance dated 16 March 1905) Unknown (in respect of restrictive covenants contained in a Conveyance dated 16 March 1905) Unknown (in respect of restrictive covenants contained in a Conveyance dated 16 March 1905) Unknown (in respect of covenants contained in a Conveyance dated 24 April 1920)	
02-004	55938 square metres of agricultural land and drains (west of Long Lane, Frinton-on-Sea)	Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of rights granted by a Conveyance dated 30 March 1990) Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929)	
02-005	122 square metres of footbridge and drain (south of Short Lane, Frinton-on-Sea)	Unknown	
02-006	561 square metres of access track, agricultural land and hedgerow (west of Long Lane, Frinton-on-Sea)	Andrea Woods 9 Beaumont Close WALTON ON THE NAZE CO14 8TX (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-006 cont'd		Andrew Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 OUF (in respect of a right of way) Barry Cullum S6 Crown Road CLACTON-ON-SEA CO15 JAT (in respect of a right of way) Courtney Byrne S3 Cornflower Road Jaywick CLACTON-ON-SEA CO15 ZSA (in respect of a right of way) Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH CO12 SBY (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FINITON-ON-SEA	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-006 cont'd		Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE NAZE CO14 8RN (in respect of a right of way) Louisa Thatcher 104 Salisbury Road Holland-on-Sea CLACTON-ON-SEA CO15 SLT (in respect of a right of way) Mazy King 2 Manor Road Great Holland FRINTON-ON-SEA CO13 UT (in respect of a right of way) Michelle Miller 22 Homerton Close CLACTON-ON-SEA CO15 4UJ (in respect of a right of way) Pat Watson 25 Hillcrest CLACTON-ON-SEA CO15 4HU (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-006 cont'd		Tanya Wheeler         2st. Andrews Close         Weeley         CLACTON-ON-SEA         COI6 9FX         (in respect of a right of way)         Tendring District Council         Town Hall         Station Road         CLACTON-ON-SEA         EXEX         CO15 1SE         (in respect of rights granted by a Conveyance dated 30 March 1990)         Unknown         (in respect of rights granted by a Conveyance dated 14 November 1929)         Victoria Oxland         8 Colthorpe Road         CLACTON-ON-SEA         CO15 4PU         (in respect of a right of way)         Wendy Robinson         4 Polley Close         Kirby Closs         FINITON -ON-SEA         CO13 QUF         (in respect of a right of way)         Yonne Cullum         S6 Crown Road         CLACTON-ON-SEA         CO13 QUF         (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-007	5537 square metres of private roads (Short Lane and Long Lane), and public footpaths (FP 1 164 and FP 3 164) and public bridleway (BR 2 164) (south of Church Lane, Great Holland)	Andrea Woods 9 Beaumont Close WALTON ON THE NAZE (In respect of a right of way) Andrew Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 OUF (in respect of a right of way) Anglian Water Services Limited 1 Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 6XU (in respect of severage apparatus) Barry Cullum S6 Crown Road CLACTON-ON-SEA CO15 IAT (in respect of a right of way) Courtney Byrne S3 Cornflower Road Jaywick CLACTON-ON-SEA CO15 25A (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-007 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH CO12 5BY (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FINITON-ON-SEA CO13 OIT (in respect of a right of way) Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE NAZE CO14 8RN (in respect of a right of way) Louisa Thatcher 104 Salisbury Road Holland-on-Sea CLATCIN-ON-SEA CO15 SLT (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-007 cont'd		Mazy King 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way) Michelle Miller 22 Homerton Close CLACTON-ON-SEA CO15 4UJ (in respect of right of way) Pat Watson 25 Hillcrest CLACTON-ON-SEA CO15 4HU (in respect of a right of way) Tanya Wheeler 2 St. Andrews Close Weeley CLACTON-ON-SEA CO16 9FX (in respect of a right of way) Victoria Oxland 8 Colthorpe Road CLACTON-ON-SEA CO15 4PU (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-007 cont'd		Wendy Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way) Yvonne Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)	
02-008	8503 square metres of agricultural land, drains and access track (west of Long Lane, Frinton on-Sea)	Andrea Woods 9 Beaumont Close WALTON ON THE NAZE CO14 8TX (in respect of a right of way) Andrew Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way) Barry Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-008 cont'd		Courtney Byrne S3 Cornflower Road Jaywick CLACTON-ON-SEA CO15 25A (in respect of a right of way) Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH CO12 5BY (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FINITON-ON-SEA CO13 0JT (in respect of a right of way) Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE NAZE CO14 8RN (in respect of a right of way) Louisa Thatcher 104 Salisbury Road Holland-on-Sea CLACTON-ON-SEA	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-008 cont'd		Mazy King 2 Manor Road Great Holland FRINTON-ON-SEA CO13 OT (in respect of a right of way) Michelle Miller 22 Homerton Close CLACTON-ON-SEA CO15 AUJ (in respect of a right of way) Pat Watson 25 Hillerest CLACTON-ON-SEA CO15 4HU (in respect of a right of way) Tanya Wheeler 2 St. Andrews Close Weeley CLACTON-ON-SEA CO16 9FX (in respect of a right of way) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 JSE (in respect of rights granted by a Conveyance dated 14 November 1929)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-008 cont'd		Victoria Oxland 8 Colthorpe Road CLACTON-ON-SEA CO15 4PU (in respect of a right of way) Wendy Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way) Yvonne Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)	
02-009	56235 square metres of agricultural land, access track, hedgerow and drain (south of Church Lane, Great Holland)	Andrea Woods 9 Beaumont Close WALTON ON THE NAZE CO14 8TX (in respect of a right of way) Andrew Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 OUF (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-009 cont'd		Barry Cullum S6 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way) Courtney Byrne S3 Cornflower Road Jaywick CLACTON-ON-SEA CO15 2SA (in respect of a right of way) Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH CO12 5BY (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FRIIJON-ON-SEA CO13 00 (in respect of a right of way) Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE INAZE CO14 8RN (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-009 cont'd		Louisa Thatcher 104 Salisbury Road Holland-on-Sea CLACTON-ON-SEA CO15 SLT (in respect of a right of way) Mazy King 2 Manor Road Great Holland FRINTON-ON-SEA CO13 OJT (in respect of a right of way) Michelle Miller 22 Homerton Close CLACTON-ON-SEA CO15 4UJ (in respect of right of way) Pat Watson 25 Hillcrest CLACTON-ON-SEA CO15 4HJ (in respect of a right of way) Tanya Wheeler 2 St. Andrews Close Weeley CLACTON-ON-SEA CO15 9FX (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-009 cont'd		Victoria Oxland 8 Colthorpe Road CLACTON-ON-SEA CO15 4PU (in respect of a right of way) Wendy Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 0UF (in respect of a right of way) Yvonne Cullum 56 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way)	
	66097 square metres of agricultural land, hedgerow and access track at Manor Farm (east of Clacton Road, B1032)	Andrea Woods 9 Beaumont Close WALTON ON THE NAZE CO14 8TX (in respect of a right of way) Andrew Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA CO13 OUF (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-010 cont'd		Barry Cullum S6 Crown Road CLACTON-ON-SEA CO15 1AT (in respect of a right of way) Courtney Byrne S3 Cornflower Road Jaywick CLACTON-ON-SEA CO15 25A (in respect of a right of way) Julie Watson 3 Skighaugh Cottages Clacton Road Stones Green HARWICH CO12 5BY (in respect of a right of way) Katy O'Donnell 2 Manor Road Great Holland FRIENTON-ON-SEA CO13 ONT (in respect of a right of way) Lindsey-Cher Johnson 4 Rochford Way WALTON ON THE INAZE CO14 8RN (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-010 cont'd		Louisa Thatcher 104 Salisbury Road Holland-on-Sea CLACTON-ON-SEA CO15 5LT (in respect of a right of way) Mazy King 2 Manor Road Great Holland FRINTON-ON-SEA CO13 0JT (in respect of a right of way) Michelle Miller 22 Homerton Close CLACTON-ON-SEA CO15 4UJ (in respect of a right of way) Pat Watson 25 Hillcrest CLACTON-ON-SEA CO15 4UJ (in respect of a right of way) Pat Watson 25 Hillcrest CLACTON-ON-SEA CO15 4UJ (in respect of a right of way) Tanya Wheeler 2 St. Andrews Close Weeley CLACTON-ON-SEA CO16 9FX (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-010 cont'd		Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex C015 15E (in respect of rights granted by a Conveyance dated 30 March 1990) Unknown (in respect of restrictive covenants contained in a Conveyance dated 14 November 1929) Victoria Oxland 8 Colthorpe Road CLACTON-ON-SEA C015 APU (in respect of a right of way) Wendy Robinson 4 Polley Close Kirby Cross FRINTON-ON-SEA C013 0UF (in respect of a right of way) Vonne Cullum 56 Crown Road CLACTON-ON-SEA C015 1AT (in respect of a right of way)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-002	43609 square metres of agricultural land, copse and hedgerow (east of Clacton Road, B1032)	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
03-003	14204 square metres of agricultural land and hedgerow (east of Clacton Road, B1032)	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-003 cont'd		Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
03-004	1568 square metres of public highway, verges and field accessway (Clacton Road, B1032)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
03-004A	297 square metres of public highway, verges and field accessway (Clacton Road, B1032)	Adam Charles Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 ONG (in respect of rights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-004A cont'd		Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9FZ (in respect of water apparatus) Joanna Marie Brown Woodthorpe House Main Road Great Holland FRINTON-ON-SEA Essex CO13 ONG (in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
03-005	92 square metres of agricultural land (east of Clacton Road, B1032)	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-005 cont'd		Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
03-006	16 square metres of agricultural land and hedgerow (east of Clacton Road, B1032)	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019) Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights contained in a Transfer dated 4 February 2019)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
	1340 square metres of public highway, verges, hedgerow and access splay (Clacton Road, B1032)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
	1640 square metres of public highway, verges, hedgerow and access splay (Clacton Road, B1032)	Affinity Water Limited Tamblin Way HATFIELD AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
03-008	23 square metres of agricultural land and hedgerow (west of Clacton Road, B1032)	John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-008 cont'd		Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)	
03-009	740 square metres of public highway and verges (Clacton Road, B1032)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
03-011	12346 square metres of agricultural land and hedgerow (west of Clacton Road, B1032)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-011 cont'd		Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)	
03-012	10870 square metres of agricultural land and hedgerows (west of Clacton Road, B1032)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951) Lesley Grayson Glover Great Holland Erat Holland FRINTON-ON-SEA Essex Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951)	

#### $\bigvee \Xi$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-013	58 square metres of agricultural land and hedgerow (west of Clacton Road, B1032)	John William Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU (in respect of rights reserved as contained in a Deed dated 27 March 1951) Lesley Grayson Glover Great Holland Lodge Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0JU	
03-014	56162 square metres of agricultural land hedgerow (Little Clacton Road, Great Holland)	(in respect of rights reserved as contained in a Deed dated 27 March 1951) Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights as contained in a Transfer dated 4 February 2019 and in a Conveyance dated 22 December 1982) Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respects of rights contained in a Transfer dated 4 February 2019)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-014 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
03-015	3759 square metres of agricultural land, access track, hardstanding, hedgerow, access splay and public footpath (FP 5 164) (Dairy House Farm, Little Clacton Road, Great Holland)	Alison Margaret Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respect of rights as contained in a Transfer dated 4 February 2019 and in a Conveyance dated 22 December 1982) Douglas Kenneth Brown Dairy House Farm Little Clacton Road Great Holland FRINTON-ON-SEA Essex CO13 0EX (in respects of rights contained in a Transfer dated 4 February 2019)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-015 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
03-016	940 square metres of access track and public footpath (FP 7 164 and FP 10 164) (north of Little Clacton Road, Great Holland)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-017	27 square metres of access track (north of Little Clacton Road, Great Holland) and public footpath (FP 11 164)	Brenda Jane Gibson-Wynes Flat 11 Linkswood Compton Place Road EASTBOURNE BN21 1EE (in respect of covenants by a Transfer dated 5 November 1982) Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access) Unknown (in respect of rights granted by a Deed 23 August 1968)
03-018	249 square metres of access track (north of Little Clacton Road, Great Holland) and public footpath (FP 10 164)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-001	23 square metres of hedgerow and garden (Shorelmist Cottage)	Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of street apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)	
04-002	1004 square metres of public highway, verges, access splay (Little Clacton Road) and public footpath (FP 6 164)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of gas apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-002 cont'd		Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of street furniture) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
04-003	96 square metres of agricultural land (east of Mill Lane, Little Clacton Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
04-004	29040 square metres of agricultural land (north of Little Clacton Road, Great Holland)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-004 cont'd		Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of street apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
04-006	1263 square metres of agricultural land, hedgerow (east of Great Holland Mill, Little Clacton Road) and public footpaths (FP 38 164 and FP 11 164)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
04-007	30147 square metres of agricultural land, hedgerow (east of Great Holland Mill, Little Clacton Road) and public footpaths (FP 38 164 and FP 11 164)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
04-008	1479 square metres of agricultural land (north of Little Clacton Road, Great Holland)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-010	187 square metres of agricultural land, hedgerow (east of Great Holland Mill, Little Clacton Road) and public footpath (FP 38 164)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
04-011	19 square metres of agricultural land (east of Great Holland Mill, Little Clacton Road) and public footpath (FP 38 164)	Unknown
04-015	3 square metres of agricultural land (east of Great Holland Mill, Little Clacton Road)	Unknown
04-016	185 square metres of access track (north of Little Clacton Road, Great Holland) and public footpath (FP 10 164)	Brenda Jane Gibson-Wynes Flat 11 Linkswood Compton Place Road EASTBOURNE BN21 1EE (in respect of covenants by a Transfer dated 5 November 1982) Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access) Unknown (in respect of rights granted by a Deed 23 August 1968)
04-017	1893 square metres of agricultural land, access track and public footpaths (FP 10 164, FP 38 164 and FP 11 164) (east of Great Holland Mill, Little Clacton Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-017 cont'd		Reedlands Farm Angling Club 67 Alton Park Road CLACTON-ON-SEA CO15 1EA (in respect of rights of access)	
04-018	17 square metres of access track (west of Pork Lane, Great Holland) and public footpath (FF 10 164)	Unknown	
04-019	552 square metres of agricultural land, copse, hedgerow (west of Pork Lane, Great Holland) and public footpath (FP 10 164)	Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of statutory rights as contained in a Deed dated 19 December 1966) Unknown (in respect of right to enter to maintain a water main and in respect of restrictive covenants as stated in a Deed dated 16 July 1968)	
04-020	38061 square metres of agricultural land, copse and hedgerow (west of Pork Lane, Great Holland)	Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of statutory rights as contained in a Deed dated 19 December 1966)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-020 cont'd 05-001	8773 square metres of agricultural land and hedgerow (west of Pork Lane, Great Holland)	Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of gas apparatus) Unknown (in respect of right to enter to maintain a water main and in respect of restrictive covenants as stated in a Deed dated 16 July 1968)	
		Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Unknown	
05-002	43670 square metres of agricultural land, woodland, brook and access track at Birch Hoe Farm (west of Pork Lane, Great Holland)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of rights as stated in Conveyance dated 31 March 1982)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-003	129 square metres of agricultural land and hedgerow (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Unknown	
05-004	460 square metres of access track (west of Pork Lane, Great Holland)	John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of right of access as stated in a Conveyance dated 16 July 1968) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of right of access as stated in a Conveyance dated 16 July 1968) Unknown (in respect of rights reserved as contained in a Conveyance dated 16th July 1968)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-005	290 square metres of access track and access splay (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of right of access as stated in a Conveyance dated 16 July 1968) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of right of access as stated in a Conveyance dated 16 July 1968)	
05-006	3 square metres of agricultural land and access splay (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-006 cont'd		John Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of rights of access) Lorna Marion Hutley Birch Hoe Farm Pork Lane Great Holland FRINTON-ON-SEA Essex CO13 0ER (in respect of rights of access) Unknown	
	5689 square metres of agricultural land, access track, brook and hedgerow (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-007 cont'd		Unknown	
05-008	5609 square metres of agricultural land, brook and hedgerow (west of Pork Lane, Great Holland)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of rights granted by a Conveyance dated 29 September 1954) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998) Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)	
05-009	982 square metres of agricultural land, brook and hedgerow (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-010	20 square metres of agricultural land, brook and hedgerow (west of Pork Lane, Great Holland)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of an easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	
05-011	25 square metres of agricultural land and brook (west of Pork Lane, Great Holland)	Unknown (in respect of rights granted by a Conveyance dated 29 September 1954) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998) Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)	
05-012	889 square metres of agricultural land and brook (south of Thorpe Park Lane, Thorpe-le- Soken)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of rights granted by a Conveyance dated 29 September 1954)	

### $\bigvee \Xi$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-012 cont'd		Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998) Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)	
	30555 square metres of agricultural land and woodland (south of Thorpe Park Lane, Thorpe le-Soken)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of rights granted by a Conveyance dated 29 September 1954) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998) Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)	
	3558 square metres of railway (Thorpe-le-Soken and Kirby Cross) and scrubland (south of Thorpe Park Lane)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown	
05-015	1021 square metres of agricultural land (south of Thorpe Park Lane, Thorpe-le-Soken)	Unknown (in respect of rights granted by a Conveyance dated 29 September 1954) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-015 cont'd		Unknown (in respect of rights contained in a Conveyance dated 29 April 1953)	
05-016	103 square metres of railway (Thorpe-le-Soken and Kirby Cross) access track and scrubland (south of Thorpe Park Lane)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown	
05-017	5373 square metres of access track (south of Thorpe Park Lane)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
05-018	408 square metres of private road (Thorpe Park Lane)	Andrzej Tomasz Wiecek 1 Thorpe Park Cottages Thorpe Park Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0HN (in respect of right of access) Annis Roberts Thorpe Park House Thorpe Park Lane Thorpe-le-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-018 cont'd		Benjamin Mark Worrallo 3 Thorpe Park Cottages Thorpe Park Lane Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access) Georgios Tsaousellis 4 Thorpe Park Cottages Thorpe Park Lane Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access) John Whiten 2 Thorpe Park Cottages Thorpe Park Cottages Thorpe-Ie-Soken CLACTON-ON-SEA CO16 0HN (in respect of rights of access) Justyna Magdalena Wiecek 1 Thorpe Park Cottages Thorpe Park Co	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-018 cont'd		Rhiannon Wheeler         4 Thorpe Park Cottages         Thorpe Park Lane         Thorpe-le-Soken         CLACTON-ON-SEA         CO16 0HN         (in respect of rights of access)         Sam Worrallo         3 Thorpe Park Cottages         Thorpe Park Lane         Thorpe Park Cottages         Thorpe Park Lane         Thorpe Park Lane         Thorpe Park Lane         Thorpe-le-Soken         CLACTON-ON-SEA         CO16 0HN         (in respect of rights of access)         Shirley Whiten         2 Thorpe Park Cottages         Thorpe Park Lane         Thorpe Park Cottages         Thorpe Park Cottages         Thorpe Park Cottages         Thorpe Park Lane         Thorpe Park Cottages         Thorpe Park Cottages         Thorpe Park Cottages         Thorpe Park Cottages         Thorpe Park Lane         Thorpe-le-Soken         CLACTON-ON-SEA         CO16 0HN         (in respect of rights of access)         Unknown	
		(in respect of rights reserved by a Conveyance dated 14 March 1998)	
05-019	56801 square metres of agricultural land (south of Thorpe Park Lane)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-019 cont'd		Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
05-020	14516 square metres of agricultural land (to west of Pork Lane)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
05-022	61 square metres of agricultural land (west of Pork Lane, Great Holland)	Unknown	
05-023	3249 square metres of agricultural land and hedgerow (south of Thorpe Road, B1035)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
05-024	38017 square metres of agricultural land, woodland (Parklane Grove) and hedgerow (south of Thorpe Road, B1035)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-024 cont'd		Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
05-025	232 square metres of agricultural land and hedgerow (south of Thorpe Road, B1035)	Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
05-026	666 square metres of agricultural land (Grove Fruit Farm) (west of Pork Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
06-001	3067 square metres of public highway, verges and lay-by (Thorpe Road, B1033)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	

### $\bigvee \Xi$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-005	23298 square metres of agricultural land and hedgerow (south of Thorpe Road, B1035)	Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
06-006	249 square metres of agricultural land (south of Thorpe Road, B1035)	Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
06-007	224 square metres of agricultural land (west of Thorpe Road, B1033)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights reserved by a Conveyance dated 14 March 1998)	
06-008	361 square metres of verge and lay-by (Thorpe Road, B1033)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-011	29988 square metres of agricultural land (south of Sneating Hall Lane, B1034)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
06-012	732 square metres of agricultural land (south of Sneating Hall Lane, B1034)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
06-014	2785 square metres of public highway, verges, hardstanding and drain (Sneating Hall Lane B1034)	, Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-014 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
06-015	263 square metres of agricultural land and verge (Sneating Hall Lane, B1034)	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)	
06-016	1057 square metres of agricultural land (north of Sneating Hall Lane, B1034)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
06-017	53667 square metres of agricultural land (north of Sneating Hall Lane, B1034)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-018	2430 square metres of public highway, verges and access splay (Damants Farm Lane, Thorpe-le-Soken)	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus) Unknown	
07-002	8745 square metres of agricultural land and drain (south of Walton Road, Thorpe-Le- Soken)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	
07-004	3729 square metres of agricultural land, hedgerow and private access track (south of Walton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
07-005	2540 square metres of agricultural land (south of Walton Road, Thorpe-Le-Soken)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-007	751 square metres of woodland (south of Walton Road, Thorpe-Le-Soken)	M Scott Property Group Limited Suite 5 Oyster House Severalls Lane Colchester CO4 9PD (in respect of a restriction dated 22 March 2021)	
07-008	1272 square metres of agricultural land and hedgerow (south of Walton Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
07-009	5430 square metres of agricultural land and hedgerow (south of Walton Road, Thorpe-Le- Soken)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
07-010	4212 square metres of agricultural land and hedgerow (east of Landermere Road, B1414)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
07-011	128000 square metres of agricultural land and hedgerow (south of Walton Road) and public footpath (FP 13 180)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-011 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) M Scott Property Group Limited Suite 5 Oyster House Severalls Lane Colchester CO4 9PD (in respect of a restriction dated 22 March 2021)	
07-013	3120 square metres of public highway (Landermere Road, B1414)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of gas apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-013 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
08-001	7 square metres of access splay (east of Landermere Road)	Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of gas apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
08-004	70 square metres of hedgerow, verge and access splay (Landermere Road, B1414)	Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-008	15530 square metres of agricultural land and hedgerow (west of Landermere Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	
08-009	1029 square metres of agricultural land and public footpath (FP 7 180)	Unknown (in respect of covenants contained in various Conveyances)	
08-010	418 square metres of private road and verges (Lonsdale Road, Thorpe-Le-Soken)	Andrew William Bacon Green Lane Farm Colchester Road Weeley CLACTON-ON-SEA Essex CO16 9AD (in respect of rights of way and maintenance) Benjamin Furness 7 Crownfields Crown Street Dedham COLCHESTER CO7 6AT (in respect of rights of way and maintenance) Brendan Malachy Gormley Gunfleet Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-010 cont'd		Cadent Gas Limited Ansty Pilot Way Ansty COVENTRY West Mildands CV7 9JU (in respect of gas apparatus) Colin Ernest Anson Framble Barn Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 OLF (in respect of rights of way and maintenance) Delicia Maria Ransom The Old Cottage The Street Copdock IPSWICH IP8 3HS (in respect of rights of way and maintenance) Dominic Furness Portiniemntie 21 Hameenlinna 1320 FINLAND (in respect of rights of way and maintenance)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-010 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Gwendolyn Rose Batley 87 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LW (in respect of rights of way and maintenance) Jane Elizabeth Anson Framble Barn Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance) John Frederick Peirson Blacksmiths Farm Harwich Road Beaumont CLACTON-ON-SEA CO16 0AS (in respect of rights of way and maintenance)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-010 cont'd		Julian Furness Cambridge House Amberfield Drive Nacton IPSWICH IP10 0GQ (in respect of rights of way and maintenance) Linda Draper Hamford Lonsdale Road Thorpe-ie-Soken CLACTON-ON-SEA C016 0LF (in respect of rights of way and maintenance) Margaret Shona Gormley Gunfleet Lonsdale Road Thorpe-ie-Soken CLACTON-ON-SEA C016 0LF (in respect of rights of way and maintenance) Mark Terry Sangwine 87 Landermere Road Thorpe-ie-Soken CLACTON-ON-SEA C016 0LF (in respect of rights of way and maintenance) Mark Terry Sangwine 87 Landermere Road Thorpe-ie-Soken CLACTON-ON-SEA C016 0LW (in respect of rights of way and maintenance)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-010 cont'd		Michael David Pollard 89 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NA (in respect of rights of way and maintenance) Michael William Gay Trinity Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Orwell Housing Association Limited Crane Hill Lodge 325 London Road IPSWICH Suffolk IP2 0BE (in respect of rights of way and maintenance)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-010 cont'd		Principal Homes Limited Oak House Church Lane Great Holland FRINTON-ON-SEA Essex CO13 0JS (in respect of rights of way and maintenance) Sharon Susan Gay Trinity Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance) Simon Stone Jimilda Lonsdale Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance) Susan Mary Pollard 89 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0LF (in respect of rights of way and maintenance) Susan Mary Pollard 89 Landermere Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0NA (in respect of rights of way and maintenance) Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-011	17889 square metres of agricultural land (west of Landermere Road)	Unknown (in respect of covenants contained in various Conveyances)	
08-013	31370 square metres of agricultural land and hedgerows (south of Golden Lane)	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by the Transfer dated 21 July 2010) Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by a Transfer dated 21 July 2010)	
08-014	2138 square metres of agricultural land and hedgerow (south of Golden Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Strutt & Parker (Farms) Limited Unit 1 Old Park Farm Main Road Ford End CHELMSFORD Essex CM3 1LN (in respect of rights granted by a Deed dated 11 August 1967)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-014 cont'd		Unknown (in respect of rights contained in a Conveyance dated 18 February 1952)	
08-015	405 square metres of agricultural land (east of Golden Lane)	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 OLE (in respect of rights reserved by the Transfer dated 21 July 2010) Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 OLE (in respect of rights reserved by a Transfer dated 21 July 2010)	
08-016	176 square metres of agricultural land (east of Golden Lane)	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by the Transfer dated 21 July 2010)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-016 cont'd		Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by a Transfer dated 21 July 2010)	
08-017	56 square metres of agricultural land (east of Golden Lane)	Anne Dorette Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by the Transfer dated 21 July 2010) Sidney Desmond Hutchby Frost Farm Golden Lane Thorpe-le-Soken Clacton-on-Sea Essex CO16 0LE (in respect of rights reserved by a Transfer dated 21 July 2010)	
08-018	1801 square metres of public highway and verges (Golden Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-019	34286 square metres of agricultural land, copse and public footpath (FP 3 180) (west of Golden Lane)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
08-020	4361 square metres of paddock and copse (east of Tendring Road, B1035)	Julie Wendy Thompson 92 Station Road CLACTON-ON-SEA Essex CO15 1SG (in respect of rights granted by a Transfer dated 28 May 2015)	
08-021	777 square metres of paddock, hedgerow and public footpath (FP 3 180) (east of Tendring Road, B1035)	Unknown	
08-022	2452 square metres of paddock and hedgerow (east of Tendring Road, B1035)	Julie Wendy Thompson 92 Station Road CLACTON-ON-SEA Essex CO15 1SG (in respect of rights granted by a Transfer dated 28 May 2015)	
08-023	1827 square metres of agricultural land and public footpaths (FP 1 180) (east of Tendring Road, B1035)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-024	12800 square metres of agricultural land, private access track and public footpaths (FP 3 180 and FP 4 180) (west of Golden Lane)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	
08-026	93108 square metres of agricultural land, pond, drain and public footpaths (FP 18 180 and FP 1 180) (east of Tendring Road, B1035)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
09-001	1136 square metres of agricultural land (east of Tendring Road, B1035)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-002	1910 square metres of public highway and verges (Tendring Road, B1035)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
09-003	32107 square metres of agricultural land and public footpath (FP 18 180) (north of Tendring Road, B1035)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
09-004	62 square metres of public highway and verges (Tendring Road, B1035)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-004 cont'd		Unknown	
09-005	694 square metres of agricultural land (east of Tendring Road, B1035)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
09-006	771 square metres of agricultural land (east of Tendring Road, B1035)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
09-007	852 square metres of public highway and verges (Tendring Road, B1035)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-008	13526 square metres of agricultural land (south of Thorpe Road and west of Tendring Road, B1035)	Laura Lilian Lord The Barn Bradley Hall Farm Whitehall Lane Thorpe-Le-Soken CLACTON-ON-SEA Essex CO16 0AH (in respect of rights reserved as contained in a Transfer dated 25 October 2017) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
09-009	1798 square metres of public highway and verges (Thorpe Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Unknown	
09-010	8269 square metres of agricultural land, woodland (Pond Farm) and public footpath (FP 18 159) (south of Barker's Lane)	Gopenreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-011	449 square metres of agricultural land (south of Swan Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
09-012	737 square metres of public highway and verges (Swan Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
09-013	6 square metres of hedgerow (west of Swan Road)	Unknown	

### $\bigvee \Xi$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-014	4271 square metres of garden (2 Barkers Hall Cottages), hedgerow and paddock (west of Swan Road)	Arthur Philip Wallis Pond Farm Swan Road Beaumont CLACTON-ON-SEA Essex CO16 0AN (in respect of rights as granted by a Transfer dated 15 June 2006)
09-015	339 square metres of agricultural land (north of Thorpe Road, B1035)	Unknown
09-016	4388 square metres of agricultural land (north of Thorpe Road, B1035)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
09-017	9105 square metres of agricultural land (north of Thorpe Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-018	22937 square metres of agricultural land (north of Thorpe Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	
09-020	42129 square metres of agricultural land and hedgerow (north of Thorpe Road, B1035)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-023	581 square metres of agricultural land (north of Thorpe Road, B1035)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Rob Long Jasmine Cottage Tendring Road Thorpe-le-Soken CLACTON-ON-SEA CO16 0AA (in respect of rights of access) Unknown	
09-024	42538 square metres of agricultural land and hedgerow (north of Thorpe Road, B1035)	David Milburn Cannell The Old Barn Thorpe Road Tendring CLACTON-ON-SEA Essex CO16 9AR (in respect of sporting rights granted by a Transfer dated 7 May 2004)	
10-002	515 square metres of agricultural land (north of Thorpe Road, B1035)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-002 cont'd		David Milburn Cannell The Old Barn Thorpe Road Tendring CLACTON-ON-SEA Essex CO16 9AR (in respect of sporting rights granted by a Transfer dated 7 May 2004)	
10-003	3447 square metres of agricultural land, private access track and hedgerow (north of Thorpe Road, B1035) and public footpaths (FP 22 179 and FP 8 179)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	
10-004	5435 square metres of agricultural land (north of Thorpe Road, B1035)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	
10-006	693 square metres of agricultural land, woodland and private access track (north of Thorpe Road, B1035)	Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-006 cont'd		Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)	
10-007	4639 square metres of agricultural land (north of Thorpe Road, B1035)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 OBS (in respect of rights granted by a Transfer dated 9 June 2011) Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 OBS (in respect of rights granted by a Transfer dated 9 June 2011)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-008		Affinity Water Limited Tamblin Way HATFIELD HATFORDSHIP AL10 9EZ (In respect of water apparatus) Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (In respect of rights granted by a Transfer dated 9 June 2011) Joanna Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (In respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (In respect of telecommunication apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-008 cont'd		Stephen Nigel Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access) Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)	
10-009	126 square metres of private road (known as Lodge Lane)	Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011) Joanna Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-009 cont'd		Stephen Nigel Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access) Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)	
10-011	22499 square metres of private road (known as Lodge Lane) and agricultural land (east of Lodge Lane)	Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 OBS (in respect of rights granted by a Transfer dated 9 June 2011) Joanna Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 OBS (in respect of rights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-011 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Stephen Nigel Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access) Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)	
10-012	175 square metres of private road (known as Lodge Lane)	Ian Martin Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA Essex CO16 OBS (in respect of rights granted by a Transfer dated 9 June 2011)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-012 cont'd		Joanna Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Stephen Nigel Mangham The Lodge Lodge Lane Tendring CLACTON-ON-SEA CO16 0BS (in respect of rights of access) Susan Beverley Patch Tanzara Lodge Lane Tendring CLACTON-ON-SEA ESEX CO16 0BS (in respect of rights granted by a Transfer dated 9 June 2011)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-013	120236 square metres of agricultural land (east of Heath Road) and public footpaths (FP 3 179 and FP 8 179)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of the unknown covenants contained in Conveyance dated 29 November 1982) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 15 October 1954) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 6 October 1956)	
11-003	578 square metres of private access track (south of Wolves Hall Lane)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-003 cont'd		Tendring District Council Town Hall Station Road CLACTON-ON-SEA Essex CO15 1SE (in respect of the unknown covenants contained in Conveyance dated 29 November 1982) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 15 October 1954) Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 6 October 1956)	
11-004	10 square metres of private access track (south of Wolves Hall Lane)	Unknown	
11-007	1879 square metres of public highway and verge (Wolves Hall Lane)	Unknown	
11-012	51258 square metres of agricultural land (south of Stones Green Road) and public footpath (FP 1 179)	Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 15 October 1954)	
11-013	2959 square metres of agricultural land (south of Stones Green Road)	Unknown (in respect of rights and restrictive covenants reserved by a Conveyance dated 15 October 1954)	
11-015	39 square metres of public highway and verge (Stones Green Road)	Unknown	
11-016	244 square metres of agricultural land (north of Stones Green Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-016 cont'd		Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)	
11-017	205 square metres of agricultural land and hedgerow (north of Stones Green Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)	
11-018	3001 square metres of public highway and verges (Stones Green Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
11-019	28 square metres of agricultural land (north of Stones Green Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-019 cont'd		Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)	
11-020	1069 square metres of agricultural land and public footpath (FP 31 183) (north of Stones Green Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)	
12-001	7 square metres of public highway and hedgerow (Stones Green Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	
12-002	680 square metres of agricultural land (north of Stones Green Road)	Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)	
12-003	1825 square metres of agricultural land (north of Stones Green Road)	Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-004	103132 square metres of agricultural land and public footpaths (FP 32 183 and FP 31 183) (north of Stones Green Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)	
12-005	2021 square metres of agricultural land, private track and public footpaths (FP 32 183, FP 14 183 and FP 31 183) (north of Stones Green Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-006	2123 square metres of agricultural land, private access track to Hempstalls Farm and public footpath (FP 15 183) (south of A120)	Aurlius Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Chris Driver Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Chris Mudd Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-006 cont'd		Harry Weavers Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Jay McDermot Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Lewis Clarke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Lisa Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Lisa Jiggens Bradfield Lodge Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NS (in respect of rights of access to Hempstalls Farm)	

#### $\bigvee \Xi$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-006 cont'd		Mark Taylor Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Nick Bowingcoke Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE CO11 2NZ (in respect of rights of access to Hempstalls Farm) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)	
12-007	92705 square metres of agricultural land and hedgerow (south of A120)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-007 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
12-008	279 square metres of public highway lay-by and verge (B1035)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	

#### $\bigvee \Xi$

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-008 cont'd		Tungsten Colchester Limited Gateway House 4 Penman Way Grove Park Enderby Leicester LE19 1SY (in respect of a commercial interest for development) Unknown	
12-009		Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
	371 square metres of agricultural land, private access track and public footpath (FP 37 138) (east of B1035)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-011	1061 square metres of agricultural land and hedgerow (east of B1035)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Unknown (in respect of rights contained in a Conveyance dated 17 November 1977)	
12-012	1178 square metres of public highway and verges (A120) (excluding all the interests of the National Highways Limited)	Unknown	
	63308 square metres of agricultural land, hedgerow and private access track (north of A120)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-013 cont'd		Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of CK8 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-013 cont'd		Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Claton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Claton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGIREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-013 cont'd		East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 85E (in respect of rights of access) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SEI 6NP (in respect of electricity apparatus) Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Clatton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-013 cont'd		Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 15 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of Unit 10 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-013 cont'd		Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of Units 8 and 16 Bradfield Lodge) Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-013 cont'd		S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge) Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-014	3224 square metres of agricultural land and private access track (north of A120)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of access to CK8 and Scaffold Yard Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-014 cont'd		Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Carl Tarrant Unit 2 Bradfield Lodge Carl Tarrant Unit 2 Bradfield Lodge Carl Tarrant Unit 2 Bradfield Lodge Caclon Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-014 cont'd		Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 85E (in respect of fights of access) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-014 cont'd		Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Clatcon Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge) Jim Storey Unit 6 Bradfield Lodge Clatcon Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clatcon Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clatcon Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-014 cont'd		Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 15 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of Unit 10 Bradfield Lodge) Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of Unit 23 Bradfield Lodge) Mick Innis Unit 23 Bradfield Lodge) Mick Innis Unit 23 Bradfield Lodge) Mick Innis Unit 24 Bradfield Lodge) Mick Innis Unit 25 Bradfield Lodge) Mick Innis Unit 26 Bradfield Lodge) Mick Innis Unit 27 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGIRTEE CO11 2NS (in respect of Unit 28 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGIRTEE CO11 2NS (in respect of Units 8 and 16 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-014 cont'd		Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (In respect of Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (In respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (In respect of access to Unit 7 Bradfield Lodge) S cott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (In respect of access to CK7 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-014 cont'd		Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)	
13-001	230 square metres of private access track (north of A120)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-001 cont'd		Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of access to CK7 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-001 cont'd		Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Claton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Claton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-001 cont'd		East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 85E (in respect of rights of access) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SEI 6NP (in respect of electricity apparatus) Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-001 cont'd		lim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 15 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of Unit 10 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-001 cont'd		Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of Units 8 and 16 Bradfield Lodge) Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree Clacton Road Manningtree CO11 2NS (in respect of Unit 21 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-001 cont'd		S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge) Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access in Unit 14 Bradfield Lodge)	

Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-002 106	6 square metres of private access track and hardstanding (Bradfield Lodge)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNINGTREE Essex CO11 2NX (in respect of access to CK8 and Scaffold Yard Bradfield Lodge)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-002 cont'd		Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Carl Tarrant Unit 2 Bradfield Lodge Carl Tarrant Unit 2 Bradfield Lodge Carl Tarrant Unit 2 Bradfield Lodge Caclon Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-002 cont'd		Darren Smith Units 17 and 18a Bradfield Logg Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 17 Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 85E (in respect of rights of access) Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-002 cont'd		Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2) Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge) Kanye Ault Unit 15 Bradfield Lodge Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 15 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-002 cont'd		Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge) Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 8 Bradfield Lodge) Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE CO11 2NS (in respect of access to Unit 8 Bradfield Lodge) Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-002 cont'd		Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge) Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to CK7 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-002 cont'd		Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)	
13-003	299 square metres of private access track and hardstanding (Bradfield Lodge)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-003 cont'd		Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of access to CK8 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 5 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-003 cont'd		Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 14 Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of frights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-003 cont'd		Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Claton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge) Jim Storey Unit 6 Bradfield Lodge Claton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-003 cont'd		Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 15 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Claton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge) Mick Curmins Unit 23 Bradfield Lodge Claton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Claton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 28 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Claton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 28 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Claton Road MANNGINGTREE CO11 2NS (in respect of access to Units 8 and 16 Bradfield Lodge) (in respect of access to Units 8 and 16 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-003 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-003 cont'd		Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree Claton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)	
13-004	78 square metres of private access track to Bradfield Lodge (south of Clacton Road)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-004 cont'd		Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINCTREE Essex CO11 2NX (in respect of access to CK8 and Scaffold Yard) Arron Ingram Unit 18b Bradfield Lodge Clacton Road MANNIGINCTREE Essex CO11 2NX (in respect of access to Unit 18 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-004 cont'd		Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 12 Bradfield Lodge) Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNIGNEREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-004 cont'd		Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 85E (in respect of rights of access) Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-004 cont'd		Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 15 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-004 cont'd		Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 8 and 16 Bradfield Lodge) Openreach Limited 6 Graecchurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Peter Barrat Unit 3 Bradfield Lodge MANNINTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-004 cont'd		Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge) Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to CK7 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-004 cont'd		Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)	
13-005	235 square metres of private access track to Bradfield Lodge (off Clacton Road)	A Johns Unit 22 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Alan Pole Unit 11 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-005 cont'd		Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINSTREE ESSEX CO11 2NX (in respect of CK8 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge MANNINGTREE ESSEX CO11 2NS (in respect of access to Unit 12 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-005 cont'd		Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 17 Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of frights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-005 cont'd		Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Claton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge) Jim Storey Unit 6 Bradfield Lodge Claton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge) Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-005 cont'd		John Charles Jiggens Hempstalls Farm Clacton Road Horsley Cross MANNINGTREE Essex CO11 2NZ (in respect of rights contained in a transfer dated the 22nd February 2011) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 15 Bradfield Lodge) Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge) Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNINGTREE CO11 2NS (in respect of access to Unit 10 Bradfield Lodge) Mick Cummins Unit 23 Bradfield Lodge Clacton Road MANNINGTREE CO11 2NS (in respect of access to Unit 23 Bradfield Lodge) (in respect of access to Unit 23 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-005 cont'd		Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 8 and 16 Bradfield Lodge) Openreach Limited 6 Gracechurch Street LONDON EC3V DAT (in respect of telecommunication apparatus) Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-005 cont'd		S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge) Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-006	829 square metres of private access track and verges (south of Clacton Road)	A Johns Unit 22 Bradfield Lodge Clatton Road Manningtree CO11 2NS (in respect of access to Unit 22 Bradfield Lodge) Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Alan Pole Unit 11 Bradfield Lodge Clatton Road Manningtree CO11 2NS (in respect of access to Unit 11 Bradfield Lodge) Albert Du Plessis Unit 19 Bradfield Lodge Clatton Road Manningtree CO11 2NS (in respect of access to Unit 19 Bradfield Lodge) (in respect of access to Unit 19 Bradfield Lodge) (in respect of access to Unit 19 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-006 cont'd		Arron CK8 and Scaffold Yard Bradfield Lodge Clacton Road MANNGINGTREE Essex CO11 2NX (in respect of access to CK8 and Scaffold Yard Bradfield Lodge) Arron Ingram Unit 18b Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 18b Bradfield Lodge) Ben Ross Unit 5 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 5 Bradfield Lodge) Blake Birchill Unit 12 Bradfield Lodge Clatton Road ManNINGTREE Essex CO11 2NS (in respect of access to Unit 2 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-006 cont'd		Carl Tarrant Unit 2 Bradfield Lodge Clacton Road Manningtree CO11 2SN (in respect of access to Unit 2 Bradfield Lodge) Darren Smith Units 17 and 18a Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 17 and 18a Bradfield Lodge) Dave Sayer Units 1 and 4 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 1 and 4 Bradfield Lodge) East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-006 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SEI GNP (in respect of electricity apparatus) Jason Childs Units 9 and 13 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Units 9 and 13 Bradfield Lodge) Jason Hinsen Feed Store 2 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge) Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 2 Bradfield Lodge) Jim Storey Unit 6 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 6 Bradfield Lodge)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-006 cont'd		Joe Piggot Unit 20 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 20 Bradfield Lodge) Kanye Ault Unit 15 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 15 Bradfield Lodge) Kelsy Jane Bamford 2 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 2 Jubilee Cottage) Leah Faye Banks 1 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 1 Jubilee Cottage) Leah Faye Banks 1 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 1 Jubilee Cottage)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-006 cont'd		Les Troller Unit 10 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 10 Bradfield Lodge) Mick Curmins Unit 23 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Unit 23 Bradfield Lodge) Mick Innis Units 8 and 16 Bradfield Lodge Clacton Road MANNGINGTREE CO11 2NS (in respect of access to Units 8 and 16 Bradfield Lodge) (in respect of access to Units 8 and 16 Bradfield Lodge) Openreach Limited 6 Graeechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-006 cont'd		Peter Anthony Banks 1 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 1 Jubilee cottage) Peter Barrat Unit 3 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to Unit 3 Bradfield Lodge) Rickman Unit 21 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 21 Bradfield Lodge) Robin Neal Banks 1 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 1 Jubilee Cottage) Robin Neal Banks 1 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 1 Jubilee Cottage)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-006 cont'd		S Smith Unit 7 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 7 Bradfield Lodge) Scott Adams CK7 Bradfield Lodge MANNINGTREE Essex CO11 2NS (in respect of access to CK7 Bradfield Lodge) Stairform Feed Store 1 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Feed Store 1 Bradfield Lodge) Thomas William Wright 2 Jubilee Cottages Clacton Road Horsley Cross MANNINGTREE CO11 2NS (in respect of a right of access to 2 Jubilee Cottage)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-006 cont'd		Tony Robinson Unit 14 Bradfield Lodge Clacton Road Manningtree CO11 2NS (in respect of access to Unit 14 Bradfield Lodge)	
13-008	20577 square metres of agricultural land (east of Clacton Road B1035)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
13-009	10798 square metres of agricultural land (east of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-010	1594 square metres of agricultural land (east of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970)	
13-011	350 square metres of agricultural land (east of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970)	
13-012	195 square metres of agricultural land and hedgerow (east of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights granted in a Deed of Grant dated 6 June 1961 and in a Deed of Grant dated 8 July 1970) Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	
13-013	183 square metres of public highway and verge (Clacton Road) (excluding all the interests of the National Highways Limited)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-013 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
13-014	259 square metres of public highway and verge (east of Clacton Road) (excluding all the interests of the National Highways Limited)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
13-015	60 square metres of agricultural land, verge and access splay (west of Clacton Road) (excluding all the interests of the National Highways Limited)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	
13-016	3 square metres of agricultural land, verge and access splay (west of Clacton Road) (excluding all the interests of the National Highways Limited)	Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-017	25799 square metres of agricultural land (west of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights and restrictive covenants as contained in Deed dated 29 April 1936, Deed dated 6 June 1961, Deed dated 8 July 1970 and Deed dated 31 December 1990) Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Nigel Graham Dyson Bentley Manor Church Road Little Bentley COLCHESTER Essex CO7 8SE (in respect of right of access to sell goods) Robert Christmas Mulleys Cottage Bentley Road Little Bromley MANNINGTREE CO11 2PL (in respect of shooting rights)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-018	537 square metres of public highway and verges (Clacton Road) (excluding all the interests of the National Highways Limited)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)
13-019	3199 square metres of public highway and verges (Clacton Road) (excluding all the interests of the National Highways Limited)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown
13-020	800 square metres of public highways, verges and agricultural land (east of Clacton Road) (excluding all the interests of the National Highways Limited)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-022	39078 square metres of agricultural land, private access track and verge (west of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (In respect of water apparatus) Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (In respect of the rights and restrictive covenants as contained in Deed dated 29 April 1936, Deed dated 6 June 1961, Deed dated 8 July 1970 and Deed dated 31 December 1990) Nigel Graham Dyson Bentley Manor Church Road Little Bentley COLCHESTER Essex CO7 8SE (In respect of right of access to sell goods) Robert Christmas Mulleys Cottage Bentley Road Little Bentley COL1 2PL (In respect of shooting rights)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-023	32744 square metres of agricultural land and brook (Holland Brook) (west of Clacton Road)	Robert Christmas Mulleys Cottage Bentley Road Little Bromley MANNINGTREE CO11 2PL (in respect of shooting rights)	
14-001	377 square metres of agricultural land (west of Clacton Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of the rights and restrictive covenants as contained in Deed dated 29 April 1936, Deed dated 6 June 1961, Deed dated 8 July 1970 and Deed dated 31 December 1990) Robert Christmas Mulleys Cottage Bentley Road Little Bromley MANNINGTREE CO11 2PL (in respect of shooting rights)	
14-003	2114 square metres of private access track and brook (Welhams Farm) (east of Bentley Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-010	24 square metres of private access track (known as Mulberry Lane) and access splay (east of Bentley Road)	Unknown	
14-012	254 square metres of verge (east of Bentley Road)	Unknown	
14-013	701 square metres of agricultural land, public highway and verge (Bentley Road)	Unknown	
14-014	591 square metres of public highway (Bentley Road)	Unknown	
14-015	313 square metres of agricultural land, public highway and verges (Bentley Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
14-016	345 square metres of agricultural land, public highway and verges (Bentley Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
14-017	26 square metres of agricultural land and verge (west of Bentley Road)	Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-018	125 square metres of agricultural land (west of Bentley Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
14-019	10 square metres of agricultural land (west of Bentley Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
14-020	18 square metres of agricultural land and verge (west of Bentley Road)	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)	
14-021	14 square metres of agricultural land (west of Bentley Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
14-022	4709 square metres of agricultural land (west of Bentley Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-022 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
14-023	1438 square metres of agricultural land (west of Bentley Road)	Unknown	
14-025	99 square metres of verge (east of Bentley Road)	Unknown	
14-026	592 square metres of public highway (Bentley Road)	Unknown	
14-027	285 square metres of public highway (Bentley Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
14-028	66 square metres of agricultural land and verge (west of Bentley Road)	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)	
14-029	12 square metres of agricultural land and verge (west of Bentley Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-029 cont'd		Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)	
14-030	103 square metres of agricultural land (west of Bentley Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
14-031	453 square metres of agricultural land (west of Bentley Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
14-032	2818 square metres of agricultural land, drain, verge and access splay (east of Bentley Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
14-033	3495 square metres of public highway (Bentley Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-033 cont'd		Unknown	
14-034	57 square metres of public highway and verge (Bentley Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)	
14-036	838 square metres of agricultural land (west of Bentley Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
14-038	17211 square metres of agricultural land (east of Payne's Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
14-039	51 square metres of verge (east of Payne's Lane)	Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-040	7 square metres of verge (east of Payne's Lane)	Henry Fairley & Son Limited Park Farm Hilliards Road Great Bromley COLCHESTER Essex CO7 7US (in respect of right of way) Unknown	
14-041	2 square metres of verge (east of Payne's Lane)	Unknown	
14-042	64 square metres of public highway and verges (Payne's Lane)	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)	
14-043	715 square metres of public highway (Payne's Lane)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-044	146 square metres of agricultural land (west of Payne's Lane)	Unknown	
14-045	23636 square metres of agricultural land and grass land (west of Payne's Lane)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
14-046	23615 square metres of agricultural land and grass land (west of Payne's Lane)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
15-001	1085 square metres of agricultural land and hedgerow (east of Bentley Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-002	953 square metres of public highway and verges (Bentley Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
15-003	10 square metres of verge (east of Bentley Road)	Unknown	
15-005	62 square metres of public highway and verges (Bentley Road)	Unknown (in respect of rights of access over a track contained in a Agreement dated 15 September 1930)	
15-006	3089 square metres of agricultural land (west of Bentley Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights contained in a Deed dated 22 February 2008)	
15-007	8381 square metres of public highway and verges (Bentley Road) (excluding all the interests of the National Highways Limited)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-007 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
15-011	646 square metres of agricultural land (north of Bentley Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights contained in a Conveyance dated 26 June 1968)	
15-013	348 square metres of agricultural land and hedgerow (north of Bentley Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-013 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus) Unknown (in respect of rights contained in a Conveyance dated 26 June 1968)
15-014	74 square metres of public highway and verges (Bentley Road) (excluding all the interests of the National Highways Limited)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown
15-016	2142 square metres of public highway and verge (Harwich Road, A120) (excluding all the interests of the National Highways Limited)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus)

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-016 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
15-017	2857 square metres of public highway (A120) (excluding all the interests of the National Highways Limited)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
15-018	200 square metres of public highway and verge (A120) (excluding all the interests of the National Highways Limited)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-020	334 square metres of public highway and verge (A120) (excluding all the interests of the National Highways Limited)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
15-021	219 square metres of public highway and verges (A120) (excluding all the interests of the National Highways Limited)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
15-022	103 square metres of public highway (Little Bromley Road) hedgerow and verge (south of A120) (excluding all the interests of the National Highways Limited)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
16-001	336 square metres of agricultural land (east of Spratts Lane)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-002	42 square metres of public highway and verge (Spratts Lane)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-003	10 square metres of public highway and verge (Spratts Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-004	18 square metres of agricultural land (east of Spratts lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-005	1559 square metres of public highway and verges (Spratts Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
16-006	15 square metres of public highway and verges (Spratts Lane)	Unknown	
16-007	41 square metres of public highway and verge (Spratts Lane)	Unknown	
16-008	274 square metres of agricultural land and hedgerow (west of Spratts Lane)	Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-009	144 square metres of agricultural land (west of Spratts Lane)	Unknown	
16-010	198 square metres of agricultural land (west of Spratts Lane)	Unknown	
16-011	146 square metres of agricultural land (west of Spratts Lane)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-012	30979 square metres of agricultural land and public footpath (FP 17 172) (east of Barlon Road)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-013	33129 square metres of agricultural land (east of Barlon Road) and public footpath (FP 17 172)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-014	85 square metres of verge (east of Barlon Road)	Unknown	
16-015	41 square metres of public highway and agricultural land (Barlon Road)	Unknown (in respect of rights granted as contained in a Conveyance dated 25 March 1963)	
16-016	710 square metres of public highway and verges (Barlon Road)	Unknown	
16-020	32985 square metres of agricultural land, private access track and public footpath (FP 16 172) (west of Barlon Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-021	33719 square metres of agricultural land, private access track and public footpath (FP 16 172) (west of Barlon Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)	
16-022	7974 square metres of agricultural land, private access tack and public footpath (FP 16 172) (west of Barlon Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access)	
16-023	1494 square metres of agricultural land, private access track (part of Hall Farm) and public footpath (FP 16 172) (west of Barlon Road)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-001	17377 square metres of agricultural land and hedgerow (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	
17-002	15087 square metres of agricultural land and hedgerow (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-002 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
17-003	2442 square metres of agricultural land and drain (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-004	9600 square metres of agricultural land and hedgerow (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
17-005	12584 square metres of agricultural land (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-005 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	
17-006	12405 square metres of private access track to Catts Green Farm (off Ardleigh Road), hedgerow and public footpath (FP 15 172)	East Anglian Farm Rides Manor Bungalow Church Road Little Bentley COLCHESTER CO7 8SE (in respect of rights of access) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N SEH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-006 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus)	
17-007	14 square metres of private access track to Catts Green Farm (off Ardleigh Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
17-008	1623 square metres of public highway, verges (Ardleigh Road) and public footpath (FP 15 172)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-009	10 square metres of agricultural land and verge (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Unknown (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	
17-010	16 square metres of agricultural land (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	
17-011	51 square metres of agricultural land (south of Ardleigh Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-012	411 square metres of agricultural land (south of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus)	
17-013	61 square metres of agricultural land and verge (south of Ardleigh Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-014	489 square metres of public highway and verges (Ardleigh Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
17-015	30 square metres of agricultural land and verge (north of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Unknown (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965)	
17-016	175 square metres of agricultural land and verge (north of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-017	107 square metres of agricultural land and verge (north of Ardleigh Road)	Unknown	
17-018	428 square metres of agricultural land and verge (north of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)	
17-019	35 square metres of agricultural land and verge (north of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-019 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)	
17-020	407 square metres of public highway and verges (Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
17-021	342 square metres of public highway and verges (Ardleigh Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-022	933 square metres of public highway and verges (Ardleigh Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
17-023	120 square metres of agricultural land (west of Ardleigh Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
17-024	296427 square metres of agricultural land and drain (north of Ardleigh Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-024 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)	
17-025	182196 square metres of agricultural land and verge (east of Grange Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights contained in a Deed dated 10 December 1942)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-026	2436 square metres of agricultural land and verge (north of Ardleigh Road)	Affinity Water Limited Tamblin Way HATFIELD Hertfordshire A.110 9E2 (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to a erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N SEH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-027	64 square metres of agricultural land and verge (east of Grange Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)	
17-028	463 square metres of agricultural land (east of Grange Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to erect and maintain electrical lines contained in a Deed of Grant dated 27 May 1965) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Description of Land is proposed shall be extinguished, suspended or interfered with un		Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-028 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain and repair electrical lines contained in a Deed of Grant dated 27 May 1965) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of right of way contained in a Conveyance dated 11 January 1962)	
17-029	6 square metres of verge (east of Grange Road)	Unknown	
17-030	2838 square metres of public highway and verges (Grange Road)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-031	338602 square metres of agricultural land, hedgerow and electricity pylon (west of Grange Road)	Affinity Water Limited Tambiin Way HATFIELD Hertfordshire AL10 9EZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Moorhouse Farms Limited Begbies Traynor Town Wall House 4 Balkerne Hill COLCHSTER CO2 3AD (in respect of rights granted by a Deed dated 19 July 2018) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of telecommunication apparatus) Philip Douglas Reeve Waterhouse Lane Ardleigh COLCHSTER Essex CO7 7NB (in respect of rights contained in a Conveyance dated 4 July 1985)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Description of Land is proposed shall be extinguished, suspended or interfer		Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-031 cont'd		Unknown (in respect of restrictive covenants and rights reserved by a Transfer dated 3 April 1995)	
cont'd       (in respect of restrictive covenants and rights reserved by a Transfer dated 3 April 1995)         19-001       506 square metres of river (River Ore) and jetty (south of Orford Quay) (excluding all the interests of the Crown)       Orford and Gedgrave Parish Council         The Town Hall       Market Hill       Orford         Woodbridge       IP12 2NZ       (in respect of restrictive covenants as contained in a Deed dated 5 February 1987)         Secretary of State for Defence       Property Legal Team         Defence Infrastructure Organisation Malipoint 2216       Poplar 2         Abbey Woodb       Bristol         BS34 8JH       (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987)         The National Trust For Places Of Historic Interest Or Natural Beauty       Kemble Drive         SWINDON       Willshire         SWINDN       SWINDN		The Town Hall Market Hill Orford Woodbridge IP12 2NZ (in respect of restrictive covenants as contained in a Deed dated 5 February 1987) Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987) The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire	
19-002	14323 square metres of private road and jetty (south of River Ore) (excluding all the interests of the Crown)	East Anglia One North Limited 1 Tudor Street LONDON EC4Y 0AH (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-002 cont'd		East Anglia Two Limited 1 Tudor Street LONDON ECAY QAH (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 GNP (in respect of electricity apparatus) Norfolk Boreas Limited 70 St Mary Axe LONDON EC3A 88E (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Norfolk Vanguard West Limited 70 St Mary Axe LONDON EC3A 88E (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Norfolk Vanguard West Limited 70 St Mary Axe LONDON EC3A 88E (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Openreach Limited 6 Gracechurch Street LONDON EC3V QAT (in respect of telecommunication apparatus)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-002 cont'd		Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987, in a Conveyance dated 26 August 1913 and in a Conveyance dated 20 January 1914)	
		The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA (in respect of rights of way and access to maintain services as contained in Conveyance dated 31 March 1993)	
		The United Kingdom Atomic Energy Authority Culham Centre for Fusion Energy Culham Science Centre ABINGDON OX14 3DB (in respect of rights and restrictive covenants a contained a Deed dated 25 July 1962)	
		Unknown (in respect of rights as stated in Conveyance dated 26 August 1913) Unknown (in respect of rights as stated in Conveyance dated 20 January 1914)	
		Unknown (in respect of right of way as contained in Conveyance dated 30 June 1925)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-003	16 square metres of river (River Ore) (excluding all the interests of the Crown)	Orford and Gedgrave Parish Council The Town Hall Market Hill Orford Woodbridge IP12 2NZ (in respect of restrictive covenants as contained in a Deed dated 5 February 1987) Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987)	
19-004	14 square metres of private access track and scrubland (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993) Unknown (in respect of rights as stated in Conveyance dated 26 August 1913)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-005	78 square metres of private access track (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993)	
19-006	86 square metres of private access track (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-006 cont'd		Unknown (in respect of right of way as stated in Conveyance dated 30 June 1925)	
19-007	97 square metres of private access track (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993)	
20-001	60 square metres of private access track (south of River Ore)	Cobra Mist Limited The Old Vicarage Priory Road Castle Acre KING'S LYNN PE32 2AA (in respect of rights reserved as stated in Conveyance dated 31 March 1993) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown (in respect of rights as stated in Conveyance dated 31 March 1993)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
20-003	69153 square metres of scrubland, drains, tracks and radio masts (east of River Ore) (excluding all the interests of the Crown)	East Anglia One North Limited 1 Tudor Street LONDON EC4Y DAH (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) East Anglia Two Limited 1 Tudor Street LONDON EC4Y DAH (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Norfolk Boreas Limited 70 St Mary Axe LONDON EC3A BBE (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Norfolk Vanguard West Limited 70 St Mary Axe LONDON EC3A BBE (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Norfolk Vanguard West Limited 70 St Mary Axe LONDON EC3A BBE (in respect of rights of way and access to maintain services granted as contained in Lease dated 29 July 2022) Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 B/H (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987, in a Conveyance dated 26 August 1913 and dated 20 January 1914)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Essex and Suffolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
20-003 cont'd		The National Trust For Places Of Historic Interest Or Natural Beauty Kemble Drive SWINDON Wiltshire SN2 2NA (in respect of rights of way and access to maintain services as contained in Conveyance dated 31 March 1993) The United Kingdom Atomic Energy Authority Culham Centre for Fusion Energy Culham Science Centre ABINGDON OX14 3DB (in respect of rights and restrictive covenants a contained a Deed dated 25 July 1962) Unknown (in respect of rights as stated in Conveyance dated 20 January 1914)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 Counties of Essex and Suffolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
19-001	Acquisition of Rights	506 square metres of river (River Ore) and jetty (south of Orford Quay) (excluding all the interests of the Crown)	Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987)	
19-002	Acquisition of Rights	14323 square metres of private road and jetty (south of River Ore) (excluding all the interests of the Crown)	Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987, in a Conveyance dated 26 August 1913 and in a Conveyance dated 20 January 1914)	
19-003	Acquisition of Rights	16 square metres of river (River Ore) (excluding all the interests of the Crown)	Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987)	

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 Counties of Essex and Suffolk						
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made				
20-003		the interests of the Crown)	Secretary of State for Defence Property Legal Team Defence Infrastructure Organisation Mailpoint 2216 Poplar 2 Abbey Wood Bristol BS34 8JH (in respect of rights and restrictive covenants as contained in a Deed dated 5 February 1987, in a Conveyance dated 26 August 1913 and dated 20 January 1914)				

	Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 5 Counties of Essex and Suffolk						
Number on Land Plans	Extent of acquisition or use	Description of land	Category of Land				
01-001	Temporary Possession	52 square metres of verges, copse and access splay (east of Frinton Road, B1032)	Open Space				
01-002	Temporary Possession	6416 square metres of access way (off Frinton Road) leading to public road (Manor Way) verges and access tracks (north of The Esplanade, Holland-on-Sea) and public footpath (FP 29 167)	Open Space				
01-003	Temporary Possession	10490 square metres of access way off Frinton Road leading to public road (Manor Way) grassland, verges, access tracks and buildings (south of The Esplanade, Holland-on-Sea) and public footpath (FP 29 167)	Open Space				
01-004	Temporary Possession	63 square metres of hardstanding (south of Holland Brook) and public footpath (FP 29 167)	Open Space				
01-005	Temporary Possession	93 square metres of hardstanding (south of Holland Brook) and public footpath (FP 29 167)	Open Space				
01-006	Temporary Possession	3218 square metres of foreshore, sea wall and access track (east of Manor Way, Holland-on-Sea) and public footpath (FP 29 167)	Open Space				
01-007	Acquisition of Rights	21703 square metres of foreshore, sea wall, rock armour, grassland and sloping masonry (west of Second Avenue, Frinton-on-Sea)	Open Space				
01-008	Acquisition of Rights	29586 square metres of foreshore, grassland, access track, sea wall, rock armour, sloping masonry and drains (west of Second Avenue, Frinton-on-Sea) public bridleway (BR 2 164) and public footpaths (FP 3 164, FP 41 164 and FP 29 167)	Open Space				
01-009	Acquisition of Rights	9511 square metres of agricultural land (east of Manor Way, Holland-on-Sea)	Open Space				
02-001	Acquisition of Rights	16315 square metres of beach, rock armour, sloping masonry, groynes and foreshore at Frinton-On-Sea (south of Frinton Golf Course)	Open Space				
02-001A	Acquisition of Rights	1294 square metres of beach, rock armour and sloping masonry at Frinton-On-Sea (south of Frinton Golf Course)	Open Space				
02-002	Acquisition of Rights	937 square metres of access track (west of Second Avenue, Frinton-on-Sea) and public bridleway (BR 2 164)	Open Space				

Five Estuaries Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 5 Counties of Essex and Suffolk							
Number on Land Plans	Extent of acquisition or use	Description of land	Category of Land				
19-004	Acquisition of Rights	14 square metres of private access track and scrubland (south of River Ore)	National Trust Land				
19-005	Acquisition of Rights	78 square metres of private access track (south of River Ore)	National Trust Land				
19-006	Acquisition of Rights	86 square metres of private access track (south of River Ore)	National Trust Land				
19-007	Acquisition of Rights	97 square metres of private access track (south of River Ore)	National Trust Land				
20-001	Acquisition of Rights	60 square metres of private access track (south of River Ore)	National Trust Land				



0333 880 5306 fiveestuaries@rwe.com www.fiveestuaries.co.uk

Five Estuaries Offshore Wind Farm Ltd Windmill Hill Business Park Whitehill Way, Swindon, SN5 6PB Registered in England and Wales company number 12292474